UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, D.C. 20549

FORM 10-Q

(Mark One)

X QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934.

For the quarterly period ended June 30, 2008

OR

o TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934.

For the transition period from

to

Commission file number 0-51813

LIQUIDITY SERVICES, INC.

(Exact Name of Registrant as Specified in Its Charter)

Delaware

(State or Other Jurisdiction of Incorporation or Organization)

52-2209244

(I.R.S. Employer Identification No.)

1920 L Street, N.W., 6th Floor, Washington, D.C.

(Address of Principal Executive Offices)

20036

(Zip Code)

(202) 467-6868

(Registrant's Telephone Number, Including Area Code)

(Former Name, Former Address and Former Fiscal Year, If Changed Since Last Report)

Indicate by check mark whether the registrant: (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes x No o

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, or a non-accelerated filer. See definition of "accelerated filer and large accelerated filer" in Rule 12b-2 of the Exchange Act. (Check One):

Large accelerated filer o

Accelerated filer s

Non-accelerated filer o

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act).

Yes o No x

The number of shares outstanding of the issuer's common stock, par value \$.001 per share, as of August 8, 2008 was 27,990,987.

Table of Contents

INDEX

Part I. FINANCIAL INFORMATION (UNAUDITED)

Page

<u>Item 1.</u> <u>Consolidated Financial Statements</u>

3

Consolidated Balance Sheets

3

	Consolidated Statements of Operations	4
	Consolidated Statements of Cash Flows	5
	Notes to the Unaudited Consolidated Financial Statements	6
Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	13
<u>Item 3.</u>	Quantitative and Qualitative Disclosures About Market Risk	26
<u>[tem 4.</u>	Controls and Procedures	27
Part II. OTH	ER INFORMATION	
<u>Item 1.</u>	<u>Legal Proceedings</u>	28
Item 1A.	Risk Factors	28
<u>Item 6.</u>	<u>Exhibits</u>	29
SIGNATURE	<u>28</u>	30

Table of Contents

PART I—FINANCIAL INFORMATION

Item 1. Consolidated Financial Statements.

Liquidity Services, Inc. and Subsidiaries Consolidated Balance Sheets (Dollars in Thousands)

June 30,

September 30,

		2008		2007
	(Unaudited)		
Assets				
Current assets:	ф	E0 (22)	ф	20.05.4
Cash and cash equivalents	\$	50,622	\$	39,954
Short-term investments		726		21,655
Accounts receivable, net of allowance for doubtful accounts of \$1,072 and \$371 at June 30, 2008 and				
September 30, 2007, respectively		4,396		5,098
Inventory		16,637		16,467
Prepaid expenses and other current assets		7,119		5,486
Total current assets		79,500		88,660
Property and equipment, net		4,717		4,202
Intangible assets, net		6,088		4,568
Goodwill		36,516		11,446
Other assets		3,034		2,266
Total assets	\$	129,855	\$	111,142
Liabilities and stockholders' equity				
Current liabilities:				
Accounts payable	\$	8,902	\$	3,333
Accrued expenses and other current liabilities		7,808		10,299
Profit-sharing distributions payable		8,015		6,919
Customer payables		7,687		6,329
Current portion of capital lease obligations		3		5
Current portion of long-term debt		_		13
Total current liabilities		32,415		26,898
Capital lease obligations, net of current portion		3		5
Long-term debt, net of current portion		_		29
Other long-term liabilities		2,971		2,176
Total liabilities	-	35,389	-	29,108
Stockholders' equity:		, i		Í
Common stock, \$0.001 par value; 120,000,000 shares authorized; 27,968,465 and 27,939,059 shares issued				
and outstanding at June 30, 2008 and September 30, 2007, respectively		28		28
Additional paid-in capital		64,386		60,820
Accumulated other comprehensive income		663		653
Retained earnings		29,389		20,533
Total stockholders' equity		94,466		82,034
Total liabilities and stockholders' equity	\$	129,855	\$	111,142
	-	120,000	*	

See accompanying notes to the unaudited consolidated financial statements.

Liquidity Services, Inc. and Subsidiaries Unaudited Consolidated Statements of Operations (Dollars in Thousands Except Per Share Data)

	Three Months Ended June 30,			Nine Months Ended June 30,				
		2008	_	2007		2008	_	2007
Revenue	\$	71,473	\$	52,505	\$	193,579	\$	146,952
Costs and expenses:								
Cost of goods sold (excluding amortization)		19,552		13,291		51,117		33,292
Profit-sharing distributions		24,200		17,559		67,636		54,178
Technology and operations		10,411		8,125		30,689		24,365
Sales and marketing		4,469		3,556		12,519		9,745
General and administrative		5,827		4,704		15,941		12,189
Amortization of contract intangibles		203		203		610		610
Depreciation and amortization		584		355		1,436		935
Total costs and expenses		65,246		47,793		179,948		135,314
Income from operations		6,227		4,712		13,631		11,638
Interest income (expense) and other income, net		292		475		1,402		1,624
Income before provision for income taxes		6,519		5,187		15,033		13,262
Provision for income taxes		(2,672)		(2,134)		(6,176)		(5,422)
Net income	\$	3,847	\$	3,053	\$	8,857	\$	7,840
Basic earnings per common share	\$	0.14	\$	0.11	\$	0.32	\$	0.28
		_		_				
Diluted earnings per common share	\$	0.14	\$	0.11	\$	0.32	\$	0.28
Basic weighted average shares outstanding		27,964,662		27,857,115		27,953,526		27,720,937
Diluted weighted average shares outstanding		28,237,150		28,321,395		28,201,988		28,215,689

See accompanying notes to the unaudited consolidated financial statements.

4

Table of Contents

Purchases of property and equipment

Liquidity Services, Inc. and Subsidiaries Unaudited Consolidated Statements of Cash Flows (In Thousands)

		Nine Months Ended June 30,		ne 30
	-	2008	naca sa	2007
Operating activities				
Net income	\$	8,857	\$	7,840
Adjustments to reconcile net income to net cash provided by operating activities:				
Depreciation and amortization		2,046		1,545
Stock compensation expense		3,440		1,409
Provision for doubtful accounts		(64)		
Changes in operating assets and liabilities:				
Accounts receivable		1,846		(1,519)
Inventory		1,304		(7,821)
Prepaid expenses and other assets		(1,975)		(2,540)
Accounts payable		3,200		247
Accrued expenses and other		(4,537)		2,101
Profit-sharing distributions payable		1,096		(1,519)
Customer payables		1,358		(1,566)
Other long-term liabilities		8		1,062
	·			
Net cash provided by (used in) operating activities		16,579		(761)
Investing activities				
Purchases of short-term investments		(25,307)		(28,594)
Proceeds from the sale of short-term investments		46,043		21,887
(Decrease) increase in goodwill and intangibles		(41)		12
Cash paid for acquisitions, net of cash acquired		(25,627)		(10,232)

Net cash used in investing activities	(6,174)	(19,209)
Financing activities		
Principal repayments of capital lease obligations and debt	(46)	(64)
Proceeds from exercise of common stock options and warrants (net of tax)	107	748
Incremental tax benefit from exercise of common stock options	18	781
Net proceeds from the issuance of common stock	_	1,046
Net cash provided by financing activities	79	2,511
Effect of exchange rate differences on cash and cash equivalents	184	203
Net increase (decrease) in cash and cash equivalents	10,668	(17,256)
Cash and cash equivalents at beginning of the period	39,954	54,359
Cash and cash equivalents at end of period	\$ 50,622	\$ 37,103
Supplemental disclosure of cash flow information		
Cash paid for income taxes	\$ 9,384	\$ 5,584
Cash paid for interest	\$ 20	\$ 4

See accompanying notes to the unaudited consolidated financial statements.

5

Table of Contents

Liquidity Services, Inc. and Subsidiaries Notes to the Unaudited Consolidated Financial Statements

1. Organization

Liquidity Services, Inc. and subsidiaries (LSI or the Company) is a leading online auction marketplace for wholesale, surplus and salvage assets. LSI enables buyers and sellers to transact in an efficient, automated online auction environment offering over 500 product categories. The Company's marketplaces provide professional buyers access to a global, organized supply of wholesale, surplus and salvage assets presented with digital images and other relevant product information. Additionally, LSI enables its corporate and government sellers to enhance their financial return on excess assets by providing a liquid marketplace and value-added services that integrate sales and marketing, logistics and transaction settlement into a single offering. LSI organizes its products into categories across major industry verticals such as consumer electronics, general merchandise, apparel, scientific equipment, aerospace parts and equipment, technology hardware, and specialty equipment. The Company's online auction marketplaces are www.liquidation.com, www.govliquidation.com, www.govleals.com and www.liquibiz.com. LSI also operates a wholesale industry portal, www.goWholesale.com, that connects advertisers with buyers seeking products for resale and related business services.

2. Summary of Significant Accounting Policies

Unaudited Interim Financial Information

The accompanying unaudited consolidated balance sheet as of June 30, 2008, unaudited consolidated statements of operations for the three- and nine-months ended June 30, 2008 and 2007 and the unaudited statements of cash flows for the nine months ended June 30, 2008 and 2007 have been prepared in accordance with generally accepted accounting principles for interim financial information. Accordingly, they do not include all of the information and notes required by generally accepted accounting principles for complete financial statements. In the opinion of management, all adjustments, consisting of normal, recurring adjustments, considered necessary for a fair presentation have been included. The information disclosed in the notes to the consolidated financial statements for these periods is unaudited. Operating results for the three- and nine-months ended June 30, 2008 are not necessarily indicative of the results that may be expected for the year ending September 30, 2008 or any future period.

Short-Term Investments

Available-for-sale securities are stated at fair value, with the unrealized gains and losses reported in accumulated other comprehensive income. For the three- and nine-months ended June 30, 2008 and 2007, the amount of unrealized losses reported in accumulated other comprehensive income was \$60,000 and \$171,000, and \$5,000 and \$1,000, respectively.

Stock-Based Compensation

The Company accounts for stock-based compensation under Statement 123(R). The Company's income before provision for income taxes and net income for the three- and nine-months ended June 30, 2008 and 2007 was approximately \$1,177,000 and \$695,000, and \$3,440,000 and \$2,030,000; and \$525,000 and \$310,000, and \$1,408,000 and \$831,000 lower, respectively, than if it had continued to account for share-based compensation under APB Opinion No. 25. The total compensation cost related to nonvested awards not yet recognized at June 30, 2008 was approximately \$12,356,000, which is being recognized over the weighted average vesting period of 33 months. The Company utilizes the Black-Scholes option pricing model to determine its Statement 123(R) expense. Inputs into the Black-Scholes model include volatility rates that ranged from 40% to 65%, dividend rate of 0%, and risk-free interest rates that ranged from 2.52% to 5.05% since October 1, 2005. The Company anticipates a forfeiture rate ranging from 11.4% to 33.4% based on its historical forfeiture rate. As a result of adopting Statement 123(R) on October 1, 2005, the Company's basic and diluted earnings per share for the three- and nine-months ended June 30, 2008 and 2007 were approximately \$0.02 and \$0.02, and \$0.07; and \$0.01 and \$0.01, and \$0.03 and \$0.03, respectively, lower than if it had continued to account for share-based compensation under APB Opinion No. 25.

Liquidity Services, Inc. and Subsidiaries Notes to the Unaudited Consolidated Financial Statements – (Continued)

Earnings per Share

Basic net income attributable to common stockholders per share is computed by dividing net income attributable to common stockholders by the weighted average number of common shares outstanding for the period. Diluted net income attributable to common stockholders per share includes the potential dilution that could occur if securities or other contracts to issue common stock were exercised or converted into common stock. We issued 10,292 restricted shares at a price of \$11.66 during the three months ended June 30, 2008, which we have not included in our calculation of diluted income per share because they resulted in an anti-dilutive effect due to the significant difference between the issuance price and the average market price for the three months ended June 30, 2008. We have also not included the following stock options in our calculation of diluted income per share because the option exercise prices were greater than the average market prices for the applicable periods:

- (a) for the three months ended June 30, 2008, 2,944,100 options;
- (b) for the nine months ended June 30, 2008, 2,335,168 options;
- (c) for the three months ended June 30, 2007, 19,500 options; and
- (d) for the nine months ended June 30, 2007, 19,500 options.

The following summarizes the potential outstanding common stock of the Company as of the dates set forth below:

		Three Months Ended June 30,				Nine Months I	June 30,	
		2008		2007		2008		2007
					ıdited)			
	(dollars in thousands, except per share amounts)							
Weighted average shares calculation:								
Basic weighted average shares outstanding		27,964,662		27,857,115		27,953,526		27,720,937
Treasury stock effect of options and restricted stock		272,488		464,280		248,462		494,752
Diluted weighted average common shares outstanding		28,237,150		28,321,395		28,201,988		28,215,689
			_		_		_	
Net income	\$	3,847	\$	3,053	\$	8,857	\$	7,840
Net income per common share:			_		_		_	
Basic income per common share	\$	0.14	\$	0.11	\$	0.32	\$	0.28
Diluted income per common share	\$	0.14	\$	0.11	\$	0.32	\$	0.28
			_				_	

3. Defense Reutilization and Marketing Service Contracts and U.K. Disposal Service Agency Contract

Defense Reutilization and Marketing Service (DRMS) Contracts

The Company's Surplus Contract (the Surplus Contract) with DRMS expires in December 2008. The Company responded to a Request For Proposal (RFP) from the DRMS regarding a renewal of the Surplus Contract, and has been awarded the contract. Under the terms of the Surplus Contract, the Company acquires surplus government property from DRMS at a fixed percentage of approximately 3.26% of the property's original estimated acquisition value. The Company is required to purchase all surplus government property referred to it by DRMS. The Company then markets the property through its buyer network. Under the terms of the current Surplus Contract, the Company distributes to DRMS a fixed percentage of the profits realized from the ultimate sale of the inventory, after deduction for allowable expenses and profit-sharing distributions, as provided for under the terms of the contract. Under the new Surplus contract, which is expected to begin during the first half of Fiscal Year 2009, the Company will not be required to distribute any portion of the profits realized under the Contract, as the new Contract structure requires a fixed percentage price of approximately 3.26% of the DRMS' acquisition value to be paid for the property.

As a result of the Surplus Contract, the Company is the sole remarketer of all U.S. Department of Defense (DoD) surplus turned into DRMS available for sale within the United States, Puerto Rico, and Guam.

7

Table of Contents

Liquidity Services, Inc. and Subsidiaries Notes to the Unaudited Consolidated Financial Statements – (Continued)

The Company's Scrap Contract (the Scrap Contract) with DRMS expires in June 2012. Under the terms of the Scrap Contract, the Company is required to purchase all scrap government property referred to it by DRMS. As a result of this contract, the Company is the sole remarketer of all DoD scrap turned into DRMS available for sale within the United States, Puerto Rico, and Guam.

The contracts may be terminated by either the Company or DRMS if the rate of return performance ratio does not exceed specified benchmark ratios for two consecutive quarterly periods and the preceding twelve months. The Company has performed in excess of the benchmark ratios throughout the contract period through June 30, 2008.

Based on the sales price of the inventory, after reduction for allowable expenses and other disbursements under the Surplus Contract with DRMS, the Company was required to disburse to DRMS 78.2%, and to Kormendi/Gardener Partners (KGP), 1.8% of the profits from the sale of goods under this contract. In addition, disbursements to DRMS/KGP are only required to the extent the Company has distributable cash surplus, as defined under the contract. This generally means that the Company is only required to disburse funds to the extent cash on hand at the Company's subsidiary, Government

Liquidation.com (GL), exceeds the sum of outstanding working capital advances, management's estimated accrued liabilities, contingent liabilities, and estimated operating expenses for the upcoming month. On September 12, 2006, the DoD agreed to increase the profit-sharing distribution for the Surplus Contract in exchange for the Company's agreement to implement additional inventory assurance processes and procedures with respect to the sale of demilitarized property. From August 1, 2006 until November 30, 2006, the Company was entitled to receive 27.5% of the profits and DRMS was entitled to 72.5% of the profits from the sale of goods under this contract. For property received from November 30, 2006 through June 18, 2008, the Company has been entitled to receive between 25% and 30.5% of the profits, based on the results of an audit of the effectiveness of the inventory controls the Company implemented under the contract modification, which is referred to as the Surplus Contract incentive. This incentive is measured quarterly during fiscal year 2008. On June 1, 2007, the Company agreed, as provided in the modification to the Surplus Contract that became effective as of September 12, 2006, to provide additional value-added services with respect to demilitarized property that is returned to the DoD for reutilization. In exchange for the agreement to provide these services, the DoD exercised its existing option to increase the Company's share of net proceeds under the Surplus Contract by 1%. On May 13, 2008, the DoD agreed to extend the current Surplus Contract through December 19, 2008, as well as increase the Company's share of net proceeds under the Surplus Contract to 39.5% on property received after June 18, 2008. Profit-sharing distributions to DRMS/KGP under the Surplus Contract for the three- and nine-months ended June 30, 2008 and 2007 were \$10,892,000, \$30,512,000, \$7,418,000 and \$24,138,000, respectively, including accrued amounts, as of June 30, 2008 and 2007, of \$3,777,000 and \$2,578,000, respectively.

Under the original terms of the Scrap Contract, the Company was required to disburse to DRMS 78.2%, and to KGP 1.8% of the profits realized from the ultimate sale of the inventory, after deduction for allowable expenses, calculated in a similar manner to that of the Surplus Contract. Under the Scrap Contract, the Company also has a performance incentive that allows it to receive up to an additional 2% of the profit sharing distribution. This incentive is measured annually on June 30th, and is applied to the prior 12 months. The Company earned a performance incentive for the 12 months ended June 30, 2008, of approximately \$1,447,000, in the quarter ended June 30, 2008. On May 21, 2007, the DoD agreed to increase the profit-sharing distribution, to the Company under the Scrap Contract, from 20% to 23% effective June 1, 2007, in exchange for the Company's agreement to implement additional inventory assurance processes and procedures with respect to the mutilation of demilitarized scrap property sold by the Company. For the three- and nine-months ended June 30, 2008 and 2007, profit-sharing distributions to the DRMS under the Scrap Contract amounted to \$12,838,000, \$35,797,000, \$9,332,000 and \$27,529,000, respectively, including accrued amounts, as of June 30, 2008 and 2007, of \$4,173,000 and \$3,573,000, respectively.

U.K. Disposal Services Agency (DSA) Contract

Under the contract with the DSA, the Company is required to disburse to DSA a percentage that varies based on the total annual sales volume. Distributions to DSA for the three- and nine-months ended June 30, 2008 and 2007 were \$470,000, \$1,334,000, \$583,000 and \$1,779,000, respectively, including accrued amounts, as of June 30, 2008 and 2007, of \$65,000.

8

Table of Contents

Liquidity Services, Inc. and Subsidiaries Notes to the Unaudited Consolidated Financial Statements – (Continued)

4. Acquisitions

GovDeals

On January 1, 2008, the Company acquired the stock of Information Management Specialists, Inc. and its GovDeals, Inc. subsidiary for approximately \$9,399,000 in cash net of acquired cash of \$742,000. GovDeals is an online auction marketplace for the sale of surplus and confiscated items for local and state government entities including city, county and state agencies as well as school boards and public utilities. The operating results of GovDeals have been included in the consolidated financial statements from the date of acquisition.

Goodwill was created as part of the acquisition as the Company acquired an experienced and knowledgeable workforce, along with customer lists and non-contractual customer relationships. The purchase consideration was allocated to acquired tangible assets, identifiable intangible assets, liabilities assumed and goodwill as follows:

		sideration .mount
	(in t	housands)
Goodwill	\$	8,316
Cash, receivables and other assets		1,342
Covenants not to compete, brand and technology intangibles		1,095
Property and equipment		162
Deferred tax liability		(423)
Other liabilities		(351)
Total consideration	\$	10,141

Geneva

On May 1, 2008, the Company acquired the stock of the companies in the Geneva Group (Geneva), including Geneva Industries Ltd., Willen Trading Ltd., and Geneva Auctions Ltd., for approximately \$16,238,000 million in cash net of acquired cash, of \$925,000. In addition, the Company may be required to pay contingent earn-out payments of up to an aggregate of approximately \$2.9 million payable over the next three years, which will be recorded when the Company determines that such payments are probable. The Geneva Group is a leading United Kingdom (UK) based remarketer of reverse supply chain merchandise, including customer returns and overstock merchandise. The operating results of Geneva have been included in the consolidated financial statements from the date of acquisition.

Goodwill was created as part of the acquisition as the Company acquired an experienced and knowledgeable workforce, along with customer lists and non-contractual customer relationships. The purchase consideration was allocated to acquired tangible assets, identifiable intangible assets, liabilities assumed and goodwill as follows:

	 Amount (in thousands)
Goodwill	\$ 16,754
Cash, receivables and other assets	1,829
Inventory	1,475
Covenants not to compete and technology intangibles	1,298
Property and equipment	233
Deferred tax liability	(364)
Other liabilities	(4,062)
Total consideration	\$ 17,163

9

Table of Contents

Liquidity Services, Inc. and Subsidiaries Notes to the Unaudited Consolidated Financial Statements – (Continued)

5. Intangible Assets

Intangible assets at June 30, 2008 consisted of the following:

	Useful Life (in years)	Gross Carrying Amount (dollars in	thous	Accumulated Amortization & Foreign Currency Translation Adjustment	Net Carrying Amount
Contract intangible	7	\$ 5,694	\$	(2,372)	\$ 3,322
Brand and technology	3 - 5	688		(63)	625
Covenants not to compete	3 - 5	2,385		(401)	1,984
Patent and trademarks	3 - 10	183		(26)	157
Total intangible assets, net					\$ 6,088

Future expected amortization of intangible assets at June 30, 2008 was as follows:

Years ending September 30,	(in t	housands)
2008 (remaining three months)	\$	408
2009		1,633
2010		1,619
2011		1,403
2012 and after		1,025

6. Debt

Senior Credit Facility

In December 2002, and as subsequently amended, the Company entered into a senior credit facility (the Agreement) with a bank, which provides for borrowings up to \$30.0 million. This senior credit facility will expire in March 2010.

Borrowings under the Agreement bear interest at an annual rate equal to the LIBOR rate plus 1.5% (4.811% at June 30, 2008) due monthly. As of June 30, 2008 and September 30, 2007, the Company had no outstanding borrowings under the Agreement.

Borrowings under the Agreement are secured by substantially all of the assets of the Company. The Agreement contains certain financial and non-financial restrictive covenants including, among others, the requirements to maintain a minimum level of earnings before interest, income taxes, depreciation and amortization (EBITDA). As of June 30, 2008, the Company was in compliance with these covenants.

Note Payable

Debt consisted of the following:

	2	mber 30, 2007 ousands)
Note payable—other	\$	48
Less: unamortized debt discount		(6)
Subtotal		42
Less: current portion of long-term debt		(13)
Long-term portion debt	\$	29

Liquidity Services, Inc. and Subsidiaries Notes to the Unaudited Consolidated Financial Statements – (Continued)

7. Income Taxes

The Company's interim effective income tax rate is based on management's best current estimate of the expected annual effective income tax rate. Based on current projections of taxable income for the year ending September 30, 2008, the Company expects that it will have an effective income tax rate of 41%.

As of June 30, 2008, the Company's deferred tax assets exceeded its deferred tax liabilities. The Company had a net deferred tax asset of approximately \$3.4 million at June 30, 2008.

The Company adopted the provisions of FASB Interpretation No. 48, *Accounting for Uncertainty in Income Taxes* ("FIN 48") on October 1, 2007. The adoption of FIN 48 did not impact the Company's financial position or results of operations. The Company has concluded that there were no uncertain tax positions identified during its analysis. The Company's policy is to recognize interest and penalties in the period in which they occur in the income tax provision. The Company and its subsidiaries file income tax returns in the U.S. federal jurisdiction, various state and local jurisdictions and in foreign jurisdictions, primarily the UK and Germany. The Company is no longer subject to U.S. federal, state and local, or foreign income tax examinations by tax authorities for years before fiscal 2004, although carryforward of tax attributes that were generated prior to fiscal 2004 may still be adjusted upon examination by tax authorities if they are utilized.

The Company is not currently under audit for income taxes in any jurisdiction.

8. Stockholders' Equity

Common Stock

On February 23, 2006, the Company issued 5,000,000 shares of common stock for net proceeds of \$43,977,000 in conjunction with its initial public offering. On March 13, 2007, the Company issued 100,000 shares of common stock for net proceeds of \$1,070,000 in conjunction with its follow-on offering.

2006 Omnibus Long-Term Incentive Plan (the 2006 Plan)

5,000,000 shares of common stock were initially reserved for issuance under the 2006 Plan. During fiscal year 2006, the Company issued options to purchase 1,208,000 shares to employees and directors with exercise prices between \$12.89 and \$17.63 and options to purchase 97,000 shares were forfeited. At September 30, 2006, there were 3,889,000 shares remaining reserved for issuance in connection with awards under the 2006 Plan. During fiscal year 2007, the Company issued options to purchase 497,585 shares to employees and directors with exercise prices between \$12.49 and \$20.88 and options to purchase 67,814 shares were forfeited. At September 30, 2007, there were 3,459,229 shares remaining reserved for issuance in connection with awards under the 2006 Plan. During the nine months ended June 30, 2008, the Company issued options to purchase 1,844,232 shares to employees and directors with exercise prices between \$8.40 and \$13.48, and options to purchase 378,692 shares were forfeited. During the three months ended June 30, 2008, the Company issued 10,292 restricted shares at a price of \$11.66. At June 30, 2008, there were 1,993,689 shares remaining reserved for issuance in connection with awards under the 2006 Plan. The maximum number of shares subject to options or stock appreciation rights that can be awarded under the 2006 Plan to any person is 1,000,000 per year. The maximum number of shares that can be awarded under the 2006 Plan to any person, other than pursuant to an option or stock appreciation right, is 700,000 per year.

11

Table of Contents

Liquidity Services, Inc. and Subsidiaries Notes to the Unaudited Consolidated Financial Statements – (Continued)

Stock Option Activity

A summary of the Company's stock option activity for the year ended September 30, 2007 and the three months ended December 31, 2007, March 31, 2008 and June 30, 2008 is as follows:

	Options	Weighted-Average Exercise Price
Options outstanding at September 30, 2006	1,990,196	\$ 9.48
Options granted	497,585	16.83
Options exercised	(254,451)	4.07
Options canceled	(82,359)	12.07
Options outstanding at September 30, 2007	2,150,971	11.72
Options granted (unaudited)	1,704,904	12.79
Options exercised (unaudited)	(9,889)	5.16
Options canceled (unaudited)	(29,084)	11.95
Options outstanding at December 31, 2007 (unaudited)	3,816,902	12.21
Options granted (unaudited)	45,500	12.63
Options exercised (unaudited)	(12,372)	3.41
Options canceled (unaudited)	(119,117)	13.32
Options outstanding at March 31, 2008 (unaudited)	3,730,913	12.21
Options granted (unaudited)	93,828	10.50
Options exercised (unaudited)	(7,145)	2.39
Options canceled (unaudited)	(243,390)	16.10
Options outstanding at June 30, 2008 (unaudited)	3,574,206	11.92
Options exercisable at June 30, 2008 (unaudited)	1,120,023	10.44

The intrinsic value of outstanding and exercisable options at June 30, 2008 is approximately \$4,409,000 and \$1,221,000, respectively, based on a stock price of \$11.53 on June 30, 2008.

9. Contingencies

In January 2008, KGP commenced litigation against GL and Surplus Acquisition Venture, LLC (SAV), one of the Company's other subsidiaries, seeking \$1.5 million in damages. KGP claims it is entitled to these damages because of actions GL and SAV took at the direction of DRMS pursuant to an amendment to our Surplus Contract entered into in August 2006. GL and SAV have filed a motion to dismiss this litigation in its entirety and believe they have meritorious defenses in this litigation. In addition, SAV and GL believe they likely would be able to recover their costs and damages arising out of this litigation from DRMS under the terms of the Surplus Contract.

10. Subsequent Event

On July 31, 2008 the Company was awarded the new Surplus Contract, which has an initial term of three years and includes two optional one year terms. Under the new Surplus Contract, which is expected to begin during the first half of Fiscal Year 2009, the Company will not be required to distribute any portion of the profits realized under the Contract, as the new Contract structure requires a fixed percentage price of approximately 3.26% of the DRMS' acquisition value to be paid for the property.

12

Table of Contents

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

FORWARD-LOOKING STATEMENTS

This document contains forward-looking statements. These statements are only predictions. The outcome of the events described in these forward-looking statements is subject to known and unknown risks, uncertainties and other factors that may cause our actual results, levels of activity, performance or achievements to differ materially from any future results, levels of activity, performance or achievements expressed or implied by these forward-looking statements. These risks and other factors include but are not limited to the successful completion of the Geneva Group acquisition and our ability to integrate the Geneva Group into our existing operations, continue the Geneva Group's seller relationships and buyer network and realize expected benefits of the acquisition, as well as the factors set forth in our Annual Report on Form 10-K for the year ended September 30, 2007. You can identify forward-looking statements by terminology such as "may," "will," "should," "could," "would," "expects," "intends," "plans," "anticipates," "believes," "estimates," "predicts," "potential," "continues" or the negative of these terms or other comparable terminology. Although we believe that the expectations reflected in the forward-looking statements are reasonable, we cannot guarantee future results, levels of activity, performance or achievements. There may be other factors of which we are currently unaware or deem immaterial that may cause our actual results to differ materially from the forward-looking statements.

All forward-looking statements attributable to us or persons acting on our behalf apply only as of the date of this document and are expressly qualified in their entirety by the cautionary statements included in this document. Except as may be required by law, we undertake no obligation to publicly update or revise any forward-looking statement to reflect events or circumstances occurring after the date of this document or to reflect the occurrence of unanticipated events.

The following discussion should be read in conjunction with our consolidated financial statements and related notes and the information contained elsewhere in this document.

Overview

About us. We are a leading online auction marketplace for wholesale, surplus and salvage assets. We enable buyers and sellers to transact in an efficient, automated online auction environment offering over 500 product categories. Our marketplaces provide professional buyers access to a global, organized supply of wholesale, surplus and salvage assets presented with digital images and other relevant product information. Additionally, we enable our corporate and government sellers to enhance their financial return on excess assets by providing a liquid marketplace and value-added services that integrate sales and marketing, logistics and transaction settlement into a single offering. We organize our products into categories across major industry verticals such as consumer electronics, general merchandise, apparel, scientific equipment, aerospace parts and equipment, technology hardware, and specialty equipment. Our online auction marketplaces are www.liquidation.com, www.govliquidation.com, www.govdeals.com and www.liquibiz.com. We also operate a wholesale industry portal, www.qoWholesale.com, that connects advertisers with buyers seeking products for resale and related business services.

We believe our ability to create liquid marketplaces for wholesale, surplus and salvage assets generates a continuous flow of goods from our corporate and government sellers. This flow of goods in turn attracts an increasing number of professional buyers to our marketplaces. During the last 12 months, the number of registered buyers grew from approximately 649,000 to approximately 948,000, or 46.1%, including GovDeals and Geneva.

Recent initiatives. On May 1, 2008, we acquired the stock of the companies in the Geneva Group, including Geneva Industries Ltd., Willen Trading Ltd., and Geneva Auctions Ltd., for approximately \$16,238,000 million in cash, net of acquired cash of \$925,000 and contingent earn-out payments of up to an aggregate of approximately \$2.9 million payable over the next three years. The Geneva Group is a United Kingdom (UK) based remarketer of reverse supply chain merchandise, including customer returns and overstock merchandise. Geneva serves leading UK retailers and manufactures with a product focus on consumer electronics, technology equipment and hard goods general merchandise. We expect that the acquisition will strengthen our business by adding Geneva's longstanding seller relationships in the European Union (EU) with significant organic growth opportunities, as well as a complementary buyer network of UK based wholesalers and EU exporters developed over the last 18 years. We believe Geneva's buyer base will augment our marketplace by adding international demand for consumer electronics and general merchandise across all condition categories

On May 13, 2008, the DoD agreed to extend the current Surplus Contract through December 19, 2008, as well as increase our share of net proceeds under the Surplus Contract to 39.5% on property received after June 18, 2008. We responded to a RFP from the DRMS regarding a renewal of the Surplus Contract, and have been awarded the contract. Under the new Surplus Contract, which is expected to begin during the first half of Fiscal Year 2009, we will not be required to distribute any portion of the profits realized under the Contract, as the new Contract structure requires a fixed percentage price of approximately 3.26% of the DRMS' acquisition value to be paid for the property.

Table of Contents

Our revenue. We generate substantially all of our revenue by retaining a percentage of the proceeds from the sales we manage for our sellers. We offer our sellers three primary transaction models: a profit-sharing model, a consignment model and a purchase model.

- Profit-sharing model. Under our profit-sharing model, we purchase inventory from our suppliers and share with them a portion of the profits received from a completed sale in the form of a distribution. Distributions are calculated based on the value received from the sale after deducting direct costs, such as sales and marketing, technology and operations and other general and administrative costs. Because we are the primary obligor, and take general and physical inventory risks and credit risk under this transaction model, we recognize as revenue the sale price paid by the buyer upon completion of a transaction. Revenue from our profit-sharing model accounted for approximately 62.2% and 62.5% of our total revenue for the three- and nine-months ended June 30, 2008, respectively. The merchandise sold under our profit-sharing model accounted for approximately 42.6% and 46.5% of our gross merchandise volume, or GMV, for the three- and nine-months ended June 30, 2008, respectively.
- · Consignment model. Under our consignment model, we recognize commission revenue from sales of merchandise in our marketplaces that is owned by others. These commissions, which we refer to as seller commissions, represent a percentage of the sale price the buyer pays upon completion of a transaction. We vary the percentage amount of the seller commission depending on the various value-added services we provide to the seller to facilitate the transaction. For example, we generally increase the percentage amount of the commission if we take possession, handle, ship or provide enhanced product information for the merchandise. We collect the seller commission by deducting the appropriate amount from the sales proceeds prior to their distribution to the seller after completion of the transaction. Revenue from our consignment model accounted for approximately 8.0% and 7.5% of our total revenue for the three- and nine-months ended June 30, 2008, respectively. The merchandise sold under our consignment model accounted for approximately 37.7% and 32.0% of our GMV for the three- and nine-months ended June 30, 2008, respectively.
- Purchase model. Under our purchase model, we offer our sellers a fixed amount or the option to share a portion of the proceeds received from our completed sales in the form of a distribution. Distributions are calculated based on the value we receive from the sale after deducting a required return to us that we have negotiated with the seller. Because we are the primary obligor, and take general and physical inventory risks and credit risk under this transaction model, we recognize as revenue the sale price paid by the buyer upon completion of a transaction. Revenue from our purchase model accounted for approximately 22.3% and 24.7% of our total revenue for the three- and ninemonths ended June 30, 2008, respectively. The merchandise sold under our purchase model accounted for approximately 15.3% and 18.4% of our GMV for the three- and nine-months ended June 30, 2008, respectively.

We collect a buyer premium on substantially all of our transactions under all of our transaction models. Buyer premiums are calculated as a percentage of the sale price of the merchandise sold and are paid to us by the buyer. Buyer premiums are in addition to the price of the merchandise. Under our profit-sharing model, we typically share the proceeds of any buyer premiums with our sellers.

In the three months ended June 30, 2008, we generated less than 2% of our revenue from advertisements on our wholesale industry portals.

Industry trends. We believe there are several industry trends impacting the growth of our business including: (1) the increase in the adoption of the Internet by businesses to conduct e-commerce both in the United States and abroad; (2) product innovation in the retail supply chain that has increased the pace of product obsolescence and, therefore, the supply of surplus assets; (3) the increase in the volume of returned merchandise handled by both online and offline retailers; (4) the increase in government regulations necessitating verifiable recycling and remarketing of surplus assets; and (5) the increase in outsourcing by corporate and government organizations of disposition activities for surplus and end-of-life assets.

14

Table of Contents

Our Seller Agreements

Our DoD agreements. We have three contracts with the DoD pursuant to which we acquire, manage and sell excess property:

- Surplus contract. In June 2001, we were awarded the CV2 contract, a competitive-bid exclusive contract under which we acquire, manage and sell all usable DoD surplus personal property turned into the Defense Reutilization and Marketing Service (DRMS). Surplus property generally consists of items determined by the DoD to be no longer needed, and not claimed for reuse by, any federal agency, such as computers, electronics, office supplies, scientific and medical equipment, aircraft parts, clothing and textiles. Revenue from our Surplus Contract (including buyer premiums) accounted for approximately 29.8% and 30.9% of our total revenue for the three- and nine-months ended June 30, 2008, respectively. The property sold under our Surplus Contract accounted for approximately 20.4% and 23.0% of our GMV for the three- and nine-months ended June 30, 2008, respectively. On May 13, 2008, the DoD agreed to extend the current Surplus Contract through December 19, 2008. We responded to a RFP from the DRMS regarding a renewal of the Surplus Contract, and have been awarded the contract.
- Scrap contract. In June 2005, we were awarded a competitive-bid exclusive contract under which we acquire, manage and sell substantially all scrap property of the DoD turned into the DRMS. Scrap property generally consists of items determined by DoD to have no use beyond their base material content, such as metals, alloys, and building materials. Revenue from our Scrap Contract (including buyer premiums) accounted for approximately 32.4% and 31.6% of our total revenue for the three- and nine-months ended June 30, 2008, respectively. The property sold under our Scrap Contract accounted for approximately 22.2% and 23.5% of our GMV for the three- and nine-months ended June 30, 2008, respectively. We were required to pay \$5.7 million to the DoD in fiscal 2005 for the right to manage the operations and remarket scrap material in connection with the Scrap Contract. The Scrap Contract expires in June 2012, subject to DoD's right to extend it for three additional one-year terms.

Under the Surplus Contract, we are obligated to purchase all DoD surplus property at set prices representing a percentage of the original acquisition cost, which varies depending on the type of surplus property being purchased. Under the Scrap Contract, we acquire scrap property at a per pound price. We were initially entitled to approximately 20% of the profits of sale (defined as gross proceeds of sale less allowable operating expenses) under the Contracts, and the DoD was entitled to approximately 80% of the profits. We refer to these disbursement payments to DoD as profit-sharing distributions. As a result of these arrangements, we recognize as revenue the gross proceeds from these sales. DoD also reimburses us for actual costs incurred for packing, loading and shipping property under the contracts that we are obligated to pick up from non-DoD locations. On September 12, 2006, we entered into a bilateral contract modification under which the DoD agreed to increase our profit-sharing percentage under the Surplus Contract in exchange for our agreement to implement additional inventory assurance processes and procedures with respect to the sale of demilitarized property. Under the terms of the contract modification, from August 1, 2006 until November 30, 2006, we were entitled to receive 27.5% of the profits under the Surplus Contract and the DoD was entitled to receive 72.5%. For property received from November 30, 2006 through June 18, 2008, we have been entitled to receive between 25% and 30.5% of the profits, based on the results of an audit of the effectiveness of the inventory controls we implemented under the contract modification, which is referred to as the Surplus Contract incentive. This incentive is measured quarterly during fiscal year 2008. On June 1, 2007, we agreed, as provided in the modification to the Surplus Contract that became effective as of September 12, 2006, to provide additional value-added services with respect to demilitarized property that is returned to the DoD for reutilization. In exchange for our agreement to provide these services, the DoD exercised its existing option to increase our share of net proceeds under the Surplus Contract by 1%. On May 13, 2008, the DoD agreed to extend the current Surplus Contract through December 19, 2008, as well as increase our share of net proceeds under the Surplus Contract to 39.5% on property received after June 18, 2008.

Under the Scrap Contract, we also have a small business performance incentive based on the number of scrap buyers that are small businesses that allows us to receive up to an additional 2% of the profit sharing distribution. On May 21, 2007, we entered into a bilateral contract modification under which the DoD agreed to increase the profit-sharing distribution for the Scrap Contract from 20% to 23% effective June 1, 2007, in exchange for our agreement to implement additional inventory assurance processes and procedures with respect to the mutilation of demilitarized scrap property sold.

In January 2006, we were awarded a contract to purchase DoD surplus property located in Germany. This contract generated less than 1% of our revenue in the three months ended June 30, 2008. This contract expires in January 2009.

Our UK MoD agreement. In July 2003, we were awarded a contract to manage and sell surplus property from the United Kingdom Ministry of Defence. This contract generated less than 2% of our revenue in the three months ended June 30, 2008. This contract expires in July 2009, subject to the Ministry's right to extend the contract for an additional one-year term.

Our commercial agreements. We have over 340 corporate clients each of which sells in excess of \$10,000 of wholesale, surplus and salvage assets in our marketplaces. Our agreements with these clients are generally terminable at will by either party.

15

Table of Contents

Key Business Metrics

Our management periodically reviews certain key business metrics for operational planning purposes and to evaluate the effectiveness of our operational strategies, allocation of resources and our capacity to fund capital expenditures and expand our business. These key business metrics include:

Gross merchandise volume. Gross merchandise volume, or GMV, is the total sales value of all merchandise sold through our marketplaces during a given period. We review GMV because it provides a measure of the volume of goods being sold in our marketplaces and thus the activity of those marketplaces. GMV also provides a means to evaluate the effectiveness of investments that we have made and continue to make, including in the areas of customer support, value-added services, product development, sales and marketing, and operations. The GMV of goods sold in our marketplaces during the three- and nine-months ended June 30, 2008 totaled \$104.2 million and \$260.0 million, respectively.

Completed transactions. Completed transactions represents the number of auctions in a given period from which we have recorded revenue. Similar to GMV, we believe that completed transactions is a key business metric because it provides an additional measurement of the volume of activity flowing through our marketplaces. During the three- and nine-months ended June 30, 2008 we completed approximately 108,000 and 262,000 transactions, respectively, which includes six months of GovDeals and two months of Geneva.

Total registered buyers. We grow our buyer base through a combination of marketing and promotional efforts. A person becomes a registered buyer by completing an online registration process on one of our marketplaces. As part of this process, we collect business and personal information, including name, title, company name, business address and contact information, and information on how the person intends to use our marketplaces. Each prospective buyer must also accept our terms and conditions of use. Following the completion of the online registration process, we verify each prospective buyer's e-mail address and confirm that the person is not listed on any banned persons list maintained internally or by the U.S. federal government. After the verification process, which is completed generally within 24 hours, the registration is approved and activated and the prospective buyer is added to our registered buyer list.

Total registered buyers as of a given date represents the aggregate number of persons or entities who have registered on one of our marketplaces. We use this metric to evaluate how well our marketing and promotional efforts are performing. Total registered buyers excludes duplicate registrations, buyers who are suspended from utilizing our marketplaces and those buyers who have voluntarily removed themselves from our registration database. In addition, if we become aware of registered buyers that are no longer in business, we remove them from our database. As of June 30, 2008, we had approximately 948,000 registered buyers, including GovDeals and Geneva.

Total auction participants. For each auction we manage, the number of auction participants represents the total number of registered buyers who have bid one or more times in that auction. As a result, a registered buyer who bids, or participates, in more than one auction is counted as an auction participant in each auction in which he or she participates. Thus, total auction participants for a given period is the sum of the auction participants in each auction conducted during that period. We use this metric to allow us to compare our online auction marketplaces to our competitors, including other online auction sites and traditional on-site auctioneers. In addition, we measure total auction participants on a periodic basis to evaluate the activity level of our base of registered buyers and to measure the performance of our marketing and promotional efforts. For the three- and nine-months ended June 30, 2008, approximately 499,000 and 1,285,000 total auction participants participated in auctions on our marketplaces, respectively, which includes six months of GovDeals and two months of Geneva.

Non-GAAP Financial Measures

EBITDA and adjusted EBITDA. EBITDA is a supplemental non-GAAP financial measure and is equal to net income plus (a) interest expense (income) and other expense (income), net; (b) provision for income taxes; (c) amortization of contract intangibles; and (d) depreciation and amortization. Our definition of adjusted EBITDA differs from EBITDA because we further adjust EBITDA for stock-based compensation expense.

16

Table of Contents

We believe EBITDA and adjusted EBITDA are useful to an investor in evaluating our performance for the following reasons:

- The amortization of contract intangibles relates to amortization of the Scrap Contract beginning in June 2005. Depreciation and amortization expense primarily relates to property and equipment. Both of these expenses are non-cash charges that have fluctuated significantly over the past five years. As a result, we believe that adding back these non-cash charges to net income is useful in evaluating the operating performance of our business on a consistent basis from year-to-year.
- As a result of varying federal and state income tax rates, we believe that presenting a financial measure that adjusts net income for provision for income taxes is useful to investors when evaluating the operating performance of our business.
- During July 2001, we modified the exercise price of 3,402,794 stock options issued to employees. As a result, we are accounting for the modified stock options from the date of modification to the date the stock options are exercised, forfeited or expire unexercised using variable accounting. Under variable accounting, we revalue compensation costs for the stock options at each reporting period based on changes in the intrinsic value of the stock options. We will continue to revalue compensation costs for the options based on changes in the fair value of our common stock in future periods. As a result, we present a financial measure that adjusts net income and EBITDA for the stock compensation expense that results from the July 2001 modification of these stock options. We believe that it is useful to exclude this expense because it results from a one-time event that requires us to record expense that we are not otherwise required to record in connection with new stock options granted during the same time period. In December 2004, the Financial Accounting Standards Board issued Statement of Financial Accounting Standards (SFAS) No. 123 (revised 2004), *Share-Based Payment*, or Statement 123(R), which is a revision of SFAS No. 123. Statement 123(R) requires all share-based payments to employees, including grants of employee stock options, to be recognized in the income statement based on their estimated fair values. Pro forma disclosure is no longer an alternative. We adopted the provisions of Statement 123(R) on October 1, 2005, using the prospective method. Unvested stock based awards issued prior to October 1, 2005, the date that we adopted the provisions of Statement 123(R), will be accounted for at the date of adoption using the intrinsic value method originally applied to those awards. Accordingly, we believe adjusting net income for this non-cash stock based compensation expense is useful to investors when evaluating the operating performance of our business.
- We believe these measures are important indicators of our operational strength and the performance of our business because they provide a link between profitability and operating cash flow.
- · We also believe that analysts and investors use EBITDA and adjusted EBITDA as supplemental measures to evaluate the overall operating performance of companies in our industry.

Our management uses EBITDA and adjusted EBITDA:

- as measurements of operating performance because they assist us in comparing our operating performance on a consistent basis as they remove the impact of items not directly resulting from our core operations;
- \cdot $\,$ for planning purposes, including the preparation of our internal annual operating budget;
- to allocate resources to enhance the financial performance of our business;
- \cdot to evaluate the effectiveness of our operational strategies; and
- to evaluate our capacity to fund capital expenditures and expand our business.

17

Table of Contents

EBITDA and adjusted EBITDA as calculated by us are not necessarily comparable to similarly titled measures used by other companies. In addition, EBITDA and adjusted EBITDA: (a) do not represent net income or cash flows from operating activities as defined by GAAP; (b) are not necessarily indicative of cash available to fund our cash flow needs; and (c) should not be considered as alternatives to net income, income from operations, cash provided by operating activities or our other financial information as determined under GAAP.

We prepare adjusted EBITDA by adjusting EBITDA to eliminate the impact of items that we do not consider indicative of our core operating performance. You are encouraged to evaluate these adjustments and the reasons we consider them appropriate for supplemental analysis. As an analytical tool, adjusted EBITDA is subject to all of the limitations applicable to EBITDA. Our presentation of adjusted EBITDA should not be construed as an implication that our future results will be unaffected by unusual or non-recurring items.

The table below reconciles net income to EBITDA and adjusted EBITDA for the periods presented.

Three M	Three Months		Nine Months		
Ended Ju	Ended June 30,		Ended June 30,		
2008	2007	2008	2007		
(in thousands) (unaudited)					

Net income	\$ 3,847	\$ 3,053 \$	8,857	\$	7,840
Interest expense (income) and other expense (income), net	(292)	(475)	(1,402)		(1,624)
Provision for income taxes	2,672	2,134	6,176		5,422
Amortization of contract intangibles	203	203	610		610
Depreciation and amortization	584	355	1,436		935
EBITDA	7,014	5,270	15,677		13,183
Stock compensation expense	1,177	526	3,440	,	1,409
Adjusted EBITDA	\$ 8,191	\$ 5,796 \$	19,117	\$	14,592

Critical Accounting Estimates

Our discussion and analysis of our financial condition and results of operations are based upon our consolidated financial statements, which have been prepared in accordance with GAAP. The preparation of these consolidated financial statements requires us to make estimates and judgments that affect the reported amounts of assets, liabilities, revenue and expenses, and related disclosure of contingent assets and liabilities. A "critical accounting estimate" is one which is both important to the portrayal of our financial condition and results and requires management's most difficult, subjective or complex judgments, often as a result of the need to make estimates about the effect of matters that are inherently uncertain. We continuously evaluate our critical accounting estimates. We base our estimates on historical experience and on various other assumptions that are believed to be reasonable under the circumstances, the results of which form the basis for making judgments about the carrying values of assets and liabilities that are not readily apparent from other sources. Actual results may differ from these estimates under different assumptions or conditions.

Revenue recognition. We recognize revenue in accordance with the provisions of Staff Accounting Bulletin No. 104, *Revenue Recognition.* For transactions in our online marketplaces, which generate substantially all of our revenue, we recognize revenue when all of the following criteria are met:

- a buyer submits the winning bid in an auction and, as a result, evidence of an arrangement exists and the sale price has been determined;
- title has passed to a buyer and the buyer has assumed risks and rewards of ownership;
- · for arrangements with an inspection period, the buyer has received the merchandise and has not notified us within that period that it is dissatisfied with the merchandise; and
- · collection is reasonably assured.

18

Table of Contents

Substantially all of our sales are recorded subsequent to payment authorization being received, utilizing credit cards, wire transfers and PayPal, an Internet based payment system, as methods of payments. As a result, we are not subject to significant collection risk, as goods are generally not shipped before payment is received.

Revenue is also evaluated in accordance with EITF 99-19, *Reporting Revenue Gross as a Principal Versus Net as an Agent*, for reporting revenue of gross proceeds as the principal in the arrangement or net of commissions as an agent. In arrangements in which we are deemed to be the primary obligor, bear physical and general inventory risk, and credit risk, we recognize as revenue the gross proceeds from the sale, including buyer's premiums. Arrangements in which we act as an agent or broker on a consignment basis, without taking general or physical inventory risk, revenue is recognized based on the sales commissions that are paid to us by the sellers for utilizing our services; in this situation, sales commissions represent a percentage of the gross proceeds from the sale that the seller pays to us upon completion of the transaction.

We have evaluated our revenue recognition policy related to sales under our profit-sharing model and determined it is appropriate to account for these sales on a gross basis using the criteria outlined in EITF 99-19. The following factors were most heavily relied upon in our determination:

- · We are the primary obligor in the arrangement.
- · We are the seller in substance and in appearance to the buyer; the buyer contacts us if there is a problem with the purchase. Only we and the buyer are parties to the sales contract and the buyer has no recourse to the supplier. If the buyer has a problem, he or she looks to us, not the supplier.
- The buyer does not and cannot look to the supplier for fulfillment or for product acceptability concerns.
- We have general inventory risk.
- · We take title to the inventory upon paying the amount set forth in the contract with the supplier. Such amount is generally a percentage of the supplier's original acquisition cost and varies depending on the type of the inventory purchased.
- We are at risk of loss for all amounts paid to the supplier in the event the property is damaged or otherwise becomes unsaleable. In addition, as payments made for inventory are excluded from the calculation for the profit-sharing distribution under our DoD contracts, we effectively bear inventory risk for the full amount paid to acquire the property (*i.e.*, there is no sharing of inventory risk).

Valuation of goodwill and other intangible assets. In accordance with SFAS No. 141, Business Combinations, we identify and value intangible assets that we acquire in business combinations, such as customer arrangements, customer relationships and non-compete agreements, that arise from contractual or other legal rights or that are capable of being separated or divided from the acquired entity and sold, transferred, licensed, rented or exchanged. The fair value of identified intangible assets is based upon an estimate of the future economic benefits expected to result from ownership, which represents the amount at which the assets could be bought or sold in a current transaction between willing parties, that is, other than in a forced or liquidation sale.

In accordance with SFAS No. 142, *Goodwill and Other Intangible Assets*, we test our goodwill and other intangible assets for impairment annually or more frequently if events or circumstances indicate impairment may exist. Examples of such events or circumstances could include a significant change in business climate or a loss of significant customers. We apply a two-step fair value-based test to assess goodwill for impairment. The first step compares the fair value of a reporting unit to its carrying amount, including goodwill. If the carrying amount of the reporting unit exceeds its fair value, the second step is then performed. The second step compares the carrying amount of the reporting unit's goodwill to the fair value of the goodwill. If the fair value of the goodwill is less than the carrying amount, an impairment loss would be recorded in our statements of operations. Intangible assets with definite lives are amortized over their estimated useful lives and are also reviewed for impairment if events or changes in circumstances indicate that their carrying amount may not be realizable.

Our management makes certain estimates and assumptions in order to determine the fair value of net assets and liabilities, including, among other things, an assessment of market conditions, projected cash flows, cost of capital and growth rates, which could significantly impact the reported value of goodwill and other intangible assets. Estimating future cash flows requires significant judgment, and our projections may vary from cash flows eventually realized. The valuations employ a combination of present value techniques to measure fair value, corroborated by comparisons to estimated market multiples. These valuations are based on a discount rate determined by our management to be consistent with industry discount rates and the risks inherent in our current business model.

19

Table of Contents

We cannot predict the occurrence of certain future events that might adversely affect the reported value of goodwill and other intangible assets, which totaled \$42.6 million at June 30, 2008. Such events may include strategic decisions made in response to economic and competitive conditions, the impact of the economic environment on our base of buyers and sellers or material negative changes in our relationships with material customers.

Income taxes. We account for income taxes in accordance with SFAS No. 109, *Accounting for Income Taxes*. This statement requires an asset and liability approach for measuring deferred taxes based on temporary differences between the financial statement and income tax bases of assets and liabilities existing at each balance sheet date using enacted tax rates for the years in which the taxes are expected to be paid or recovered. A valuation allowance is provided to reduce the deferred tax assets to a level that we believe will more likely than not be realized. The resulting net deferred tax asset reflects management's estimate of the amount that will be realized.

We provide for income taxes based on our estimate of federal and state tax liabilities. These estimates include, among other items, effective rates for state and local income taxes, estimates related to depreciation and amortization expense allowable for tax purposes, and the tax deductibility of certain other items. Our estimates are based on the information available to us at the time we prepare the income tax provision. We generally file our annual income tax returns several months after our fiscal year-end. Income tax returns are subject to audit by federal, state and local governments, generally years after the returns are filed. These returns could be subject to material adjustments or differing interpretations of the tax laws.

Stock-based compensation. We account for stock-based compensation in accordance with SFAS No. 123 (revised 2004), Share-Based Payment (Statement 123(R)), which is a revision of SFAS No. 123. Statement 123(R) requires all share-based payments to employees, including grants of employee stock options, to be recognized in the income statement based on their estimated fair values. We use the Black-Scholes option pricing model to estimate the fair values of share-based payments.

The above list is not intended to be a comprehensive list of all of our accounting estimates. In many cases, the accounting treatment of a particular transaction is specifically dictated by GAAP, with little need for management's judgment in their application. There are also areas in which management's judgment in selecting any available alternative would not produce a materially different result. See our audited financial statements and related notes, which contain accounting policies and other disclosures required by GAAP.

Components of Revenue and Expenses

Revenue. We generate substantially all of our revenue from sales of merchandise held in inventory and by retaining a percentage of the proceeds from the sales. Our revenue recognition practices are discussed in more detail in the section above entitled "*Critical Accounting Estimates*."

Cost of goods sold (excluding amortization). Cost of goods sold includes the costs of purchasing and transporting property for auction, as well as credit card transaction fees.

Profit-sharing distributions. Our two primary contracts with the DoD are structured as profit-sharing arrangements in which we purchase and take possession of all goods we receive from the DoD at a contractual percentage of the original acquisition cost of those goods. After deducting allowable operating expenses, we disburse to the DoD on a monthly basis a percentage of the profits of the aggregate monthly sales. We retain the remaining percentage of these profits after the DoD's disbursement. We refer to these disbursement payments to DoD as profit-sharing distributions.

Technology and operations. Technology expenses consist primarily of personnel costs related to our programming staff who develop and deploy new marketplaces, such as *liquibiz.com*, and continuously enhance existing marketplaces. These personnel also develop and upgrade the software systems that support our operations, such as sales processing. Because our marketplaces and support systems require frequent upgrades and enhancements to maintain viability, we have determined that the useful life for substantially all of our internally developed software is less than one year. As a result, we expense these costs as incurred.

Operations expenses consist primarily of operating costs, including buyer relations, shipping logistics and distribution center operating costs.

20

Table of Contents

Sales and marketing. Sales and marketing expenses include the cost of our sales and marketing personnel as well as the cost of marketing and promotional activities. These activities include online marketing campaigns such as paid search advertising.

General and administrative. General and administrative expenses include all corporate and administrative functions that support our operations and provide an infrastructure to facilitate our future growth. Components of these expenses include executive management and staff salaries, bonuses and related taxes and employee benefits; travel; headquarters rent and related occupancy costs; and legal and accounting fees. The salaries, bonus and employee benefits costs included as general and administrative expenses are generally more fixed in nature than our operating expenses and do not vary directly with the volume of merchandise sold through our marketplaces.

Amortization of contract intangibles. Amortization of contract intangibles expense consists of the amortization of our DoD Scrap Contract award during June 2005. This contract required us to purchase the rights to operate the scrap operations of the DoD during the seven year base term of the contract. The intangible asset created from the \$5.7 million purchase is being amortized over 84 months on a straight-line basis. The amortization period is correlated to the base term of the contract, exclusive of renewal periods.

Depreciation and amortization. Depreciation and amortization expenses consist primarily of the depreciation and amortization of amounts recorded in connection with the purchase of furniture, fixtures and equipment.

Interest income and expense and other income, net. Interest income and expense and other income, net consists primarily of interest income on cash and short-term investments and interest expense on borrowings under our notes payable and realized gains or losses on short-term investments.

Income taxes. During fiscal years 2006 and 2007, we had an effective income tax rate of approximately 40%, which included federal and state income taxes. We estimate that our future effective income tax rate will be approximately 41%.

21

Table of Contents

Results of Operations

The following table sets forth, for the periods indicated, selected statement of operations data expressed as a percentage of revenue.

	Three Months Ended June 30,		Nine Months Ended June 30,	
	2008	2007	2008	2007
Revenue	100.0%	100.0%	100.0%	100.0%
Costs and expenses:				
Cost of goods sold (excluding amortization)	27.4	25.3	26.4	22.7
Profit-sharing distributions	33.9	33.4	34.9	36.9
Technology and operations	14.6	15.5	15.8	16.6
Sales and marketing	6.3	6.8	6.5	6.6
General and administrative	8.2	9.0	8.2	8.3
Amortization of contract intangibles	0.4	0.4	0.4	0.4
Depreciation and amortization	0.6	0.6	0.7	0.6
Total costs and expenses	91.4	91.0	92.9	92.1
Income from operations	8.6	9.0	7.1	7.9
Interest income (expense) and other income, net	0.5	0.9	0.7	1.1
Income before provision for income taxes	9.1	9.9	7.8	9.0
Provision for income taxes	(3.7)	(4.1)	(3.2)	(3.7)
Net income	5.4%	5.8%	4.6%	5.3%

Three Months Ended June 30, 2008 Compared to Three Months Ended June 30, 2007

Revenue. Revenue increased \$19.0 million, or 36.1%, to \$71.5 million for the three months ended June 30, 2008 from \$52.5 million for the three months ended June 30, 2007, primarily due to an increase in the number of completed transactions, which increased from approximately 54,000 to 108,000, or 100.7%. The amount of gross merchandise volume increased \$41.9 million, or 67.2%, to \$104.2 million for the three months ended June 30, 2008 from \$62.3 million for the three months ended June 30, 2007, primarily due to (1) our scrap business, which generated 32.4% of our revenue and 22.2% of our gross merchandise volume for the three months ended June 30, 2008, grew 50.9%; (2) our surplus business, which generated 29.8% of our revenue and 20.4% of our gross merchandise volume for the three months ended June 30, 2008, grew 35.0%; (3) the acquisition of GovDeals, completed on January 1, 2008, which generated 2.1% of our revenue and 20.0% of our gross merchandise volume for the three months ended June 30, 2008; and (4) the acquisition of Geneva, which was completed on May 1, 2008. We also benefited from our ability to more effectively market assets to potential buyers; our marketing efforts resulted in an approximate 46.1% increase in registered buyers to approximately 948,000 at June 30, 2008 from approximately 649,000 at June 30, 2007, including GovDeals and Geneva.

Cost of goods sold (excluding amortization). Cost of goods sold (excluding amortization) increased \$6.3 million, or 47.1%, to \$19.6 million for the three months ended June 30, 2008 from \$13.3 million for the three months ended June 30, 2007. As a percentage of revenue, cost of goods sold (excluding amortization) increased to 27.4% from 25.3%. These increases are primarily due to (1) an increase in the volume of goods sold by existing sellers utilizing our purchase model; (2) a mix shift to apparel items, which realized a lower margin during the three months ended June 30, 2008; and (3) the acquisition of Geneva, which was completed on May 1, 2008 and utilizes the purchase model.

Profit-sharing distributions. Profit-sharing distributions increased \$6.6 million, or 37.8%, to \$24.2 million for the three months ended June 30, 2008 from \$17.6 million for the three months ended June 30, 2007. As a percentage of revenue, profit-sharing distributions increased to 33.9% from 33.4%. These increases are primarily due to 35.0% growth in our surplus business and 50.9% growth in our scrap business.

Technology and operations expenses. Technology and operations expenses increased \$2.3 million, or 28.1%, to \$10.4 million for the three months ended June 30, 2008 from \$8.1 million for the three months ended June 30, 2007, primarily due to (1) the addition of 74 technology and operations personnel, the majority of whom were needed to support the increased volume of transactions and merchandise discussed above; (2) the acquisition of GovDeals, which was completed on January 1, 2008; and (3) the acquisition of Geneva, which was completed on May 1, 2008. As a percentage of revenue, these expenses decreased to 14.6% from 15.5%, primarily due to our growth in revenue, while leveraging our fixed expenses, such as programming personnel.

22

Table of Contents

Sales and marketing expenses. Sales and marketing expenses increased \$0.9 million, or 25.7%, to \$4.5 million for the three months ended June 30, 2008 from \$3.6 million for the three months ended June 30, 2007, primarily due to (1) our hiring of 23 additional sales and marketing personnel; (2) \$0.5 million in increased expenditures on marketing and promotional activities across our marketplaces; and (3) the acquisition of GovDeals, which was completed on January 1, 2008. As a percentage of revenue, these expenses decreased to 6.3% from 6.8%, primarily due to our growth in revenue, while leveraging our fixed expenses, such as marketing personnel.

General and administrative expenses. General and administrative expenses increased \$1.1 million, or 23.9%, to \$5.8 million for the three months ended June 30, 2008 from \$4.7 million for the three months ended June 30, 2007, primarily due to (1) expenses of \$0.2 million related to the adoption of Statement 123(R); (2) costs of \$0.1 million for travel related expenses associated with business development efforts; (3) expenses of \$0.2 million associated with Geneva, which was acquired on January 1, 2008; and (4) expenses of \$0.3 million associated with Geneva, which was acquired on May 1, 2008. As a percentage of revenue, these expenses decreased to 8.2% from 9.0%, primarily due to our growth in revenue, while leveraging our fixed expenses, such as corporate staff.

Amortization of contract intangibles. Amortization of contract intangibles was \$0.2 million for the three months ended June 30, 2008 and 2007, as a result of our DoD Scrap Contract award during June 2005. This contract required us to purchase the rights to operate the scrap operations of the DoD during the seven-year base term of the contract. The intangible asset created from the \$5.7 million purchase is being amortized on a straight-line basis over 84 months, which began in August 2005.

Depreciation and amortization expenses. Depreciation and amortization expenses increased \$0.2 million, or 64.8%, to \$0.6 million for the three months ended June 30, 2008 from \$0.4 million for the three months ended June 30, 2007, primarily due to additional depreciation expense resulting from the purchase of \$2.7 million of property and equipment during the fiscal year ended September 30, 2007.

Interest income and expense and other income, net. Interest income and expense and other income, net decreased \$0.2 million, or 38.4%, to \$0.3 million for the three months ended June 30, 2008 from \$0.5 million for the three months ended June 30, 2007, primarily due to (1) a reduction in the cash balance due to the recent GovDeals and Geneva acquisitions and (2) a reduction in short term interest rates.

Provision for income tax expense. Income tax expense increased \$0.6 million, or 25.2%, to \$2.7 million for the three months ended June 30, 2008 from \$2.1 million for the three months ended June 30, 2007, primarily due to the increase in income before provision for income taxes.

Net income. Net income increased \$0.8 million, or 26.0%, to \$3.8 million for the three months ended June 30, 2008 from \$3.0 million for the three months ended June 30, 2007.

Nine Months Ended June 30, 2008 Compared to Nine Months Ended June 30, 2007

Revenue. Revenue increased \$46.6 million, or 31.7%, to \$193.6 million for the nine months ended June 30, 2008 from \$147.0 million for the nine months ended June 30, 2007, primarily due to an increase in the number of completed transactions from approximately 155,000 to 262,000, or 68.8%. The amount of gross merchandise volume increased \$84.5 million, or 48.1%, to \$260.0 million for the nine months ended June 30, 2008 from \$175.5 million for the nine months ended June 30, 2007, primarily due to (1) our scrap business, which generated 31.6% of our revenue and 23.5% of our gross merchandise volume for the nine months ended June 30, 2008, grew 47.1%; and (2) the acquisition of GovDeals completed on January 1, 2008, which generated 1.4% of our revenue and 14.1% of our gross merchandise volume for the nine months ended June 30, 2008. During the last 12 months, we also benefited from our ability to more effectively market assets to potential buyers as we gained transaction experience and industry knowledge in the vertical product segments auctioned through our marketplaces.

Cost of goods sold (excluding amortization). Cost of goods sold (excluding amortization) increased \$17.8 million, or 53.5%, to \$51.1 million for the nine months ended June 30, 2008 from \$33.3 million for the nine months ended June 30, 2007. As a percentage of revenue, cost of goods sold (excluding amortization) increased to 26.4% from 22.7%. These increases are primarily due to (1) an increase in the volume of goods sold by existing sellers utilizing our purchase model; (2) a ramp up in volume with existing sales programs which resulted in lower inventory turnover; and (3) a mix shift to apparel items, which realized a lower margin during the nine months ended June 30, 2008.

Profit-sharing distributions. Profit-sharing distributions increased \$13.4 million, or 24.8%, to \$67.6 million for the nine months ended June 30, 2008 from \$54.2 million for the nine months ended June 30, 2007, primarily due to 21.0% growth in our surplus business and 47.1% growth in our scrap business. As a percentage of revenue, profit-sharing distributions decreased to 34.9% from 36.9%, primarily due to a decrease in the amount of profits we are required to pay the DoD under our Surplus Contract, which was modified on September 12, 2006, as well as the Scrap Contract, which was modified on June 1, 2007. A detailed discussion of the Surplus and Scrap Contract modifications can be found above under "Our Seller Agreements."

23

Table of Contents

Technology and operations expenses. Technology and operations expenses increased \$6.3 million, or 26.0%, to \$30.7 million for the nine months ended June 30, 2008 from \$24.4 million for the nine months ended June 30, 2007, primarily due to (1) the addition of 120 technology and operations personnel, the majority of whom were needed to support the increased volume of transactions and merchandise discussed above; (2) an additional 63 operating personnel, who were needed to support our inventory assurance program under the Surplus Contract; and (3) the acquisition of GovDeals, which

was completed on January 1, 2008. As a percentage of revenue, these expenses decreased to 15.8% from 16.6%, primarily due to our growth in revenue, while leveraging our fixed expenses, such as programming personnel.

Sales and marketing expenses. Sales and marketing expenses increased \$2.8 million, or 28.5%, to \$12.5 million for the nine months ended June 30, 2008 from \$9.7 million for the nine months ended June 30, 2007, primarily due to (1) our hiring of 45 additional sales and marketing personnel; (2) \$1.2 million in increased expenditures on marketing and promotional activities across our marketplaces; and (3) the acquisition of GovDeals, which was completed on January 1, 2008. As a percentage of revenue, these expenses were 6.5% for the nine months ended June 30, 2008 and 6.6% for the nine months ended June 30, 2007.

General and administrative expenses. General and administrative expenses increased \$3.7 million, or 30.8%, to \$15.9 million for the nine months ended June 30, 2008 from \$12.2 million for the nine months ended June 30, 2007, primarily due to (1) costs of \$0.9 million related to additional accounting, legal, insurance, compliance and other expenses needed to support our growth; (2) expenses of \$0.6 million related to the adoption of Statement 123(R); (3) costs of \$0.2 million for travel related expenses associated with business development efforts; (4) expenses of \$0.5 million associated with GovDeals, which was acquired on January 1, 2008; and (5) expenses of \$0.3 million associated with Geneva, which was acquired on May 1, 2008. As a percentage of revenue, these expenses were 8.2% for the nine months ended June 30, 2008 and 8.3% for the nine months ended June 30, 2007.

Amortization of contract intangibles. Amortization of contract intangibles was \$0.6 million for the nine months ended June 30, 2008 and 2007, as a result of our DoD Scrap Contract award during June 2005. This contract required us to purchase the rights to operate the scrap operations of the DoD during the seven-year base term of the contract. The intangible asset created from the \$5.7 million purchase is being amortized on a straight-line basis over 84 months, which began in August 2005.

Depreciation and amortization expenses. Depreciation and amortization expenses increased \$0.5 million, or 53.6%, to \$1.4 million for the nine months ended June 30, 2008 from \$0.9 million for the nine months ended June 30, 2007, primarily due to additional depreciation expense resulting from the purchase of \$2.7 million of property and equipment during the fiscal year ended September 30, 2007.

Interest income and expense and other income, net. Interest income and expense and other income, net decreased \$0.2 million, or 13.7%, to \$1.4 million for the nine months ended June 30, 2008 from \$1.6 million for the nine months ended June 30, 2007, primarily due to (1) a reduction in the cash balance due to the recent GovDeals and Geneva acquisitions and (2) a reduction in short term interest rates.

Provision for income tax expense. Income tax expense increased \$0.8 million, or 13.9%, to \$6.2 million for the nine months ended June 30, 2008 from \$5.4 million for the nine months ended June 30, 2007, primarily due to the increase in income before provision for income taxes.

Net income. Net income increased \$1.0 million, or 13.0%, to \$8.9 million for the nine months ended June 30, 2008 from \$7.9 million for the nine months ended June 30, 2007.

Liquidity and Capital Resources

Historically, our primary cash needs have been working capital (including capital used for inventory purchases), which we have funded primarily through cash generated from operations. As of June 30, 2008, we had approximately \$50.6 million in cash and cash equivalents, \$0.7 million in short-term investments and \$25.9 million available under our \$30.0 million senior credit facility, due to issued letters of credit for \$4.1 million; \$1.0 million of our availability under this facility is set aside as a contractual obligation under our DoD Scrap Contract.

Substantially all of our sales are recorded subsequent to receipt of payment authorization, utilizing credit cards, wire transfers and PayPal, an Internet based payment system, as methods of payments. As a result, we are not subject to significant collection risk, as goods are generally not shipped before payment is received.

24

Table of Contents

Changes in Cash Flows: Nine Months Ended June 30, 2008 Compared to Nine Months Ended June 30, 2007

Net cash provided by operating activities increased \$17.4 million to \$16.6 million for the nine months ended June 30, 2008 from \$0.8 million of net cash used in operating activities for the nine months ended June 30, 2007. For the nine months ended June 30, 2008, net cash provided by operating activities primarily consisted of net income of \$8.9 million, depreciation and amortization expense of \$2.1 million, stock compensation expense of \$3.4 million, a net decrease in accounts receivable, inventory and prepaid assets of \$1.2 million and a net increase in accounts payable, accrued expenses and other liabilities of \$1.1 million, offset in part by a decrease in the provision for doubtful accounts of \$0.1 million. For the nine months ended June 30, 2007, net cash used in operating activities primarily consisted of net income of \$7.8 million, depreciation and amortization expense of \$1.6 million, stock compensation expense of \$1.4 million and a net increase in accounts payable, accrued expenses and other liabilities of \$0.3 million, primarily offset by an increase in inventory of \$7.8 million and accounts receivable and prepaid expenses and other assets of \$4.1 million.

Net cash used in investing activities was \$6.2 million for the nine months ended June 30, 2008 and \$19.2 million for the nine months ended June 30, 2007. Net cash used in investing activities for the nine months ended June 30, 2008 consisted primarily of \$25.6 million for the purchase of GovDeals and Geneva and capital expenditures of \$1.3 million for purchases of equipment and leasehold improvements, offset in part by net proceeds from sales of short-term investments of \$20.7 million. Net cash used in investing activities for the nine months ended June 30, 2007 consisted primarily of capital expenditures of \$2.3 million for purchases of equipment and leasehold improvements, \$10.2 million for the purchase of STR and net purchases of short-term investments of \$6.7 million.

Net cash provided by financing activities was \$0.1 million for the nine months ended June 30, 2008 and \$2.5 million for the nine months ended June 30, 2007. Net cash provided by financing activities in the nine months ended June 30, 2007 consisted primarily of net proceeds from our follow on public stock offering of \$1.1 million and proceeds from the exercise of common stock options including the tax benefit of \$1.5 million, offset in part by the repayment of notes payable and lease obligations of \$0.1 million.

Capital Expenditures. Our capital expenditures consist primarily of computers and purchased software, office equipment, furniture and fixtures, and leasehold improvements. The timing and volume of such capital expenditures in the future will be affected by the addition of new customers or expansion of

existing customer relationships. We expect capital expenditures to range from \$1.5 million to \$2.0 million in the fiscal year ending September 30, 2008. We intend to fund those expenditures primarily from operating cash flows. Our capital expenditures for the three- and nine-months ended June 30, 2008 were \$0.5 million and \$1.3 million, respectively. As of June 30, 2008, we had no outstanding commitments for capital expenditures.

Senior credit facility. We maintain a \$30.0 million senior credit facility due March 31, 2010. The senior credit facility bears an annual interest rate of LIBOR plus 1.5%. As of June 30, 2008, we had no outstanding indebtedness under our senior credit facility and our borrowing availability was \$25.9 million due to issued letters of credit for \$4.1 million; \$1.0 million of our availability under this facility is set aside as a contractual obligation under our DoD Scrap Contract. The obligations under our senior credit facility are unconditionally guaranteed by us and each of our existing and subsequently acquired or organized subsidiaries (other than our subsidiaries organized to service our DoD contracts) and secured on a first priority basis by security interests (subject to permitted liens) in substantially all assets owned by us, and each of our other domestic subsidiaries, subject to limited exceptions. Our credit agreement contains a number of affirmative and restrictive covenants including limitations on mergers, consolidations and dissolutions, sales of assets, investments and acquisitions, indebtedness and liens, and dividends and other restricted payments. As of June 30, 2008, we were in full compliance with the terms and conditions of our credit agreement.

We believe that our existing cash and cash equivalents and short term investments will be sufficient to meet our anticipated cash needs for at least the next 12 months. Our future capital requirements will depend on many factors including our rate of revenue growth, the timing and extent of spending to support development efforts, the expansion of sales and marketing activities, the development and deployment of new marketplaces, the introduction of new value added services and the costs to establish additional distribution centers. We may enter into definitive agreements with respect to potential investments in, or acquisitions of, complementary businesses, products or technologies in the future, which could also require us to seek additional equity or debt financing. The sale of additional equity securities or convertible debt securities would result in additional dilution to our stockholders. Additional debt would result in increased interest expense and could result in covenants that would restrict our operations. There is no assurance that such financing, if required, will be available in amounts or on terms acceptable to us, if at all.

Off-Balance Sheet Arrangements

We do not have any transactions, obligations or relationships that could be considered material off-balance sheet arrangements.

25

Table of Contents

New Accounting Pronouncements

In September 2006, the FASB issued SFAS No. 157, Fair Value Measurements (SFAS No. 157). SFAS No. 157 establishes a framework for measuring fair value under generally accepted accounting principles, clarifies the definition of fair value and expands disclosures about fair value measurements. SFAS No. 157 does not require any new fair value measurements. SFAS No. 157 is effective for financial statements issued for fiscal years beginning after November 15, 2007, and interim periods within those fiscal years. We do not expect SFAS No. 157 to have a material effect on our financial statements.

In February 2007, the FASB issued SFAS No. 159, The Fair Value Option for Financial Assets and Financial Liabilities—Including an amendment of FASB Statement No. 115 (SFAS No. 159). SFAS No. 159 permits entities to choose to measure certain financial instruments and other items at fair value. The fair value option generally may be applied instrument by instrument, is irrevocable, and is applied only to entire instruments and not to portions of instruments. SFAS No. 159 is effective for financial statements issued for fiscal years beginning after November 15, 2007. We do not expect SFAS No. 159 to have a material effect on our financial statements.

In June 2006, the Emerging Issues Task Force (EITF) reached a consensus on Issue No. 06-3, How Sales Taxes Collected from Customers and Remitted to Governmental Authorities Should Be Presented in the Income Statement (That Is, Gross Versus Net Presentation) (EITF 06-3) by concluding that companies should disclose their accounting policy (i.e., gross or net presentation) regarding presentation of taxes within the scope of this issue. The Task Force also concluded that, as of January 1, 2007, companies should disclose the amount of such taxes for periods in which these taxes included in gross revenues are considered material. We collect and remit sales taxes on merchandise that we purchase and sell, and report such amounts under the net method in our consolidated statements of operations.

In December 2007, the FASB issued SFAS No. 141(R), Business Combinations. This statement changes the accounting for acquisitions specifically eliminating the step acquisition model, changing the recognition of contingent consideration from being recognized when it is probable to being recognized at the time of acquisition, disallowing the capitalization of transaction costs and delays when restructurings related to acquisitions can be recognized. We will adopt this statement for fiscal year beginning October 1, 2009 and it will only impact the accounting for acquisitions we make after its adoption, except for the amendment related to income taxes, which will be applied prospectively as of the adoption date and will apply to business combinations with acquisition dates before the effective date of SFAS No. 141(R).

Item 3. Quantitative and Qualitative Disclosures about Market Risk.

Interest rate sensitivity. We did not have any debt as of June 30, 2008 and September 30, 2007 and thus do not have any related interest rate exposure. Our investment policy requires us to invest funds in excess of current operating requirements. The principal objectives of our investment activities are to preserve principal, provide liquidity and maximize income consistent with minimizing risk of material loss.

As of June 30, 2008, our cash and cash equivalents consisted primarily of money market funds and our short term investments consisted primarily of highly rated short term bonds. The recorded carrying amounts of cash and cash equivalents approximate fair value due to their short maturities. Our interest income is sensitive to changes in the general level of interest rates in the United States, particularly since the majority of our investments are short-term in nature. Due to the nature of our short-term investments, which have a duration of three to twelve months, we have concluded that we do not have material market risk exposure.

Exchange rate sensitivity. We consider our exposure to foreign currency exchange rate fluctuations to be minimal, as less than five percent of our GMV is denominated in foreign currencies. We have not engaged in any hedging or other derivative transactions to date.

Table of Contents

Item 4. Controls and Procedures.

We maintain disclosure controls and procedures that are designed to ensure that information required to be disclosed in our reports filed or submitted under the Securities Exchange Act of 1934 is recorded, processed, summarized and reported within the time periods specified in the Securities and Exchange Commission's rules and forms, and that such information is accumulated and communicated to our management, including our Chief Executive Officer and Chief Financial Officer, as appropriate, to allow for timely decisions regarding required disclosure. In designing and evaluating the disclosure controls and procedures, management recognized that any controls and procedures, no matter how well designed and operated, can provide only reasonable assurance of achieving the desired control objectives, and management is required to apply its judgment in evaluating the cost-benefit relationship of possible controls and procedures.

During the most recent fiscal quarter, there has not occurred any change in our internal control over financial reporting that has materially affected, or is reasonably likely to materially affect, our internal control over financial reporting.

As of June 30, 2008, we carried out an evaluation, under the supervision and with the participation of our management, including our Chief Executive Officer and our Chief Financial Officer, of the effectiveness of the design and operation of our disclosure controls and procedures. Based on the foregoing, our Chief Executive Officer and Chief Financial Officer concluded that our disclosure controls and procedures were effective and were operating at the reasonable assurance level.

27

Table of Contents

PART II—OTHER INFORMATION

Item 1. Legal Proceedings.

From time to time, we may become involved in litigation relating to claims arising in the ordinary course of our business. There are no claims or actions pending or threatened against us that, if adversely determined, would in our judgment have a material adverse effect on us.

Item 1A. Risk Factors.

In addition to the other information set forth in this report, you should carefully consider the factor set forth below and the factors discussed in Part I, "Item 1A. Risk Factors" in our Annual Report on Form 10-K for the year ended September 30, 2007, which could materially affect our business, financial condition or future results. The risks described in this Quarterly Report and our Form 10-K are not the only risks we face. Additional risks and uncertainties not currently known to us or that we currently deem to be immaterial also may materially adversely affect our business, financial condition and/or operating results.

We could experience a significant decrease in our revenue and earnings under our new Surplus Contract, which contains certain terms that are different from those of the current Surplus Contract.

We depend on our current Surplus Contract with the U.S. Department of Defense for a significant portion of our revenue and earnings and may experience a significant decrease in our revenue and earnings under our new Surplus Contract. Revenue from our Surplus Contract (including buyer premiums) accounted for approximately 29.8% and 30.9% of our total revenue for the three- and nine-months ended June 30, 2008, respectively. The property sold under our current Surplus Contract accounted for approximately 20.4% and 23.0% of our GMV for the three- and nine-months ended June 30, 2008, respectively. Our current Surplus Contract with the DRMS will expire on December 19, 2008; we expect to commence operations under our new Surplus Contract, which we were awarded on July 31, 2008, during the first half of our fiscal year 2009. The new Surplus Contract contains certain terms that are different from the terms of the current Surplus Contract, including, without limitation: (1) a change in the economic structure of the contract, which eliminates the profit sharing terms of the contract and requires us to bear all of our costs for the merchandising and resale of the property sold to us under the new Surplus Contract; (2) a requirement that we purchase all useable surplus property offered for sale to us by the DRMS under the new Surplus Contract; (3) a provision granting broad discretion to the DRMS to determine what property will be made available for sale to us under the New Surplus Contract; (4) a provision granting authority to the DRMS to retrieve or restrict property previously sold to us under the New Surplus Contract for national security reasons or if the property is otherwise needed to support the mission of the U.S. Department of Defense; (5) a 36 month term (with two 12 month renewal options exercisable by the DRMS) and (6) a mutual termination of the contract for convenience. The terms of the new Surplus Contract can be found in Exhibit 10.1 to our Current Report on Form 8-K filed with the Securities and Exchange Commission on August 6, 2008. We can not predict what effect these terms will have on our business and we cannot assure you that such terms will not have a material adverse affect on our revenue and earnings. We may incur additional expense if our competitors protest or challenge the award of a new surplus contract to us. Any such challenge could result in a requirement that we resubmit our bid in a new bidding process for the contract and/or the termination of the award of the new contract to us.

28

Table of Contents

Item 6. Exhibits.

Exhibit No.

2.1 Share Purchase Agreements, dated as of April 5, 2008 and April 6, 2008, by and between Liquidity Services, Inc., and its wholly owned subsidiary, Liquidity Services, Ltd., on the one hand, and David Mark Jacobs, Simon Jacobs, Darren Lee Innocent and Darren Malcolm Dorrington, on the other.

Fourth Amended and Restated Certificate of Incorporation, incorporated herein by reference to Exhibit 3.1 to Amendment No. 2 to the Company's Registration Statement on Form S-1 (Registration No. 333-129656), filed with the SEC on January 17, 2006.

3.2	Amended and Restated Bylaws, incorporated herein by reference to Exhibit 3.2 to Amendment No. 2 to the Company's Registration
	Statement on Form S-1 (Registration No. 333-129656), filed with the SEC on January 17, 2006.
10.1	Supplemental Agreement, dated as of May 13, 2008, relating to Commercial Venture II (CV-II) (Sales Contract Number 99-0001-0002)
	between Surplus Acquisition Venture, LLC (a wholly owned subsidiary of Liquidity Services, Inc.) and the Defense Reutilization and
	Marketing Service of the U.S. Department of Defense.
31.1	Certification of Chief Executive Officer pursuant to Rule 13a-14(a) of the Securities Exchange Act of 1934.
31.2	Certification of Chief Financial Officer pursuant to Rule 13a-14(a) of the Securities Exchange Act of 1934.
32.1	Certification of Chief Executive Officer pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.
32.2	Certification of Chief Financial Officer pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.

29

Table of Contents

SIGNATURES

Pursuant to the requirements of Section 13 or 15(d) of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized on August 8, 2008.

LIQUIDITY SERVICES, INC. (Registrant)

By: /s/ William P. Angrick, III
William P. Angrick, III
Chairman of the Board of Directors
and Chief Executive Officer

By: /s/ James M. Rallo
James M. Rallo
Chief Financial Officer and Treasurer

30

29

Execution version

Dated 5 and 6 April 2008

(1) DAVID MARK JACOBS
(2) SIMON JACOBS
(3) DARREN LEE INNOCENT
(4) DARREN MALCOLM DORRINGTON
(5) LIQUIDITY SERVICES LIMITED

– and –
(6) LIQUIDITY SERVICES, INC.

SHARE PURCHASE AGREEMENTS

in relation to the sale and purchase of the entire issued share capitals of

(1) Geneva Industries Limited and Willen Trading Limited; and

(2) Geneva Auctions Limited

GIBSON, DUNN & CRUTCHER LLP

Telephone House
2-4 Temple Avenue, London EC4Y 0HB
020 7071 4000 020 7071 4244 Fax
Ref: 56526/00001

CONTENTS				
Clause	Subject Matter	Page		
1.	Definitions and Interpretation	3		
2.	Effective Dates	12		
3.	Sale and Purchase	13		
4.	Price	13		
5.	DLI Earn-Out	14		
6.	SJ Earn-Out	14		
7.	DMJ Earn-Outs	15		
8.	Escrow Accounts	15		
9.	Pre-Closing Seller undertakings	18		
10.	Closing	19		
11.	Warranties, Undertakings and Indemnities	19		
12.	Purchaser Warranties	20		
13.	Conduct of Purchaser Claims	20		
14.	Purchaser's rights to terminate	21		
15.	Tax	22		
16.	Insurance	22		
17.	Information, Records and Assistance Post-closing	22		
18.	Protective Covenants Post-closing	22		
19.	Post-Closing Undertakings	23		
20.	Payments	23		
21.	Announcements	24		
22.	Confidentiality	24		
23.	Assignment	25		
24.	Further Assurances	26		
25.	Costs	26		
26.	Notices	26		
27.	Conflict with other Agreements	27		
28.	Whole Agreement	27		
29.	Waivers, Rights and Remedies	28		
30.	Guarantees and Indemnities	28		
31.	Effect of Closing	29		
32.	Counterparts	29		
33.	Variations	29		

34.

Invalidity

35.	No third party enforcement rights	29
36.	Governing law and jurisdiction	29
Schedule	· · · · · · · · · · · · · · · · · · ·	31
Sellers a	and Price	31
Schedule	le 2	32
Post-Clo	osing Financial Adjustments	32
	Preliminary	32
	Specific Accounting Treatments	33
	Closing Statement	34
	Financial Adjustments	36
Schedule	le 3	38
Earn-Ou	ut	38
	1	
Part A -	Definitions and Interpretation	38
Part B -	- DLI Earn-Out	40
Part C -	- SJ Earn-Out	42
Part D -	- DMJ Willen Earn-Out	44
	- DMJ Geneva Earn-Out	46
	Target Company EBITDA Calculation Procedure	48
	to Schedule 3	49
Schedule	le 4	50
Conduct	t of the Business Pre-Closing	50
Schedule		53
	Arrangements	53
	Sellers' Obligations	53
	Purchaser Obligations	55
Part C :		55
Schedule		56
	Varranties	56
Part A:		56
Part B:		64
	Real Estate	67
	Employment	69
	Pensions	74
	Insurance The Marketine	74
Schedule	Tax Warranties	75 83
	n schedule	83
Schedule		95
	er warranties	95
Schedule		96
	ions on Liability	96
Schedule		102
Property		102
Schedule		103
Owned I		103
Exhibit 1		104
	companies	104
Exhibit 2		107
	al Adjustments	107
	ts for Target Working Capital	107
Exhibit 3		108
	agreement	108
Exhibit 4		109
	ure Letter	109
Exhibit 5		110
	f Service Agreement	110
Exhibit 6	6	111
	Agreement	111
Exhibit 7		112

SHARE PURCHASE AGREEMENTS

2

112

THESE AGREEMENTS are entered into on:

Form of Loan Note Instrument

(B) 6 April 2008 in respect of Geneva Auctions Limited (the "6 April Agreement").

BETWEEN:

- 1. **DAVID MARK JACOBS** of Hollybush Farm, Botley Road, Chesham, Buckinghamshire HP5 1XG ("**DMJ**");
- 2. **SIMON JACOBS** of 44 Marsworth Avenue, Pinner, Middlesex HA5 4TT ("SJ");
- 3. **DARREN LEE INNOCENT** of Bakers Farm, High Street, Shenington, Buckinghamshire MK16 9NU ("DLI");
- DARREN MALCOLM DORRINGTON of 34 Saltwood Avenue Kingsmead Milton Keynes MK4 4HP ("DMD"),

(together, DMJ, SJ, DLI and DMD shall be known herein as the "Sellers");

- 5. **LIQUIDITY SERVICES LIMITED,** a private company limited by shares, incorporated under the laws of England with company number 04843035, whose registered office is at Unit 4 Prince Rupert House, Cavalier Court, Bumpers Farm, Chippenham, Wiltshire SN14 6LH (the "**Purchaser**"); and
- 6. **LIQUIDITY SERVICES, INC.** a corporation incorporated in the state of Delaware, USA and whose principal place of business is at 1920 L St. NW, 6th Floor, Washington DC 20036, USA (the "**Purchaser Parent**"),

(together, the Sellers, the Purchaser and the Purchaser Parent shall be know herein as the "Parties").

IT IS AGREED AS FOLLOWS:

- 1. Definitions and Interpretation
- 1.1 Definitions

In this Agreement, the following words and expressions shall have the following meanings:

- "5 **April Agreement**" shall have the meaning given to it in the preamble to this Agreement;
- "6 April Agreement" shall have the meaning given to it in the preamble to this Agreement;

3

- "Accounting Principles" has the meaning given in Part A of Schedule 2;
- "Accounts" means in relation to any financial year of each Target Company, the balance sheet of that company (and, where relevant, the consolidated balance sheet of that company and its subsidiary undertakings) as at the Accounts Date in respect of that financial year and the profit and loss account of that company (and, where relevant, the consolidated profit and loss account of that company and its subsidiary undertakings, if applicable) as at the Accounts Date in respect of that financial year together, in each case, with any notes, reports, statements or documents included in or annexed or attached to them;
- "Accounts Date" means 31 December;
- "Affiliate" means in relation to any Party, any subsidiary or parent company of that Party and any subsidiary of any such parent company, in each case from time to time:
- "Agreed Form" means, in relation to a document, the form of that document as initialled for the purpose of identification by or on behalf of the Sellers and the Purchaser (in each case with such amendments as may be agreed by them or on their behalf);
- "Business" means the business of the relevant Target Companies;
- "Business Day" means a day other than a Saturday or Sunday or public holiday in England and Wales on which banks are open in London for general commercial business;
- "Business IP" means the Owned IP and all other registered and material unregistered Intellectual Property Rights used by any Target Company in connection with its Business;
- "CAA 2001" means the Capital Allowances Act 2001;
- "Cash" means, in relation to each Target Company, the aggregate of its cash (whether in hand or credited to any account with any banking, financial, acceptance credit, lending or other similar institution or organisation) and its cash equivalents, including in each case all interest accrued thereon, as at Closing, as shown by the books of that Target Company;
- "Claim" means any claim for breach of Warranty or under the Tax Covenant;
- "Closing" means completion of the sale and purchase of the Shares in accordance with the provisions of this Agreement;
- "Closing Date" means 1 May 2008 or such other date as may be agreed between the Parties from time to time;
- "Closing Statement" has the meaning given in Part C of Schedule 2;

4

paragraphs 1.5, 1.7, 1.9, 1.11, 1.14 of Part D of Schedule 6; and Parts E, F and G of Schedule 6.

"Competing Business" shall have the meaning given to it in clause 18.1;

"Confidential Information" has the meaning given in clause 22;

"Counsel" shall have the meaning given to it in clause 8.5;

"Customers" means any third party customer of the Business (whether a buyer or seller of goods or services offered in connection therewith);

"Default Interest" means interest at two (2) per cent. above the base rate from time to time of Barclays Bank plc;

"Disclosed" means fairly disclosed (and, in relation to specific disclosures, disclosed with sufficient details to identify the nature and scope of the matter purportedly disclosed) in the Disclosure Letter and Disclosure Bundle;

"Disclosure Bundle" means the disclosure bundle comprising the actual copies of documents and other information relating to the Target Companies made available by the Sellers in the Agreed Form attached to the Disclosure Letter;

"Disclosure Letter" means the letter from the Sellers to the Purchaser executed and delivered immediately before the signing of this Agreement;

"DLI" has the meaning given to it in the preamble to this Agreement;

"**DLI Earn-Out**" has the meaning given to it in clause 5;

"DLI Escrow Account" has the meaning given to it in clause 4.3 of this Agreement;

"DLI Escrow Amount" has the meaning given to it in clause 4.3 of this Agreement;

"DMD" has the meaning given to it in the preamble to this Agreement;

"DMJ Earn-Outs" has the meaning given to it in clause 7

"DMD Escrow Account" has the meaning given to it in clause 4.3 of this Agreement;

"DMD Escrow Amount" has the meaning given to it in clause 4.3 of this Agreement;

"**DMJ**" has the meaning given to it in the preamble to this Agreement;

"DMJ Escrow Account" has the meaning given to it in clause 4.3 of this Agreement;

"DMJ Escrow Amount" has the meaning given to it in clause 4.3 of this Agreement;

"**DMJ Geneva Earn-Out**" has the meaning given to it in clause 7;

"DMJ Willen Earn-Out" has the meaning given to it in clause 7;

"**Due Amount**" shall have the meaning given to it in clause 8.11;

5

"Employee Benefit Plans" shall have the meaning given to it in paragraph 1.1 of Part D of Schedule 6;

"Employees" means the employees of the relevant Target Companies immediately prior to Closing;

"Escrow Accounts" means the DMJ Escrow Account, the DMD Escrow Account, the DLI Escrow Account and the SJ Escrow Account and each shall be known as an "Escrow Account";

"Escrow Agent" means The Bank of New York acting through its London Branch at One Canada Square, London, E14 5AL;

"Escrow Agreement" means the agreement to be entered into between the Escrow Agent, the Purchaser and each of the Sellers in the Agreed Form and as set out in Exhibit 6;

"Escrow Amounts" means the DMJ Escrow Amount, the DMD Escrow Amount, the DLI Escrow Amount and the SJ Escrow Amount;

"Escrow Payment Date" means three (3) Business Days following the expiration of the Escrow Period;

"Escrow Period" has the meaning given to it in clause 8.1 to this Agreement;

"Event" means any event, circumstance, effect, occurrence or state of affairs or any combination of them (whether existing or occurring on or before the date of this Agreement or arising or occurring afterwards), except as otherwise defined in Schedule 7;

"Exhibits" means the exhibits to this Agreement, and "Exhibit" shall be construed accordingly;

"Final Share Price" has the meaning given in clause 4.2;

"Financial Adjustments" means any adjustment(s) required in accordance with Part D of Schedule 2;

"Financial Debt" means all borrowings and other indebtedness (whether owed to third parties or shareholders of the Target Companies) by way of overdraft, acceptance credit or similar facilities, loan stocks, bonds, debentures, notes, debt or inventory financing, finance leases or sale and lease back arrangements or any other arrangements the purpose of which is to borrow money, together with forex, interest rate or other swaps, hedging obligations, bills of exchange, recourse obligations on factored debts and obligations under other derivative instruments;

"Guaranteed Obligations" means all present and future payment obligations and liabilities of the Purchaser to the Sellers pursuant to this Agreement;

"ICTA 1988" the Income and Corporation Taxes Act 1988;

"IHTA 1984" means the Inheritance Tax Act 1984:

6

"Income" has the meaning given to it in clause 8.2(b) of this Agreement;

"Initial Share Prices" has the meaning given in clause 4.2;

"Intellectual Property Rights" means patents, trade marks, service marks, logos, get-up, trade names, internet domain names, rights in designs, copyright (including rights in computer software) and moral rights, database rights, semi-conductor topography rights, utility models, rights in know-how and other intellectual property rights, in each case whether registered or unregistered, and all rights or forms of protection having equivalent or similar effect anywhere in the world and "registered" includes registration, and applications for registration;

"IT Systems" means the information and communications technologies used by the Target Companies including hardware, proprietary and third party software, services, networks, peripherals and associated documentation;

"Key Executives" means DMJ, DMD and DLI;

"Last Accounts" means, in relation to any Target Company, the Accounts of that entity in respect of its financial year ended on the Last Accounts Date, as set out in the Disclosure Bundle;

"Last Accounts Date" means 31 December 2007;

"Lease" means an office and warehouse lease in the Agreed Form between DMJ and the Purchaser in relation to the Property, as set out in Exhibit 3;

"Loan Note Instrument" means the loan note instrument in the Agreed Form and as set out in Exhibit 7;

"Loan Notes" means the loan notes constituted by the Loan Note Instrument in the Agreed Form;

"Management Accounts" means the unaudited monthly management accounts during the period commencing on the Last Accounts Date and ending on the Management Accounts Date, each in the form contained in the Disclosure Bundle;

"Management Accounts Date" means 31 March 2008;

"Material Adverse Change" any Event which is, or is reasonably likely to be, materially adverse to the business, operations, assets, liabilities (including contingent liabilities), the Property or the business or financial condition, results or prospects of the relevant Target Company;

"Non-Tax Claim" means a Claim other than a claim for breach of any of the Tax Warranties or a claim under the Tax Covenant;

"Notice of Claim" has the meaning given to it in Schedule 9;

"Owned IP" means the registered and material unregistered Intellectual Property Rights owned by any Target Company and the registered and material unregistered

7

Intellectual Property Rights owned by any Target Company and used in relation to its Business. The registered Owned IP is listed in Schedule 11;

"**Opinion**" shall have the meaning given to it in clause 8.6;

"Outstanding Claims" shall have the meaning given to it in clause 8.11;

"parent company" means any company which holds a majority of the voting rights in another company, or which is a member of another company and has the right to appoint or remove a majority of its board of directors, or which is a member of another company and controls a majority of the voting rights in it under an agreement with the other members, in each case whether directly or indirectly through one or more companies;

"parties" or "Parties" means the parties to this Agreement (and each shall be known as a Party or party);

"Permitted Assignee" has the meaning given in clause 23;

"Property" means the property, short particulars of which are set out in Schedule 10;

"Proposed Transactions" means the transactions contemplated by the Transaction Documents;

"Protected Territories" shall have the meaning given to it in clause 18.1;

"Purchaser" has the meaning given to it in the preamble to this Agreement;

"Purchaser Group" means the Purchaser and its Affiliates from time to time (including the Target Companies from Closing);

"Purchaser Obligation" means any representation, warranty or undertaking to indemnify (including any covenant to pay pursuant to the Tax Covenant) given by the Purchaser to the Sellers under this Agreement;

"Purchaser Parent" has the meaning given to it in the preamble to this Agreement;

"Purchaser's Solicitors" means Gibson, Dunn & Crutcher LLP of Telephone House, 2-4 Temple Avenue, London EC4Y 0HB;

"Purchaser's Solicitor's Bank Account" means the Purchaser's Solicitor's bank account at Barclays Bank Plc (UK Banking), 1 Churchill Place, London E14 5HP; Account Name: Gibson, Dunn & Crutcher LLP Clients' Premium Account; Account Number: 80372226; Sort Code: 20-00-00; Swift Code BARCGB22; IBAN Number: GB06 BARC 2000 0080 3722 26 (and/or such other account(s) as the Purchaser notifies to the Sellers in writing);

"Records" has the meaning given in clause 17.1;

"Regulations" means the Transfer of Undertakings Protection of Employment Regulations 2006;

8

"Relevant Claim Percentage" means, in relation to a Seller, the percentage set out against their name in column 6 of Schedule 1, corresponding to a particular Set of Shares;

"Relevant Working Capital Percentage" means, in relation to a Seller, the percentage set against their name in column 7 of Schedule 1, corresponding to the relevant Set of Shares;

"Schedules" means the schedules to this Agreement, and "Schedule" shall be construed accordingly;

"Seller Obligation" means any Warranty or undertaking to indemnify (including any covenant to pay pursuant to the Tax Covenant and any Sellers' Obligations listed in Part A of Schedule 5) given by that Seller to the Purchaser under this Agreement;

"Sellers" has the meaning given to it in the preamble to this Agreement;

"Sellers' Representative" means DMJ;

"Sellers' Solicitors" means Levine Mellins Klarfeld of 24-26 Church Road, Stanmore, Middlesex HA7 4AW;

"**Set of Shares**" means, in relation to a Seller, the shares comprising the issued share capital of any particular Target Company which are to be sold by that Seller under this Agreement;

"Shares" means the shares comprising the entire issued share capital of each of the Target Companies set out opposite the names of the Sellers in column 2 of Schedule 1;

"SJ" has the meaning given to it in the preamble to this Agreement;

"SJ Earn-Out" has the meaning given to it in clause 6;

"SJ Escrow Account" has the meaning given to it in clause 4.3 of this Agreement;

"SJ Escrow Amount" has the meaning given to it in clause 4.3 of this Agreement;

"Specific Accounting Treatments" has the meaning given in Part A of Schedule 2;

"**Strikeable Claim**" shall have the meaning given to it in clause 8.7;

"Strikeable Portion" shall have the meaning given to it in clause 8.8;

"subsidiary" and "subsidiaries" means any company in relation to which another company is its parent company;

"Surviving Provisions" means 1 (Definitions and Interpretation), 21 (Announcements), 22 (Confidentiality), 23 (Assignment), 25 (Costs), 26 (Notices), 27 (Conflict with other Agreements), 28 (Whole Agreement), 29 (Waivers, Rights and Remedies), 33 (Variations), 34 (Invalidity), 35 (No third party enforcement rights), 36 (Governing law and jurisdiction);

9

"Target Companies" means the companies the Shares of which are listed in column 2 of Schedule 1 and details of which are set out in Exhibit 1, and "Target Company" means any of them;

"Target Working Capital" means, £875,222 in aggregate (being £188,830 in relation to Willen Trading Limited and £686,392 in relation to Geneva Industries Limited) as set out in Exhibit 2;

"Tax" shall have the meaning given to it in Schedule 7, and Taxation shall have the same meaning;

"**Taxation Authority**" shall have the meaning given to it in Schedule 7;

"Tax Claim" means a claim for a breach of any of the Tax Warranties or a claim under the Tax Covenant;

"**Tax Covenant**" means the covenant relating to Tax set out in Schedule 7;

"Taxation Statute" means any directive, statute, enactment, law or regulation wheresoever enacted or issued, coming into force or entered into providing for or imposing any Tax and including orders, regulations, instruments, bye-laws or other subordinate legislation made under the relevant statute or statutory provision and any directive, statute, enactment, law, order, regulation or provision which amends, extends, consolidates or replaces the same or which has been amended, extended, consolidated or replaced by the same;

"Tax Warranties" means the warranties set out in Part G of Schedule 6;

"TCGA 1992" means the Taxation of Chargeable Gains Act 1992;

"Third Party Claim" has the meaning given to it in clause 13.1;

"Third Party Contracts" shall have the meaning given to it in paragraph 12.1 of Part B of Schedule 6;

"Third Party Right" means any interest or equity of any person (including any right to acquire, option or right of pre-emption or conversion) or any mortgage, charge, pledge, lien, assignment, hypothecation, security interest, title retention or any other security agreement or arrangement, or any agreement to create any of the above;

"TMA 1970" means the Taxes Management Act 1970;

"Transaction Documents" means this Agreement, the Disclosure Letter, the Escrow Agreement, the Loan Note Instrument, the Loan Notes, the Lease, and any other documents in the Agreed Form;

"UK GAAP" means generally accepted accounting principles applied in the UK, incorporating Statements of Standard Accounting Practice, Financial Reporting Standards and Urgent Issues Task Force Abstracts issued by the Accounting Standards Board, International Accounting Standards and International Financial Reporting Standards issued by International Accounting Standards Board in each case as in force at the Accounts Date;

10

"VAT" means value added tax and any similar sales or turnover tax;

"VATA 1994" means the Value Added Tax Act 1994;

"Warranties" means the warranties set out in clause 11 and Schedule 6;

"Warranty Claim" means a Claim for breach of any of the Warranties;

"Working Capital" means, in relation to Geneva Industries Limited and Willen Trading Limited, the working capital of those companies as at Closing, comprising each of the line items set out in the reference balance sheets included at Exhibit 2 and no other; for the avoidance of doubt Working Capital includes all Cash and all interest payable or receivable accrued as at Closing;

"Working Capital Adjustment" means the amount of the difference between the aggregate Working Capital and the Target Working Capital if that aggregate Working Capital is greater than the Target Working Capital (or subtracting the amount of such difference if that aggregate Working Capital is less than the Target Working Capital); and

"Working Hours" means 9.30 a.m. to 5.30 p.m. in England and Wales on a Business Day.

1.2 Interpretation

In this Agreement, unless the context otherwise requires:

- (a) references to a "**person**" include any individual, firm, body corporate (wherever incorporated), government, state or agency of a state or any joint venture, association, partnership, works council or employee representative body (whether or not having separate legal personality);
- (b) headings do not affect the interpretation of this Agreement; the singular shall include the plural and vice versa; and references to one gender include all genders;
- (c) references to any English legal term or concept shall, in respect of any jurisdiction other than England, be construed as references to the term or concept which most nearly corresponds to it in that jurisdiction;
- (d) references to "**sterling**" or "**pounds sterling**" or "£" are references to the lawful currency from time to time of England;
- (e) any statement in this Agreement qualified by the expression "to the best of the Sellers' knowledge" or "so far as the Sellers are aware" or any similar expression shall be deemed to include an additional statement that it has been made after due and careful enquiry and shall be deemed also to include the knowledge of each director of the Target Companies and Barry Plummer;

11

- (f) any phrase introduced by the terms "**including**", "**include**", "**in particular**" or any similar expression shall be construed as illustrative and shall not limit the sense of the words preceding those terms; and
- (g) any reference to "**this Agreement**" shall be deemed to be a reference to the 5 April Agreement or the 6 April Agreement, as applicable.

1.3 Enactments

Except as otherwise expressly provided in this Agreement, any express reference to an enactment (which includes any legislation in any jurisdiction) includes references to (i) that enactment as amended, consolidated or re-enacted by or under any other enactment before or after the date of this Agreement (ii) any enactment which that enactment re-enacts (with or without modification) and (iii) any subordinate legislation (including regulations) made (before or after the date of this agreement) under that enactment, as amended, consolidated or re-enacted as described in (i) or (ii) above except to the extent that any of the matters referred to in (i) to (iii) above occurs after the date of this Agreement and increases or alters the liability of the Sellers or the Purchaser (or any person on whose behalf it is acting as agent pursuant to this Agreement) under this Agreement.

1.4 Schedules and Exhibits

The Schedules and Exhibits comprise schedules and exhibits to this Agreement and form part of this Agreement.

1.5 Inconsistencies

Where there is any inconsistency between the definitions set out in this clause 1 and the definitions set out in any other clause or any Schedule, then, for the purposes of construing such clause or Schedule, the definitions set out in such clause or Schedule shall prevail.

2. Effective Dates

- 2.1 The 5 April Agreement shall be entered into and effective:
 - (a) as between DMJ and SJ, in respect of the sale of their Set of Shares in Geneva Industries Limited: and
 - (b) as between DMJ and DLI in respect of the sale of their Set of Shares in Willen Trading Limited,

on 5 April 2008.

- 2.2 The 6 April Agreement shall be entered into and effective as between DMJ and DMD, in respect of the sale of their Set of Shares in Geneva Auctions Limited, on 6 April 2008.
- 2.3 For the avoidance of doubt, the sale of shares contemplated in clause 2.1 above shall not be conditional upon the sale of shares contemplated in clause 2.2 above.

12

3. Sale and Purchase

- 3.1 Each Seller shall sell the Set of Shares for which it is identified as the respective Seller in Schedule 1, and the Purchaser shall purchase the Shares. Each Set of Shares shall be sold with all rights attaching to them at Closing including the right to receive all distributions and dividends declared, paid or made in respect of each such Set of Shares after Closing.
- 3.2 The Shares shall be sold free from all Third Party Rights and ownership and risk in the Shares shall (except as otherwise set out in this Agreement) pass to the Purchaser with effect from Closing.

4. Price

- 4.1 The overall price for the Shares shall be the aggregate of the Final Share Prices as set out in this clause 4.
- 4.2 In relation to each Set of Shares, the "Final Share Price" shall be the amount which results from:
 - (a) taking the price for that Set of Shares (set against such Seller's name in column 3 of Schedule 1) (the "Initial Share Prices" and each an "Initial Share Price");
 - (b) in the case of DMJ, DLI and SJ, and adding or subtracting, as appropriate, the amount of the Working Capital Adjustment, multiplied by the Relevant Working Capital Percentage; and
 - (c) in the case of DMJ, DLI and SJ, adding any amounts actually received pursuant to clauses 5, 6, and 7 respectively.
- 4.3 The Initial Share Prices shall be satisfied by the Purchaser at Closing, by the Purchaser procuring:
 - (a) the payment by telegraphic transfer of the sum set out opposite each of the names of the Sellers in column 5 of Schedule 1 in cash to the Sellers' Solicitors;
 - (b) the payment by telegraphic transfer of the sum set out opposite DMJ in column 4 of Schedule 1 (the "**DMJ Escrow Amount**") in cash, to be held on deposit account with the Escrow Agent in the joint names of the Purchaser and DMJ, which sum shall be dealt with in accordance with the provisions of clause 8 and the provisions of the Escrow Agreement (the "**DMJ Escrow Account**");
 - (c) the payment by telegraphic transfer of the sum set out opposite SJ in column 4 of Schedule 1 (the "SJ Escrow Amount") in cash, to be held on deposit account with the Escrow Agent in the joint names of the Purchaser and SJ, which sum shall be dealt with in accordance with the provisions of clause 8 and the provisions of the Escrow Agreement (the "SJ Escrow Account");

13

- (d) the payment by telegraphic transfer of the sum set out opposite DMD in column 4 of Schedule 1 (the "**DMD Escrow Amount**") in cash, to be held on deposit account with the Escrow Agent in the joint names of the Purchaser and DMD, which sum shall be dealt with in accordance with the provisions of clause 8 and the provisions of the Escrow Agreement (the "**DMD Escrow Account**").
- (e) the payment by telegraphic transfer of the sum set out opposite DLI in column 4 of Schedule 1 (the "**DLI Escrow Amount**") in cash, to be held on deposit account with the Escrow Agent in the joint names of the Purchaser and DLI, which sum shall be dealt with in accordance with the provisions of clause 8 and the provisions of the Escrow Agreement (the "**DLI Escrow Account**").
- 4.4 Any payments required to be made under the Financial Adjustments shall be treated as adjusting each respective Initial Share Price, thus resulting after such adjustment in each respective Final Share Price. Each Final Share Price shall (subject to any further adjustment, if applicable, pursuant to clause 4.5) be adopted for all Tax reporting purposes.
- 4.5 Any payment made in satisfaction of a liability arising under a Seller Obligation or a Purchaser Obligation shall be made on the following basis:
 - (a) if it is specifically referable to any particular Set of Shares, it shall so far as possible adjust the price paid for that Set of Shares;
 - (b) otherwise, it shall adjust the price for such Set of Shares as the relevant Seller (or Sellers) and the Purchaser agree to be appropriate in the circumstances or, if they do not agree, it shall adjust the price paid for the Shares according to the Relevant Claim Percentage.

5. DLI Earn-Out

- 5.1 As additional consideration for the sale by DLI of his Set of Shares in Willen Trading Limited, DLI will on Closing be issued Loan Notes in accordance with the provisions of Part B of Schedule 3 (the "**DLI Earn-Out**").
- 5.2 DLI hereby covenants to pay the Purchaser an amount equal to the amount payable by any member of the Purchaser Group in respect of income tax attributable to DLI or any employees' national insurance contributions, but excluding any employer's national insurance contributions, for which any member of the Purchaser Group becomes liable as a result of the payment of the DLI Earn-Out.

6. SJ Earn-Out

- As additional consideration for the sale by SJ of his Set of Shares in Geneva Industries Limited, SJ will on Closing be issued Loan Notes in accordance with the provisions of Part C of Schedule 3 (the "SJ Earn-Out").
- 6.2 SJ hereby covenants to pay the Purchaser an amount equal to the amount payable by any member of the Purchaser Group in respect of income tax attributable to SJ or any

14

employees' national insurance contributions, but excluding any employer's national insurance contributions, for which any member of the Purchaser Group becomes liable as a result of the payment of the SJ Earn-Out.

- As additional consideration for the sale by DMJ of each of his Sets of Shares in the Target Companies, DMJ will on Closing be issued Loan Notes in accordance with the provisions of Parts D and E of Schedule 3 (respectively the "**DMJ Willen Earn-Out**", and the "**DMJ Geneva Earn-Out**", together the "**DMJ Earn-Outs**").
- 7.2 DMJ hereby covenants to pay the Purchaser an amount equal to the amount payable by any member of the Purchaser Group in respect of income tax attributable to DMJ or any employees' national insurance contributions, but excluding any employer's national insurance contributions, for which any member of the Purchaser Group becomes liable as a result of the payment of the DMJ Earn-Outs.

8. Escrow Accounts

- 8.1 Subject to the provisions of this clause 8, the Escrow Amounts shall be retained in the Escrow Accounts for a period of 12 months from the Closing Date (the "Escrow Period"). At the end of the Escrow Period (subject to this clause 8) the Sellers' Representative and the Purchaser shall, without delay, give joint written instructions to the Escrow Agent to release to each of the Sellers on the Escrow Payment Date, the funds in their respective Escrow Accounts (together with any Income, as defined below).
- 8.2 The funds in the Escrow Accounts shall be held on the following terms:
 - (a) any bank or other charges arising on each Escrow Account shall be charged to, and settled between, the Purchaser (as to 50 per cent.) and the relevant Seller (as to 50 per cent.); and
 - (b) any interest or profit generated on the Escrow Accounts (subject to any deduction of Tax at source or any bank or other charges properly charged to the Escrow Accounts) (the "**Income**") shall accrue to and form part of the Escrow Amounts. Each time part of the funds in the Escrow Accounts is paid out it shall have added to it the corresponding proportion of the Income.
- 8.3 If, in the event that upon final agreement or determination of the Working Capital Adjustment in accordance with the provisions of Schedule 2, any of DMJ, SJ and / or DLI are found to be liable to make a payment to the Purchaser, the Purchaser and the Sellers' Representative shall, without delay, issue joint written instructions to the Escrow Agent to release an amount equal to the amount of such outstanding payment, multiplied by the Relevant Working Capital Percentage (or, if funds available at the time in any of the relevant Escrow Account(s) are insufficient to discharge such outstanding payment, the amount then standing to the credit of the relevant Escrow Accounts), from the relevant Escrow Account to the Purchaser.
- 8.4 If a Notice of Claim (as defined in paragraph 1 of Schedule 9) has been delivered by the Purchaser in respect of a Claim prior to the expiry of the Escrow Period, then:

15

- (a) if such Claim or a breach of indemnity is finally disposed of and there is a Due Amount on such Claim, the Purchaser and the Sellers' Representative, following such settlement shall, without delay, issue joint instructions to the Escrow Agent (in writing) to pay to the Purchaser out of the relevant Escrow Account(s) the Due Amount multiplied by the Relevant Claim Percentage (or, if funds available at the time are insufficient to discharge such Claim, the amount then standing to the credit of the relevant Escrow Account(s) from the relevant Escrow Account(s) to the Purchaser); or
- (b) if, prior to the expiry of the Escrow Period, one or more Outstanding Claims (as defined in clause 8.11(b)) remains, then an amount equal to the aggregate amount of all the Outstanding Claims shall continue to be held in the relevant Escrow Account(s) and be dealt with in accordance with this clause 8.
- 8.5 Any balance of the relevant Escrow Account(s) and all interest accrued or due thereon, after deduction of all amounts to be paid in accordance with clauses 8.4(a) and, subject to clause 8.4(b), shall be released, without delay, to the Sellers on the Escrow Payment Date save that if any of the Sellers believe that a Claim or part thereof has no reasonable prospect of succeeding or is frivolous or vexatious then the relevant Sellers may give written notice to the Purchaser of such belief. Following such written notice, either:
 - (a) the Purchaser and the Sellers' Representative each acting reasonably may agree to refer the matter in dispute to a barrister who is counsel of at least 10 years' call in commercial chambers or tax chambers (as appropriate); or
 - (b) in the absence of such agreement within 14 days of a request by the Purchaser or the Sellers' Representative, to a barrister who is counsel of at least 10 years' call in commercial chambers or tax chambers (as appropriate) nominated (at the request of the Purchaser or the Sellers' Representative) by the President of the Law Society of England and Wales,

(the "Counsel").

- Where called upon, the Counsel shall be instructed to issue an unequivocal opinion as to whether the relevant Claim or part thereof would be struck out on the basis that it has no reasonable prospect of succeeding or is frivolous or vexatious (the "**Opinion**").
- 8.7 If the Opinion is that the Claim in question, or any part of it, would be so struck out on the basis that it has no reasonable prospect of succeeding or is frivolous or vexatious (a "**Strikeable Claim**") then such portion of the Escrow Amount as relates to the Strikeable Claim shall be released, without delay, to the relevant Seller(s) out of the relevant Escrow Account (provided that it is not otherwise required to cover any other Outstanding Claim).
- 8.8 The fees of the Counsel shall be borne by the Purchaser if the relevant Claim is a wholly Strikeable Claim. If only part of the relevant Claim is a Strikeable Claim (the

"Strikeable Portion") the fees of the Counsel in respect of the Strikeable Portion shall be borne by Purchaser and the outstanding balance of the fees of the Counsel shall be borne by the relevant Seller(s) *pro rata* to their Relevant Claim Percentage.

- 8.9 If at any time after the Escrow Payment Date:
 - (a) it shall be finally disposed of that the relevant Sellers are not liable (in whole or in part) to the Purchaser in respect of any relevant Claim; or
 - (b) the Purchaser has agreed to withdraw (in whole or in part) a relevant Claim,

then the Purchaser and the Sellers' Representative shall, without delay, issue joint written instructions to the Escrow Agent to release to the relevant Seller(s), such amounts from the relevant Escrow Account(s) (provided that such amounts are not otherwise required to cover any other Outstanding Claim). Notwithstanding any other provision of this clause 8 the Purchaser shall not be entitled to continue to withhold or procure the withholding of any Escrow Amount after the Escrow Payment Date in respect of a relevant Claim unless proceedings in respect of such Claim shall have been commenced on or before six (6) months from the date on which the relevant Claim is notified to the Sellers in accordance with paragraph 1 of Schedule 9 and if no such proceedings have been commenced before the expiry of such period of six (6) months then the Purchaser and the Sellers' Representative shall promptly issue joint written instructions to the Escrow Agent to release to the relevant Seller(s) the amount held in the relevant Escrow Account(s) in relation to the relevant Claim.

- As soon as practicable following the date an Outstanding Claim is finally disposed of after the expiry of the Escrow Period in respect of which there is a Due Amount and a sum has been retained in the Escrow Account(s) pursuant to clause 8.4(b), the Purchaser and the relevant Seller(s) shall, without delay, issue joint written instructions to the Escrow Agent to pay the Purchaser out of the relevant Escrow Account(s) the lesser of the Due Amount and the amount standing to the credit of the relevant Escrow Account(s) by way of satisfaction in whole or in part (provided that such amounts are not otherwise required to cover any other Outstanding Claim).
- 8.11 For the purposes of this clause 8:
 - (a) proceedings against a Seller or Sellers shall not be deemed to have been commenced by the Purchaser unless they shall have been both issued and served on such Seller(s);
 - (b) "Outstanding Claims" means at any time, those Claims in respect of which a Notice of Claim has been delivered by the Purchaser on or before the expiry of the Escrow Period (but which have not been finally disposed of); and
 - (c) "Due Amount" means the amount (if any) due to the Purchaser on a relevant Claim being finally disposed of in favour of the Purchaser.
- 8.12 A Claim shall be "**finally disposed of**" if:

17

- (a) the relevant Seller(s) and the Purchaser so agree in writing; or
- (b) the relevant Claim has been determined by a court of competent jurisdiction from which there is no right of appeal as from whose judgment the Purchaser or the relevant Seller(s) (as the case may be) are debarred by passage of time or otherwise from making an appeal.
- 8.13 For the avoidance of doubt, if the amount in the Escrow Account of a Seller is insufficient to discharge such Seller's liability or liabilities pursuant to this Agreement, such Seller shall remain liable to the Purchaser for any shortfall.
- 8.14 all cash sums payable to each of the Sellers shall be paid by telegraphic transfer from the Escrow Accounts pursuant to this clause 8 to the account of the Sellers' Solicitors at:

Bank: Barclays Bank plc

Address: Edgware Branch

126 Station Road

Edgware

Middlesex HA8 7RY

Sort code: 20 29 37

Account number: 00418293

Account name: Levine Mellins Klarfeld Client Account

IBAN: GB 75 BARC 202 937 004 18293

The Sellers' Solicitors are hereby authorised by each of the Sellers to receive sums transferred on behalf of them and the receipt by the Sellers' Solicitors of any sum transferred to them hereunder shall constitute a good receipt by the Sellers of the relevant sum from the Purchaser and the Purchaser shall have no obligation as to the distribution or allocation of such sums among the Sellers.

- 8.15 All cash sums payable to the Purchaser from the Escrow Accounts pursuant to this clause 8 shall be paid by telegraphic transfer to such account as the Purchaser may from time to time notify the Sellers' Representative.
- 8.16 All payments pursuant to this clause 8 shall be made free of any set-off, counterclaim or withholding whatsoever.

8.17 Both the Sellers' Representative and the Purchaser undertake to issue instructions for payment from the Escrow Accounts of the amounts due under this clause 8, without delay.

9. Pre-Closing Seller undertakings

9.1 From the date of this Agreement until Closing, the Seller shall comply with the obligations set out in Schedule 4.

18

10. Closing

- 10.1 Closing shall take place at the offices of the Purchaser's Solicitors on the Closing Date.
- 10.2 At Closing each of the Sellers and the Purchaser shall deliver or perform (or ensure that there is delivered or performed) all those documents, items and actions respectively listed in relation to that Party (or any of its Affiliates, if applicable) in Schedule 5.
- 10.3 The Parties acknowledge that immediately after Closing (or such later time as the Purchaser may determine) the Purchaser intends to undertake certain steps which may constitute a "relevant transfer" for the purposes of the Regulations. The Sellers agree that they will procure (insofar as they are able to do so) that each of the Target Companies will cooperate with the Purchaser: (a) by following the Purchaser's reasonable directions relating to the compliance by the relevant Target Companies with Regulation 13 of the Regulations; and (b) by providing "employee liability information" to the relevant transferee in accordance with Regulation 11 of the Regulations.

11. Warranties, Undertakings and Indemnities

- 11.1 Each of the Sellers severally warrants to the Purchaser in respect of each Set of Shares set out against his name in column 2 of Schedule 1 that he is the sole owner of his respective Set of Shares and he has and will at Closing have the right to sell and transfer full legal and beneficial title and ownership to and of his respective Shares free from all Third Party Rights.
- Each of the Sellers severally warrants to the Purchaser in respect of each Set of Shares set out against his name in column 2 of Schedule 1 and the Business of the Target Company to which such Set of Shares relates, that each of the Warranties is true, accurate and not misleading at the date of this Agreement. For the avoidance of doubt:
 - (a) DLI will not be liable for any Claim relating directly or indirectly to affairs, assets or any other matters whatsoever relating to Geneva Industries Limited and/or Geneva Auctions Limited;
 - (b) SJ will not be liable for any Claim relating directly or indirectly to the assets, affairs or any other matters whatsoever relating to Willen Trading Limited and/or Geneva Auctions Limited; and
 - (c) DMD will not be liable for any Claim relating directly or indirectly to the assets, affairs or any other matters whatsoever relating to Geneva Industries Limited and/or Willen Trading Limited.
- 11.3 The Closing Warranties shall be deemed to be repeated immediately before Closing by reference to the facts and circumstances then existing as if references in the Closing Warranties to the date of this Agreement were references to the Closing Date.
- Each Warranty shall be separate and independent and (except as expressly otherwise provided) no Warranty shall be limited by reference to any other Warranty.

19

- 11.5 Any claim by the Purchaser in connection with the Warranties shall be subject to the provisions of this clause 11 and Schedule 9. The Sellers acknowledge that the Purchaser is entering into this Agreement on the basis of the Warranties.
- 11.6 Each Seller agrees and undertakes to the Purchaser (for that Seller himself and as agent for each individual and entity referred to in this clause 11.6) that, except in the case of fraud, he has no rights against and shall not make any claim against any present or former employee, director, agent or officer of any Target Company or any member of the Purchaser Group on whom he may have relied before agreeing any term of or before entering into this Agreement or any other Transaction Document (including in relation to any information supplied or omitted to be supplied by any such person in connection with the Warranties, this Agreement or any other Transaction Document).
- 11.7 Each Seller severally indemnifies the Purchaser (on an after-Tax basis) if, on Closing there is any:
 - (a) Financial Debt; or
 - (b) amount owing by a Target Company to any of the Sellers or any person connected to a Seller (other than amounts properly due to a Seller under his contract of employment or any other debts owing between the Target Companies).
- 11.8 Each Seller undertakes to the Purchaser that, in addition to any amounts recoverable for a breach of any Warranty, it shall, in addition, pay or procure, on demand all costs and expenses suffered or incurred by the Purchaser or any of its Affiliates (including any Target Company), directly as a result of or in connection with the breach of Warranty, multiplied by the Relevant Claim Percentage.

12. Purchaser Warranties

12.1 The Purchaser warrants to the Sellers as at the date of this Agreement in the terms of the warranties set out in Schedule 8.

13. Conduct of Purchaser Claims

- 13.1 If the Purchaser becomes aware of any claim or potential claim by a third party (a "**Third Party Claim**") after Closing which is likely to result in a Non-Tax Claim being made under this Agreement, the Purchaser shall as soon as reasonably practicable give notice of the Third Party Claim to the Sellers' Representative and (subject to the Purchaser and each member of the Purchaser Group being indemnified (on an after-Tax basis) to the Purchaser's reasonable satisfaction by the Sellers against all reasonable out-of-pocket costs and expenses, including those of its legal advisers, incurred in respect of that Third Party Claim) shall, and shall procure that the relevant Target Company shall:
 - (a) ensure that the Sellers (as applicable) are given all reasonable facilities to investigate the Third Party Claim;

20

- (b) not admit liability or make any agreement or compromise with any person, body or authority in relation to that Third Party Claim without prior consultation with the Sellers (as applicable); and
- (c) (subject to the Purchaser being entitled to employ its own legal advisers) take any action that the Sellers reasonably request to avoid, resist, dispute, appeal, compromise or defend that Third Party Claim.
- 13.2 The rights of the Sellers under clause 13.1 shall only apply to a Third Party Claim if any of the Sellers (or the Sellers' Representative) gives notice to the Purchaser in writing of its intention to exercise its rights within 10 Business Days of the Purchaser giving notice of the Third Party Claim. If the Sellers (or the Sellers' Representative) do not give notice during that period, the Purchaser shall be entitled in its absolute discretion to settle, compromise, or resist any action, proceedings or claim against any member of the Purchaser Group out of which that Third Party Claim may arise.
- 13.3 Neither the Purchaser nor any of its Affiliates shall be required to take any action or refrain from taking any action pursuant to clause 13.1 if the action or omission requested would, in the reasonable opinion of the Purchaser, be materially prejudicial to the business of the Purchaser or any of its Affiliates.
- 13.4 The Purchaser shall not be precluded from bringing any Non-Tax Claim under this Agreement by reason of any breach of the terms of this clause 13.

14. Purchaser's rights to terminate

- 14.1 The Purchaser may terminate this Agreement (other than the Surviving Provisions) by notice to the Sellers' Representative at any time before Closing if any of the following circumstances arises or occurs at any time before Closing, namely:
 - (a) any Material Adverse Change occurs;
 - (b) a material breach of any Warranty as given on the date of this Agreement;
 - (c) any Event occurs which would constitute a material breach of any of the Closing Warranties if they were repeated at any time before Closing by reference to the facts and circumstances then existing (on the basis that references in the Closing Warranties to the date of this Agreement were references to the relevant date);
 - (d) any material breach by any Seller of its obligations under this Agreement.
- 14.2 If the Purchaser terminates this Agreement under clause 14.1, neither Party (nor any of its Affiliates, if applicable) shall have any claim of any nature against the other Party (or any of its Affiliates) under this Agreement (except in respect of any rights and liabilities which have accrued before termination or under any of the Surviving Provisions).
- 14.3 The Sellers undertake to disclose promptly to the Purchaser in writing any breach, matter, event, condition, circumstance, fact or omission of which that Seller is or becomes aware that may give rise to a right of termination under clause 14.1.

21

15. Tax

- 15.1 The provisions of Schedule 7 shall apply in relation to Taxation.
- 15.2 The Tax Covenant shall come into effect at Closing.

16. Insurance

- 16.1 Until and including the Closing Date, the relevant Sellers shall ensure that each Target Company continues in force and comply with all policies of insurance in respect of the Target Companies.
- If any insured event occurs before Closing in relation to any Target Company, the relevant Sellers shall use all reasonable efforts to make recovery under the relevant policy prior to Closing. To the extent that recovery is made, the Seller shall, and shall procure that the relevant Target Company shall, ensure that the proceeds are dealt with in accordance with procedures normally observed by that Target Company.

17. Information, Records and Assistance Post-closing

- 17.1 For seven (7) years following the Closing Date:
 - (a) each Seller shall provide the Purchaser (at the Purchaser's cost) with reasonable access at reasonable times to (and the right to take copies of) the books, accounts, customer lists and all other records held by it after Closing to the extent that such books, accounts, customer lists and records relate to the Target Companies and to the period up to Closing (the "Records"). This obligation is subject to the provisions of clause 22 (Confidentiality); and
 - (b) none of the Sellers shall dispose of or destroy any of the Records without first giving the Purchaser at least two (2) months' notice of its intention to do so and giving the Purchaser a reasonable opportunity to remove and retain any of them (at the Purchaser's expense).

18. Protective Covenants Post-closing

- 18.1 The Sellers shall not (whether alone or jointly with another and whether directly or indirectly) carry on or be engaged or concerned or interested economically or otherwise in any manner in any Competing Business in the Protected Territories for three (3) years after the Closing Date. For this purpose:
 - "Competing Business" means a business which competes with any business carried on at any time during 12 months preceding the Closing Date by any Target Company; and

"Protected Territories" means:

- (a) England, Scotland, Wales, Northern Ireland, the Republic of Ireland and Germany; and
- (b) such areas which upon Closing or in the period of 12 months prior to Closing Date any of the Target Companies either (i) carried on the

22

Competing Business (or any part of it) to any material extent (ii) marketed or provided goods or services to any Customers or (iii) developed plans to commence the marketing of goods or services in connection with a Competing Business.

- 18.2 Nothing in this clause 18 shall prevent any of the Sellers, after Closing, from owning for investment purposes securities in any company dealt in on a stock exchange provided that they do not exceed five (5) per cent. in nominal value of the securities in that company (or of any class of its securities).
- 18.3 The Sellers shall not (whether alone, jointly with another, directly or indirectly), for three (3) years after Closing, offer to employ or seek to entice away from any Target Company or any member of the Purchaser Group, or conclude any contract for services with, any person who was employed by any Target Company in skilled or managerial work at any time during the 12 months ending on the Closing Date.
- 18.4 The Sellers shall not (whether alone, jointly with another, directly or indirectly), for three (3) years after Closing, seek to entice away from any Target Company or any member of the Purchaser Group, or conclude any contract for services with, any Customer during the 12 months ending on the Closing Date.
- Each of the undertakings in clauses 18.1 to 18.4 (inclusive) is given to the Purchaser and to each of its Affiliates. The Sellers acknowledge that each is an entirely independent restriction and is no greater than is reasonably necessary to protect the interests of the Purchaser and its Affiliates. If any such restriction shall be held void or unenforceable but would be valid if deleted in part or reduced in its application, then that restriction shall apply with such modifications as may be necessary to make it valid and effective.

19. Post-Closing Undertakings

19.1 The Sellers shall ensure that (i) all monies or other items belonging to the Purchaser, or Target Company or which should have properly been paid or provided to any of them in relation to the Business and (ii) all notices, correspondence, orders or enquiries to the extent they relate to the Business, which any Seller receives after Closing are promptly passed to the Purchaser.

20. Payments

- Any payment to be made pursuant to this Agreement by the Purchaser (or any member of the Purchaser Group) to any of the Sellers shall be made to the Sellers' Solicitors.
- Any payment to be made pursuant to this Agreement by a Seller to the Purchaser shall be made to the Purchaser's Solicitors. The Purchaser agrees to pay each member of the Purchaser Group that part of each payment to which it is entitled.
- 20.3 Payment under clause 20.1 and 20.2 shall be in immediately available funds by electronic transfer on the due date for payment. Receipt of the amount due shall be an effective discharge of the relevant payment obligation.

23

- 20.4 If any sum due for payment in accordance with this Agreement is not paid on the due date for payment, the person in default shall pay Default Interest on that sum from but excluding the due date to and including the date of actual payment calculated on a daily basis.
- The Purchaser's Solicitors are hereby authorised by the Purchaser to receive sums transferred on behalf of it and the receipt by the Purchaser's Solicitors of any sum transferred to them hereunder shall constitute a good receipt by the Purchaser of the relevant sum from the Sellers.

20.6 The Sellers' Solicitors are hereby authorised by the Sellers to receive sums transferred on behalf of any of them and the receipt by the Sellers' Solicitors of any sum transferred to them hereunder shall constitute a good receipt by the Sellers of the relevant sum from the Purchaser.

21. Announcements

- 21.1 Neither the Sellers nor the Purchaser (nor any of their respective Affiliates, if applicable) shall make any announcement or issue any circular in connection with the existence or the subject matter of this Agreement (or any other Transaction Document) without the prior written approval of the other (such approval not to be unreasonably withheld or delayed).
- 21.2 The restriction in clause 21.1 shall not apply to the extent that the announcement or circular is required by law, by any stock exchange or any regulatory or supervisory body or authority of competent jurisdiction, whether or not the requirement has the force of law. If this exception applies, the Party making the announcement or issuing the circular shall use its reasonable efforts to consult with the other Party in advance as to its form, content and timing.
- 21.3 Each Seller and the Purchaser agree to fully cooperate to prepare and release any public statement regarding this Agreement or the transactions contemplated hereby.

22. Confidentiality

22.1 For the purposes of this clause 22:

"Confidential Information" means:

- (a) (in relation to the obligations of the Purchaser) any information received or held by the Purchaser relating to the Sellers or, prior to Closing, any of the Target Companies; or
- (b) (in relation to the obligations of the Sellers) any information received or held by the Sellers relating to the Purchaser Group or, following Closing, any of the Target Companies; and
- (c) information relating to the provisions and subject matter of, and negotiations leading to, this Agreement and the other Transaction Documents,

and includes written information and information transferred or obtained orally, visually, electronically or by any other means.

24

- 22.2 Each of the Sellers and the Purchaser shall maintain Confidential Information in confidence and not disclose that Confidential Information to any person except (i) as this clause 22 permits or (ii) as the other Party approves in writing.
- 22.3 Clause 22.2 shall not prevent disclosure by a Party or its representatives to the extent it can demonstrate that:
 - (a) disclosure is required by law or by any stock exchange or any regulatory, governmental or antitrust body (including any Tax authority) having applicable jurisdiction (provided that the disclosing Party shall first inform the other Party of its intention to disclose such information and take into account the reasonable comments of the other Party);
 - (b) disclosure is of Confidential Information which was lawfully in the possession of that Party (in either case as evidenced by written records) without any obligation of secrecy prior to its being received or held;
 - (c) disclosure is of Confidential Information which has previously become publicly available other than through that Party's fault; or
 - (d) disclosure is required for the purpose of any arbitral or judicial proceedings arising out of this Agreement (or any other Transaction Document).

23. Assignment

- 23.1 Except as provided in this clause 23 or unless the Sellers and the Purchaser specifically agree in writing, no person shall assign, transfer, charge or otherwise deal with all or any of its rights under this Agreement nor grant, declare, create or dispose of any right or interest in it.
- 23.2 Subject to clause 23.4, the Purchaser may assign the benefit of this Agreement and/or of any other Transaction Document to which it is a Party (in whole or in part) to, and it may be enforced by, any Permitted Assignee as if it were the Purchaser under this Agreement. Any Permitted Assignee to whom an assignment is made in accordance with the provisions of this clause 23 may itself make an assignment as if it were the Purchaser under this clause 23. For this purpose, a "**Permitted Assignee**" means any member or members of the Purchaser Group and/or any third party or third parties.
- 23.3 Subject to clause 23.4, the Purchaser may assign its rights under this Agreement and/or under any other Transaction Document to which it is a Party by way of security to any bank(s) and/or financial institution(s) lending money or making other banking facilities available to the Purchaser for the acquisition of the Shares. The Purchaser acknowledges and agrees that the rights conferred on any such assignee shall only be exercisable at the same time as it exercises its security under such finance arrangements.
- 23.4 If the benefit of this Agreement and/or any other Transaction Document is assigned in accordance with clauses 23.2 or 23.3, then the liability of the Sellers under this Agreement or any Transaction Document shall be no greater and their rights against the Purchaser no less, than they respectively would have been had the Purchaser

remained the sole owner of the Shares and had retained solely the benefit of this Agreement and Transaction Documents.

24. Further Assurances

24.1 Each of the Sellers and the Purchaser shall perform (or procure the performance of) all further acts and things and execute and deliver (or procure the execution and delivery of) such further documents, as may be required by law or as may be necessary or required by the Purchaser to implement and give effect to this Agreement.

25. Costs

25.1 Except as otherwise provided in this Agreement, each of the Sellers and the Purchaser shall be responsible for its own costs, charges and other expenses incurred in connection with the Proposed Transactions.

26. Notices

- Any notice in connection with this Agreement shall be in writing in English and delivered by hand, fax, registered post or courier using an internationally recognised courier company. A notice shall be effective upon receipt and shall be deemed to have been received (i) at the time of delivery, if delivered by hand, registered post or courier or (ii) at the time of transmission if delivered by fax provided that, in either case, where delivery occurs outside Working Hours, notice shall be deemed to have been received at the start of Working Hours on the next following Business Day.
- 26.2 The addresses and fax numbers of the Parties for the purpose of clause 26.1 are:

	Address:	Fax:
David Mark Jacobs	Hollybush Farm Botley Road Chesham Buckinghamshire HP5 1XG	N/A
Simon Jacobs	44 Marsworth Avenue Pinner Middlesex HA5 4TT	N/A
Darren Lee Innocent	Bakers Farm, High Street Shenington Buckinghamshire MK16 9NU	N/A
Darren Malcolm Dorrington	34 Saltwood Avenue Kingsmead Milton Keynes MK4 4HP	N/A
	26	

Address: Fax:

Liquidity Services Limited Unit 4, Prince Rupert +44 1249 709 291

For the attention of:

House

Holger Schwarz, Bill

Angrick and James

Williams

Chippenham

Wiltshire

SN14 6LH

Liquidity Services, Inc. 1920 L St. NW +1 202 558 6279

For the attention of: 6th Floor

Bill Angrick and James Washington DC 20036

Williams USA

With a copy to: Liquidity Services
Limited (address above)

27. Conflict with other Agreements

27.1 If there is any conflict between the terms of this Agreement and any other agreement, this Agreement shall prevail (as between the Parties to this Agreement and as between any of the Sellers and any members of the Purchaser Group) unless (i) such other agreement expressly states that it

overrides this Agreement in the relevant respect and (ii) the Sellers and the Purchaser are either also parties to that other agreement or otherwise expressly agree in writing that such other agreement shall override this Agreement in that respect.

28. Whole Agreement

- 28.1 This Agreement and the other Transaction Documents set out the whole agreement between the Parties in respect of the sale and purchase of the Shares and the Businesses and supersede any prior agreement (whether oral or written) relating to the Proposed Transactions. It is agreed that:
 - (a) no Party shall have any claim or remedy in respect of any statement, representation, warranty or undertaking, made by or on behalf of any other Party in relation to the Proposed Transactions which is not expressly set out in this Agreement or any other Transaction Document; and
 - (b) except for any liability in respect of a breach of this Agreement or any other Transaction Document, no Party shall owe any duty of care or have any liability in tort or otherwise to any other Party in relation to the Proposed Transactions.
- 28.2 This clause 28 shall not exclude any liability for, or remedy in respect of, fraudulent misrepresentation.

27

29. Waivers, Rights and Remedies

- 29.1 No failure or delay by any Party to this Agreement in exercising any right or remedy provided by law or under this Agreement or any Transaction Document shall impair such right or remedy or operate or be construed as a waiver or variation of it or preclude its exercise at any subsequent time and no single or partial exercise of any such right or remedy shall preclude any further exercise of it or the exercise of any other remedy.
- The rights and remedies of the Purchaser under this Agreement shall not be affected, and the liabilities of the Sellers under this Agreement shall not be released, discharged or impaired by (i) Closing, (ii) any knowledge held or gained of any such affairs by or on behalf of the Purchaser (except, in respect of the Warranties only, for matters fairly and specifically Disclosed in the Disclosure Letter and Disclosure Bundle), (iii) subject to compliance with the notice requirements in Schedule 9, the expiry of any limitation period prescribed by law in relation to a Claim, or (iv) any event or matter, other than a specific and duly authorised written waiver or release by the Purchaser.

30. Guarantees and Indemnities

- 30.1 In consideration of the Sellers entering into this Agreement, the Purchaser Parent guarantees to the Sellers and their successors, transferees and assigns the due and punctual payment to the Sellers on demand of all or any of the Guaranteed Obligations.
- 30.2 If the Guaranteed Obligations are not recoverable from the Purchaser by reason of illegality, incapacity, lack of or exceeding of powers, invalidity of execution or any other reason, the Purchaser Parent shall remain liable under this clause 30 for the Guaranteed Obligations as if it were a principal debtor.
- 30.3 The Purchaser Parent as principal obligor agrees to indemnify the Sellers (on an after Tax basis) in full and on demand from and against all and any losses, costs, claims, liabilities, damages, demands and expenses suffered or incurred by the Sellers arising out of, or in connection with, any failure of the Purchaser to perform or discharge the Guaranteed Obligations.
- 30.4 This clause 30 is and shall at all times be a continuing security and shall cover the ultimate balance of all monies payable under this Agreement.
- 30.5 The liability of the Purchaser Parent under this guarantee shall not be reduced, discharged or otherwise adversely affected by anything done or omitted by any person which, but for this provision, might operate or exonerate or discharge the Purchaser Parent or otherwise reduce or extinguish its liability under this clause 30.
- 30.6 This guarantee shall be in addition to and independent of all other security which Sellers may hold from time to time in respect of the discharge and performance by Purchaser of the Guaranteed Obligations.

28

31. Effect of Closing

31.1 Notwithstanding Closing (i) each provision of this Agreement and any other Transaction Document not performed at or before Closing but which remains capable of performance (ii) the Warranties and (iii) all covenants, indemnities and other undertakings and assurances contained in or entered into pursuant to this Agreement or any other Transaction Document, will remain in full force and effect and (except as otherwise expressly provided) without limit in time.

32. Counterparts

32.1 This Agreement may be executed in any number of separate counterparts, each of which is an original but all of which taken together shall constitute one and the same instrument.

33. Variations

33.1 No amendment of this Agreement (or of any other Transaction Document) shall be valid unless it is in writing and duly executed by or on behalf of all of the Parties to it.

34. Invalidity

Each of the provisions of this Agreement and the other Transaction Documents is severable. If any such provision is held to be or becomes invalid or unenforceable in any respect under the law of any jurisdiction, it shall have no effect in that respect and the Parties shall use all reasonable efforts to replace it in that respect with a valid and enforceable substitute provision the effect of which is as close to its intended effect as possible.

35. No third party enforcement rights

35.1 Except where expressly provided in this Agreement, a person who is not a Party to this Agreement shall have no right under the Contracts (Rights of Third Parties) Act 1999.

36. Governing law and jurisdiction

- 36.1 This Agreement shall be governed by, and interpreted in accordance with, English law.
- 36.2 Except as expressly provided otherwise in this Agreement, the courts of England are to have exclusive jurisdiction to settle any disputes (including claims for set-off and counterclaims) which may arise in connection with the creation, validity, effect, interpretation or performance of, or the legal relationships established by, this Agreement or otherwise arising in connection with this Agreement. For such purposes each Party irrevocably submits to the jurisdiction of the English courts.
- 36.3 The Parties irrevocably waive any objections to the jurisdiction of any court referred to in this clause and irrevocably agree that a judgment or order of any court referred to in this clause in connection with this Agreement is conclusive and binding on them and may be enforced against it in the courts of any other jurisdiction.

29

- 36.4 The Parties irrevocably consent to service of process or any other documents in connection with proceedings in any court by facsimile transmission, personal service, delivery at any address specified in this Agreement or any other usual address, mail or in any other manner permitted by the law of the place of service or the law of the jurisdiction where proceedings are instituted.
- 36.5 The Purchaser Parent hereby appoints the Purchaser as its agent for service of process in England and the Purchaser has confirmed in writing to each of the Parties its acceptance of such appointment. All correspondence with the Purchaser (acting in its capacity as agent for service of process) shall be marked for the attention of Darryl Mark Aylett and shall be delivered in accordance with clause 26; and
- 36.6 If for any reason an agent appointed under clause 36.5 ceases to act as such, the Purchaser Parent shall promptly appoint another such agent and promptly notify the Sellers of the appointment and the new agent's name and address. If the Purchaser Parent does not make such an appointment within seven (7) days of such cessation, then the Sellers may do so on their behalf and at their cost and shall notify the Purchaser Parent if they do so.

30

Schedule 1

SELLERS AND PRICE

(1) Seller Name	(2) Shares in Target Companies	1	(3) Initial Share Prices	J	(4) Escrow Amounts		(5) Amount of the nitial Share Prices wayable at Closing	(6) Relevant Claim Percentage	(7) Relevant Working Capital Percentage
David Mark Jacobs	68 Ordinary Shares of £1.00 each in the capital of Geneva Industries Limited	£	4,727,209	£	472,720	£	4,254,489	68%	68%
	100 Ordinary Shares of £1.00 each in the capital of Willen Trading Limited	£	556,000	£	56,000	£	500,000	50%	50%
	50 Ordinary Shares of £1.00 each in the capital of Geneva Auctions Limited	£	278,000	£	28,000	£	250,000	50%	N/A
Simon Jacobs	32 Ordinary Shares of £1.00 each in the capital of Geneva Industries Limited	£	2,224,569	£	222,456	£	2,002,113	32%	32%
Darren Malcolm Dorrington	50 Ordinary Shares of £1.00 each in the capital of Geneva Auctions Limited	£	278,000	£	28,000	£	250,000	50%	N/A
Darren Lee Innocent	100 Ordinary Shares of £1.00 each in the capital of Willen Trading Limited	£	556,000	£	56,000	£	500,000	50%	50%

Schedule 2

POST-CLOSING FINANCIAL ADJUSTMENTS

Part A: Preliminary

1. In preparing the Closing Statement:

- the items and amounts to be included in the calculation of Working Capital for the purposes of the Closing Statement shall be identified by applying the definitions of Working Capital and Closing Statement in clause 1 (subject, where applicable, to the provisions of Part A of this Schedule);
- 1.2 in applying each such definition and the provisions of Part A of this Schedule and determining which items and amounts are to be included in the Closing Statement, if and to the extent that the treatment or characterisation of the relevant item or amount or type or category of item or amount:
 - (a) is dealt with in the specific accounting treatments set out in Part B of this Schedule (the "**Specific Accounting Treatments**"), the relevant Specific Accounting Treatment(s) shall apply;
 - (b) is not dealt with in the Specific Accounting Treatments but is dealt with in the accounting principles, policies, treatments, practices and categorisations used in the preparation of the Last Accounts (the "Accounting Principles"), the applicable Accounting Principle(s) shall apply (including in relation to the exercise of accounting discretion and judgement); and
 - (c) is not dealt with in either the Specific Accounting Treatments or the Accounting Principles, UK GAAP shall apply.

32

Part B: Specific Accounting Treatments

The following Specific Accounting Treatments shall apply in the preparation of the Closing Statement in respect of each of Geneva Industries and Willen Trading Limited:

WORKING CAPITAL

1. Stock balances

- 1.1 Stocks of inventory held for resale, storeroom materials and consumables held at the Closing Date will be valued at the lower of (i) their cost and (ii) their realisable value, to the extent that such stocks exist at the Closing Date and are reasonably expected to be used in the business after the Closing Date.
- 1.2 Provisions for stock made in the Closing Statement shall not exceed:
 - (a) £93,386 in respect of Geneva Industries Limited; and
 - (b) £93,630 in respect of Willen Trading Limited.

2. Trade debtors

- 2.1 The trade debtors balance in the calculation of Working Capital will reflect invoices issued on or before the Closing Date to the extent that they have not been paid on or before the Closing Date.
- 2.2 Provisions for trade debtors made in the Closing Statement shall not exceed £109,764 in respect of Geneva Industries Limited.

3. Prepayments, accrued income and other debtors

- 3.1 Prepayments shall be included in the calculation of Working Capital in respect of amounts paid on or before the Closing Date in respect of goods and services to be received after the Closing Date.
- 3.2 There shall be included in the calculation of Working Capital an accrual for corporation tax.
- 3.3 The calculation of insurance, rates and licence prepayments in the calculation of Working Capital at the Closing Date will be made by reference to the total amounts paid on or before the Closing Date pro-rated between the periods before and after the Closing Date on a daily basis.

4. Trade creditors and accruals

In this Part B, "**month**" means a period (of either four (4) or five (5) weeks ending on the last Sunday of each calendar month) for which each of Geneva Industries Limited and Willen Trading Limited prepares its Management Accounts. Except for accruals/provisions which are over six (6) months old at the Closing Date and which are due for release in accordance with past practice, there shall be no release of that

provision or accrual unless (i) cash has been paid on or before the Closing Date or (ii) subsequent external evidence/documentation has been received to indicate that the provision/accrual should be reduced.

- 4.2 Except as provided elsewhere in this Schedule 2, trade creditors and accruals shall be included in the calculation of Working Capital to the extent that a liability exists and is still outstanding at the Closing Date in respect of goods and services that have been received or supplied on or before the Closing Date.
- 4.3 Where a purchase order has been raised and placed with suppliers/contractors in respect of capital expenditure projects on or before the Closing Date, a provision shall be included in the calculation of Working Capital to the extent it has not otherwise been paid, accrued or provided for.
- 4.4 The calculation of Working Capital shall include balances:
 - (a) for payments on account in respect of amounts invoiced and received in Cash on or before the Closing Date from customers in respect of shows and events being reported after the Closing Date in accordance with the existing revenue recognition policies of each of Geneva Industries and Willen Trading as applied in the preparation of the Management Accounts;
 - (b) for amounts invoiced on or before the Closing Date in respect of services to be supplied after the Closing Date (such as rents invoiced for future periods).

Reserves

5.1 All reserves will be deducted from the calculation of Working Capital.

Part C: Closing Statement

The Purchaser shall, or shall procure that the Purchaser's accountants shall, after Closing prepare a draft statement (the "Closing Statement") showing the Working Capital of each of Geneva Industries Limited and Willen Trading Limited.

The Closing Statement shall be in the form set out in Exhibit 2 and incorporate separate statements in the form set out in that Exhibit showing the calculation of the Working Capital of each of Geneva Industries Limited and Willen Trading Limited. The Purchaser shall deliver the draft Closing Statement to the Sellers within 45 days after Closing.

- 2. No individual adjustment for an amount of less than £5,000 shall be made to the draft Closing Statement as a result of the process described in this Part C unless the Parties agree otherwise. Where more than one adjustment arises from the same or similar subject matter, facts, events or circumstances, those adjustments may be aggregated and shall together be treated as an individual adjustment for this purpose.
- 3. The Sellers' Representative shall notify the Purchaser in writing (an "**Objection Notice**") within 30 days after receipt whether or not it accepts the draft Closing Statement for the purposes of this Agreement. An Objection Notice shall set out in detail the reasons for such non-acceptance and specify the adjustments which, in the

34

opinion of the Sellers (or any one of them), should be made to the draft Closing Statement in order for it to comply with the requirements of this Agreement. Except for the matters specifically set out in the Objection Notice, the Sellers shall be deemed to have agreed the draft Closing Statement in full.

- 4. If the Sellers' Representative serves an Objection Notice in accordance with paragraph 3, the Purchaser and the relevant Seller(s) shall meet and discuss the objections set out in the Objection Notice and to agree the adjustments (if any) required to be made to the draft Closing Statement, in each case within 30 days after receipt by the Purchaser of the Objection Notice.
- 5. If the Sellers are satisfied with the draft Closing Statement (either as originally submitted or after adjustments agreed between the Seller and the Purchaser pursuant to paragraph 4) or if the Seller fails to give a valid Objection Notice within the 30 day period referred to in paragraph 3, then the draft Closing Statement (incorporating any agreed adjustments) shall constitute the Closing Statement for the purposes of this Agreement.
- If the Sellers and the Purchaser do not reach agreement within 30 days of receipt by the Purchaser of the Objection Notice, then the matters in dispute shall be referred (on the application of either the Sellers' Representative or the Purchaser) for determination by an independent firm of chartered accountants of international standing as the Sellers and the Purchaser shall agree or, failing agreement, appointed by the President for the time being of the Institute of Chartered Accountants in England and Wales (the "Firm"). The Firm shall be requested to make its decision within seven (7) days (or such later date as the Seller, the Purchaser and the Firm agree in writing) of confirmation and acknowledgement by the Firm of its appointment. The following provisions shall apply once the Firm has been appointed:
 - (a) the Sellers and the Purchaser shall each prepare a written statement within 15 days of the Firm's appointment on the matters in dispute which (together with the relevant supporting documents) shall be submitted to the Firm for determination and copied at the same time to the other;
 - (b) following delivery of their respective submissions, the Purchaser and the Sellers shall each have the opportunity to comment once only on the other's submission by written comment delivered to the Firm not later than 10 days after receipt of the other's submission and, thereafter, none of the Sellers or the Purchaser shall be entitled to make further statements or submissions except insofar as the Firm so requests (in which case it shall, on each occasion, give the other Party (unless otherwise directed) 10 days to respond to any statements or submission so made);
 - (c) in giving its determination, the Firm shall state what adjustments (if any) are necessary, solely for the purposes of this Agreement, to the draft Closing Statement in respect of the matters in dispute in order to comply with the requirements of this Agreement and to determine finally the Closing Statement;

binding on the Parties and, without prejudice to any other rights which they may respectively have under this Agreement, the Parties expressly waive, to the extent permitted by law, any rights of recourse they may otherwise have to challenge it.

- 7. Each Seller and the Purchaser shall each be responsible for their own costs in connection with the preparation, review and agreement or determination of the Closing Statement. The fees and expenses of the Firm shall be borne equally between the Sellers and the Purchaser or in such other proportions as the Firm shall determine.
- 8. To enable the Purchaser to meet its obligations under this Schedule 2, the Sellers shall provide to the Purchaser and the Purchaser's accountants full access to the books and records, employees and premises of the Geneva Industries Limited and Willen Trading Limited to the date that the draft Closing Statement is agreed or determined. If the Sellers' Representative serves an Objection Notice, it shall ensure that the Purchaser and the Purchaser's accountants shall be given reasonable access to the Sellers' accountants' working papers relating to the adjustments proposed in the Objection Notice and any other submissions by or on behalf of the Sellers in relation to the Closing Statement. The Sellers shall co-operate fully with the Purchaser and shall permit the Purchaser and/or the Purchaser's accountants to take copies (including electronic copies) of the relevant books and records and shall provide all assistance reasonably requested by the Purchaser to facilitate the preparation of the Closing Statement.
- 9. When the Closing Statement has been agreed or determined in accordance with the preceding paragraphs, then the amounts shown in the Closing Statement as the Working Capital for each Business shall be final and binding for the purposes of this Agreement.

Part D: Financial Adjustments

1. When the Closing Statement has been finally agreed or determined in accordance with this Schedule 2, the following adjustments shall be made to each Initial Share Price.

2. Working Capital

- 2.1 If the aggregate Working Capital for Geneva Industries Limited and Willen Trading Limited is:
 - (a) an amount greater than 110 per cent. of the Target Working Capital, then the Purchaser shall instruct its bank to pay to the relevant Seller(s) all amounts in excess of that 110 per cent, multiplied by such Seller's Relevant Working Capital Percentage; or
 - (b) an amount less than 90 per cent. of the Target Working Capital, then the Seller(s) shall pay to the Purchaser an amount by which the aggregate Working Capital falls below 90 per cent. of the Target Working Capital, multiplied by such Seller's Relevant Working Capital Percentage to the Purchaser.

36

3. General

- 3.1 Any payment required to be made pursuant to any of paragraph 2 inclusive of this Part D shall be paid by the Sellers or the Purchaser (as the case may be) together with interest on such payment at two (2) per cent. above the base rate from time to time of Barclays Bank plc for the period from (but excluding) the Closing Date to (and including) the due date for payment pursuant to the relevant clause, calculated on a daily basis.
- 3.2 The Sellers and Purchaser agree that, once the Closing Statement has been agreed or determined in accordance with the provisions of Part C of this Schedule 2, the sums which each is respectively obliged to pay pursuant to this Part D shall be aggregated and set off against each other. Whichever of the Sellers or Purchaser is then left with any payment obligation under this Part D shall make the applicable payment(s) (in the case of the Sellers, *pro rata* to their Relevant Working Capital Percentage) within five (5) Business Days of the date on which the Closing Statement is agreed or so determined. Any such payment shall be made in accordance with the provisions of clause 20.1 or 20.2 of this Agreement, as the case may be.

37

Schedule 3

EARN-OUT

Part A - Definitions and Interpretation

Definitions

- 1.1 The definitions and rules of interpretation in this paragraph apply in this Schedule 3:
 - "Calculation Years" means Year 1, Year 2 and Year 3, and any of them shall be known as a "Calculation Year";
 - "DMJ Earn-Outs" means the DMJ Geneva Earn-Out and the DMJ Willen Earn-Out;

"GAL Business" means the business carried on by the Purchaser and previously carried on by Geneva Auctions Limited up to the Closing Date;

"Geneva EBITDA" means the aggregated earnings of the GIL Business and the GAL Business for a relevant Calculation Year before all interest, Tax, depreciation and amortisation, as determined by reference to a profit and loss statement prepared by the Purchaser under the direction of the Purchaser Parent in respect of that Calculation Year, and having regard to the past accounting practices of the GAL Business and the GIL Business;

"GIL Business" means the business carried on by the Purchaser and previously carried on by Geneva Industries Limited up to the Closing Date;

"Relevant DLI Maximum Earn-Out Award" means the maximum amount of the Loan Notes redeemable by DLI in the relevant Calculation Year as set out opposite his name in columns 1-3 of Table 3 of the Exhibit to this Schedule;

"Relevant DMJ Maximum Earn-Out Award" means the maximum amount of the Loan Notes redeemable by DMJ in the relevant Calculation Year and for the relevant Target Company as set out opposite his name in columns 1-3 of Tables 3-4 of the Exhibit to this Schedule;

"Relevant SJ Maximum Earn-Out Award" means the maximum amount of the Loan Notes redeemable by SJ in the relevant Calculation Year as set out opposite his name in columns 1-3 of Table 4 of the Exhibit to this Schedule;

"Relevant Geneva EBITDA Target" means the aggregated EBITDA Target of the GAL Business and the GIL Business for the relevant Calculation Year as set out in columns 1-3 of Table 2 of the Exhibit to this Schedule;

"Relevant Willen EBITDA Target" means the EBITDA Target of the Willen Business for the relevant Calculation Year as set out in columns 1-3 of Table 1 of the Exhibit to this Schedule;

38

"Willen Business" means the business carried on by the Purchaser and previously carried on by Willen Trading Limited up to the Closing Date;

"Willen EBITDA" means the earnings of the Willen Business for a relevant Calculation Year before all interest, Tax, depreciation and amortisation, as determined by reference to a profit and loss statement prepared by the Purchaser under the direction of the Purchaser Parent in respect of that Calculation Year, and having regard to the past accounting practices of the Willen Business;

"Year 1" means the period of 12 months beginning on the Closing Date;

"Year 2" means the period of 12 months beginning at the end of Year 1; and

"Year 3" means the period of 12 months beginning at the end of Year 2.

1.2 Unless the contrary intention appears, words and expressions defined in this Agreement have the same meaning in this Schedule 3 and any provisions in this Agreement concerning matters of construction or interpretation also apply in this Schedule 3.

39

Part B - DLI Earn-Out

1. DLI Earn-Out

- 1.1 As additional consideration for the sale by DLI of the Shares in Willen Trading Limited set against his name in column 2 of Schedule 1, the Purchaser shall issue to DLI with effect from, and conditional upon, Closing, Loan Notes in the nominal amounts set opposite his name in columns 1-3 of Table 5 in the Exhibit to this Schedule.
- 1.2 For each Calculation Year, the "**DLI Earn-Out Entitlement**" shall be:
 - (a) £61,438, conditional upon achievement by the Purchaser of a Willen EBITDA of 90 per cent. of the Relevant Willen EBITDA Target; or
 - (b) conditional upon achievement by the Purchaser of a Willen EBITDA of more than 90 per cent. but less than 95 per cent. of the Relevant Willen EBITDA Target, an amount in pounds sterling equal to:

$$\left(\frac{x}{5} \times £30,729\right) + £61,438$$

where 'x' =
$$\left(\frac{\text{Willen EBITDA}}{\text{Relevant Willen EBITDA Target}} \times 100\right) - 90$$

rounded to the nearest three decimal places and being not more than 4.999; or

- (c) £92,167 conditional upon achievement by the Purchaser of a Willen EBITDA of 95 per cent. of the Relevant Willen EBITDA Target; or
- (d) conditional upon achievement by the Purchaser of a Willen EBITDA of more than 95 per cent. but less than 100 per cent. of the Relevant Willen EBITDA Target, an amount in pounds sterling equal to:

$$\left(\frac{x}{5} \times £92,166\right) + £92,167$$

where
$$'x' = \left(\frac{\text{Willen EBITDA}}{\text{Relevant Willen EBITDA Target}} \times 100\right) - 95$$

rounded to the nearest three decimal places and being not more than 4.999; or

- (e) £184,333 conditional upon achievement by the Purchaser of a Willen EBITDA of 100 per cent. or more, of the Relevant Willen EBITDA Target.
- 1.3 DLI hereby warrants that, in respect of each Calculation Year, the Willen EBITDA will not be less than the Relevant Willen EBITDA Target for that Calculation Year.

40

- 1.4 In the event of a breach of the warranty set out at paragraph 1.3 of this Part B, the Purchaser shall be entitled to set-off against and deduct from the amount payable by it in respect of the Loan Notes of the relevant series (to which the Calculation Year relates) an amount determined in accordance with paragraph 1.5 below, and DLI's liability in respect of such breach shall be fully discharged thereby.
- 1.5 The amount which the Purchaser shall be entitled to deduct from the amount payable by it in respect of the Loan Notes of the relevant series (to which the Calculation Year relates) shall be such amount as would, with accrued interest at the rate and on the basis set out in the Loan Note Instrument be, an amount equal to the Earn-Out Shortfall, as defined below.
- 1.6 For the purposes of paragraph 1.5 "Earn-Out Shortfall" means an amount equal to the difference between the Relevant DLI Maximum Earn-Out Award and the DLI Earn-Out Entitlement for that Calculation Year (calculated in accordance with paragraph 1.2 of this Part B.
- 1.7 Except as provided in paragraph 1.4 of this Part B, the Purchaser shall have no other remedy in respect of a breach of the warranty contained at paragraph 1.3 of this Part B.

41

Part C – SJ Earn-Out

1. SJ Earn-Out

- 1.1 As additional consideration for the sale by SJ of the Shares in Geneva Industries Limited set against his name in column 2 of Schedule 1, the Purchaser shall issue to SJ with effect from, and conditional upon, Closing, Loan Notes in the nominal amounts set opposite his name in columns 1-3 of Table 6 in the Exhibit to this Schedule.
- 1.2 For each of Year 1 and Year 2, the "SJ Earn-Out Entitlement" shall be:
 - (a) £20,000 conditional upon achievement by the Purchaser of a Geneva EBITDA of 95 per cent. of the Relevant Geneva EBITDA Target; or
 - (b) conditional upon achievement by the Purchaser of a Geneva EBITDA of more than 95 per cent. but less than 100 per cent. of the Relevant Geneva EBITDA Target, an amount in pounds sterling equal to:

$$\left(\frac{x}{5} \times £20,000\right) + £20,000$$

where
$$x' = \left(\frac{\text{Geneva EBITDA}}{\text{Relevant Geneva EBITDA Target}} \times 100\right) - 95$$

rounded to the nearest three decimal places and being not more than 4.999; or

(c) £40,000 conditional upon achievement by the Purchaser of a Geneva EBITDA of 100 per cent. or more, of the Relevant Geneva EBITDA Target,

and for Year 3, the "SJ Earn-Out Entitlement" shall be:

- (d) £32,000 conditional upon achievement by the Purchaser of a Geneva EBITDA of 95 per cent. of the Relevant Geneva EBITDA Target; or
- (e) conditional upon achievement by the Purchaser of a Geneva EBITDA of more than 95 per cent. but less than 100 per cent. of the Relevant Geneva EBITDA Target, an amount in pounds sterling equal to:

$$\left(\frac{x}{5} \times £32,000\right) + £32,000$$

where
$$'x' = \left(\frac{\text{Geneva EBITDA}}{\text{Relevant Geneva EBITDA Target}} \times 100\right) - 95$$

rounded to the nearest three decimal places and being not more than 4.999; or

(f) £64,000 conditional upon achievement by the Purchaser of a Geneva EBITDA of 100 per cent. or more, of the Relevant Geneva EBITDA Target.

42

- 1.3 SJ hereby warrants that, in respect of each Calculation Year, the Geneva EBITDA will not be less than the Relevant Geneva EBITDA Target for that Calculation Year.
- 1.4 In the event of a breach of the warranty set out at paragraph 1.3 of this Part C, the Purchaser shall be entitled to set-off against and deduct from the amount payable by it in respect of the Loan Notes of the relevant series (to which the Calculation Year relates) an amount determined in accordance with paragraph 1.5 below, and SJ's liability in respect of such breach shall be fully discharged thereby.
- 1.5 The amount which the Purchaser shall be entitled to deduct from the amount payable by it in respect of the Loan Notes of the relevant series (to which the Calculation Year relates) shall be such amount as would, with accrued interest at the rate and on the basis set out in the Loan Note Instrument be, an amount equal to the Earn-Out Shortfall, as defined below.
- 1.6 For the purposes of paragraph 1.5 of this Part C "Earn-Out Shortfall" means an amount equal to the difference between the Relevant SJ Maximum Earn-Out Award and the SJ Earn-Out Entitlement for that Calculation Year (calculated in accordance with paragraph 1.2 of this Part C.
- 1.7 Except as provided in paragraph 1.4 of this Part C, the Purchaser shall have no other remedy in respect of a breach of the warranty contained at paragraph 1.3 of this Part C.

43

Part D - DMJ Willen Earn-Out

1. DMJ Willen Earn-Out

- 1.1 As additional consideration for the sale by DMJ of the Shares in Willen Trading Limited set against his name in column 2 of Schedule 1, the Purchaser shall issue to DMJ with effect from, and conditional upon, Closing, Loan Notes in the nominal amounts set opposite his name in columns 1-3 of Table 5 in the Exhibit to this Schedule.
- 1.2 For each of Year 1 and Year 2, the "**DMJ Willen Earn-Out Entitlement**" shall be:
 - (a) £62,500 conditional upon achievement by the Purchaser of a Willen EBITDA of 95 per cent. of the Relevant Willen EBITDA Target; or
 - (b) conditional upon achievement by the Purchaser of a Willen EBITDA of more than 95 per cent. but less than 100 per cent. of the Relevant Willen EBITDA Target, an amount in pounds sterling equal to:

$$\left(\frac{x}{5} \times £62,500\right) + £62,500$$

where 'x' =
$$\left(\frac{\text{Willen EBITDA}}{\text{Relevant Willen EBITDA Target}} \times 100\right) - 95$$

rounded to the nearest three decimal places and being not more than 4.999; or

(c) £125,000 conditional upon achievement by the Purchaser of a Willen EBITDA of 100 per cent. or more, of the Relevant Willen EBITDA Target;

and for Year 3, the "DMJ Willen Earn-Out Entitlement" shall be:

- (d) £100,000 conditional upon achievement by the Purchaser of a Willen EBITDA of 95 per cent. of the Relevant Willen EBITDA Target; or
- (e) conditional upon achievement by the Purchaser of a Willen EBITDA of more than 95 per cent. but less than 100 per cent. of the Relevant Willen EBITDA Target, an amount in pounds sterling equal to:

$$\left(\frac{x}{5} \times £100,000\right) + £100,000$$

where
$$'x' = \left(\frac{\text{Willen EBITDA}}{\text{Relevant Willen EBITDA Target}} \times 100\right) - 95$$

(f) £200,000 conditional upon achievement by the Purchaser of a Willen EBITDA of 100 per cent. or more, of the Relevant Willen EBITDA Target.

44

- 1.3 DMJ hereby warrants that, in respect of each Calculation Year, the Willen EBITDA will not be less than the Relevant Willen EBITDA Target for that Calculation Year.
- 1.4 In the event of a breach of the warranty set out at paragraph 1.3 of this Part D, the Purchaser shall be entitled to set-off against and deduct from the amount payable by it in respect of the Loan Notes of the relevant series (to which the Calculation Year relates) an amount determined in accordance with paragraph 1.5 below, and DMJ's liability in respect of such breach shall be fully discharged thereby.
- 1.5 The amount which the Purchaser shall be entitled to deduct from the amount payable by it in respect of the Loan Notes of the relevant series (to which the Calculation Year relates) shall be such amount as would, with accrued interest at the rate and on the basis set out in the Loan Note Instrument be, an amount equal to the Earn-Out Shortfall, as defined below.
- 1.6 For the purposes of paragraph 1.5 of this Part D "Earn-Out Shortfall" means an amount equal to the difference between the Relevant DMJ Maximum Earn-Out Award and the DMJ Earn-Out Entitlement for that Calculation Year (calculated in accordance with paragraph 1.2 of this Part D.
- 1.7 Except as provided in paragraph 1.4, the Purchaser shall have no other remedy in respect of a breach of the warranty contained at paragraph 1.3 of this Part D.

45

Part E - DMJ Geneva Earn-Out

1. DMJ Geneva Earn-Out

- 1.1 As additional consideration for the sale by DMJ of the Shares in Geneva Auctions Limited and Geneva Industries Limited set against his name in column 2 of Schedule 1, the Purchaser shall issue to DMJ with effect from, and conditional upon, Closing, Loan Notes in the nominal amounts set opposite his name in columns 1-3 of Table 6 in the Exhibit to this Schedule.
- 1.2 For each of Year 1 and Year 2, the "**DMJ Geneva Earn-Out Entitlement**" shall be:
 - (a) £42,500 conditional upon achievement by the Purchaser of a Geneva EBITDA of 95 per cent. of the Relevant Geneva EBITDA Target; or
 - (b) conditional upon achievement by the Purchaser of a Geneva EBITDA of more than 95 per cent. but less than 100 per cent. of the Relevant Geneva EBITDA Target, an amount in pounds sterling equal to:

$$\left(\frac{x}{5} \times £42,500\right) + £42,500$$

where
$$'x' = \left(\frac{\text{Geneva EBITDA}}{\text{Relevant Geneva EBITDA Target}} \times 100\right) - 95$$

rounded to the nearest three decimal places and being not more than 4.999; or

£85,000 conditional upon achievement by the Purchaser of a Geneva EBITDA of 100 per cent. or more, of the Relevant Geneva EBITDA Target.

and for Year 3, the "DMJ Geneva Earn-Out Entitlement" shall be:

- (d) £68,000 conditional upon achievement by the Purchaser of a Geneva EBITDA of 95 per cent. of the Relevant Geneva EBITDA Target; or
- (e) conditional upon achievement by the Purchaser of a Geneva EBITDA of more than 95 per cent. but less than 100 per cent. of the Relevant Geneva EBITDA Target, an amount in pounds sterling equal to:

$$\left(\frac{x}{5} \times £68,000\right) + £68,000$$

where
$$x' = \left(\frac{\text{Geneva EBITDA}}{\text{Relevant Geneva EBITDA Target}} \times 100\right) - 95$$

rounded to the nearest three decimal places and being not more than 4.999; or

46

- 1.3 DMJ hereby warrants that, in respect of each Calculation Year, the Geneva EBITDA will not be less than the Relevant Geneva EBITDA Target for that Calculation Year.
- 1.4 In the event of a breach of the warranty set out at paragraph 1.3 of this Part E, the Purchaser shall be entitled to set-off against and deduct from the amount payable by it in respect of the Loan Notes of the relevant series (to which the Calculation Year relates) an amount determined in accordance with paragraph 1.5 below, and DMJ's liability in respect of such breach shall be fully discharged thereby.
- 1.5 The amount which the Purchaser shall be entitled to deduct from the amount payable by it in respect of the Loan Notes of the relevant series (to which the Calculation Year relates) shall be such amount as would, with accrued interest at the rate and on the basis set out in the Loan Note Instrument be, an amount equal to the Earn-Out Shortfall, as defined below.
- 1.6 For the purposes of paragraph 1.5 of this Part E "Earn-Out Shortfall" means an amount equal to the difference between the Relevant DMJ Maximum Earn-Out Award and the DMJ Earn-Out Entitlement for that Calculation Year (calculated in accordance with paragraph 1.2 of this Part E.
- 1.7 Except as provided in paragraph 1.4 of this Part E , the Purchaser shall have no other remedy in respect of a breach of the warranty contained at paragraph 1.3 of this Part E.

47

Part F – Target Company EBITDA Calculation Procedure

- 1. The Purchaser undertakes to DLI, DMJ and SJ that for the period from the Closing Date up to and including the end of Year 3:
 - (a) the Willen Business shall be carried on by the Purchaser as a separate division to the GAL Business and GIL Business;
 - (b) the GAL Business, the Willen Business and the GIL Business shall be carried out, so far as is practicable, in the ordinary course and substantially in the same manner that such business had been carried on prior to Closing;
 - (c) proper and correct accounting records of the GAL Business, the Willen Business and the GIL Business shall be kept and maintained;
 - (d) DMJ and SJ shall promptly be given all necessary information concerning the GAL Business and the GIL Business which is relevant for the purposes of calculating any sums due to them pursuant to this Schedule 3;
 - (e) DMJ and DLI shall promptly be given all necessary information concerning the Willen Business which is relevant for the purposes of calculating all sums due to them pursuant to this Schedule 3;
- 2. Within 60 Business Days following the end of each Calculation Year, the Purchaser shall issue to each of DLI, DMJ and SJ (as applicable) a written notice specifying the Geneva EBITDA or the Willen EBITDA and (together with supporting calculations) the amount of the relevant earn-out entitlement(s) for that Calculation Year. The Purchaser and each of DLI and DMJ (for himself and on behalf of SJ) shall seek to agree the earn-out entitlement.
- 3. In the event of a dispute between the Parties as to the calculation in respect of any Calculation Year of the Geneva EBITDA or the Willen EBITDA, such matter in dispute shall be referred for final settlement to a firm of chartered accountants to be agreed between the Sellers' Representative and the Purchaser or in default of such agreement within a period of 14 days to be selected by the President for the time being of the Institute of Chartered Accountants in England and Wales (or any person for the time being performing the functions of that official) on the application of the Purchaser or the Sellers' Representative. In making their determination, the accountants shall act as experts and not as arbitrators, their decision shall (in the absence of manifest error) be final and binding on the Parties and their fees shall be borne and paid by the Party nominated by the accountant, taking account of the merits of each of the respective Party's case.

48

EXHIBIT TO SCHEDULE 3

49

Schedule 4

CONDUCT OF THE BUSINESS PRE-CLOSING

1. As soon as practicable following the date of this Agreement, and in any event before Closing, DMJ shall use all reasonable endeavours to put in place an insurance policy in accordance with the terms of the Lease, based on the reinstatement value of the Property contained in the valuation carried out in 2007 for the purposes of the refinancing with Barclays Bank plc (if found), or the recommendations of an external valuer following a

formal valuation of the Property for insurance purposes. If a formal valuation is carried out for insurance purposes, the cost of such valuation shall be borne equally by the Purchaser and DMJ. A report on progress is to be provided every Tuesday and Thursday and a copy of the insurance policy shall be provided to the Purchaser before Closing. If DMJ is unable to obtain insurance in accordance with the terms of the Lease, the Lease shall be amended to incorporate the terms of the insurance policy that has been obtained.

- 2. From the date of this Agreement until Closing, the Sellers shall each ensure (in respect of the Target Company or Target Companies in which they hold any of the Shares) that (except with the Purchaser's prior written consent):
 - (a) the affairs of the Target Companies are conducted only in the ordinary and usual course and that no payment is made by a Target Company other than routine payments in the ordinary and usual course of business;
 - (b) all reasonable steps are taken to preserve and protect the assets of each of the Target Companies and to preserve and retain the goodwill of their global Business (including the existing relationships with customers and suppliers);
 - (c) subject to clause 22 (Confidentiality), the Purchaser's representatives shall be allowed such access as is reasonably requested, upon reasonable notice and during Working Hours, to (i) the books and records of each Target Company which relate to the Business (including all statutory books, minute books, leases, contracts, supplier lists and customer lists), with the right to take copies and (ii) the premises used by, and management of, the Business;
 - (d) no Target Company does, procures or knowingly allows or any act or omission which would constitute or give rise to a breach of any Warranty if the Warranties were repeated at any time before Closing by reference to the facts and circumstances then existing as if references in the Warranties to the date of this Agreement were references to the relevant date;
 - (e) all relevant information which comes to its notice or that of any other Seller in relation to any fact or matter (whether existing on or before the date of this Agreement or arising afterwards) which may constitute a breach of any Warranty if the Warranties were to be repeated on or at any time before Closing by reference to the facts and circumstances then existing as if references in the Warranties to the date of this Agreement were references to the relevant date, is promptly Disclosed to the Purchaser;

50

- (f) no Target Company declares, authorises, makes or pays any dividend or other distribution (whether in Cash, stock or in kind) or reduces, purchases or redeems an part of its paid-up share capital;
- (g) no Target Company (i) creates, allots or issues or agrees to create, allot, or issue any share or loan capital or other security or (ii) grants any option over or right to subscribe for any share or loan capital or other security;
- (h) no Target Company sells or purchases or disposes of any interest in any share or loan capital or other security of any Target Company;
- (i) except to replace employees of the Target Companies on substantially the same terms, no Target Company shall employ or agree to employ any new persons fully or part time in the Business or dismiss any existing employees, (except for incompetence or gross misconduct or other reasonable cause justifiable in law);
- (j) no Key Executive is given notice of termination of employment or is dismissed;
- (k) no action is taken by any member of any Target Company, which is inconsistent with the provisions of this Agreement or implementation of the Proposed Transactions;
- (l) no Target Company shall make, propose or communicate any change to the terms and conditions, remuneration or benefits of any employee;
- (m) no Target Company shall make or propose to make any loan to any person whether a Seller, a person connected to a Seller, another Target Company, or otherwise to any third party;
- 3. Pending Closing, the Sellers shall ensure that no Target Company agrees to or permits (except with the Purchaser's prior written consent):
 - (a) the reorganisation of any part of its business, or the discontinuance of any part of it;
 - (b) any failure to settle, in accordance with the payment procedures and timescales normally observed by it, any debts incurred by the Business in the normal course of trading;
 - (c) any entry into or termination of any contract or arrangement (i) having a value or involving or likely to involve expenditure in excess of £25,000 per annum or (ii) which cannot be performed within its terms within 12 months after the date on which it is entered into or undertaken or cannot be terminated on less than 12 months' notice or (iii) which may result in any material change in the nature or scope of the operations of that Target Company's business (or any modification of an existing contract or arrangement which would itself fall, or cause the contract or arrangement concerned to fall, within any of (i) to (iii)) or the making of any bid, tender, proposal or offer likely to lead to any such contract or arrangement;

51

(d) the giving of any guarantee, indemnity or other agreement to secure an obligation of a third party which if called would result in a cost to the Target Company of £25,000 or more;

- (e) the institution or settlement of any litigation where it could result in a payment to or by a Target Company of £10,000 or more except for collection in the ordinary course of trading debts, none of which exceeds £10,000;
- (f) the entry into or material modification of any agreement with any trade union or other body representing its Employees or relating to any works council;
- (g) the creation of any Third Party Right over the Shares;
- (h) the acquisition or disposal of any material asset or material stocks, in each case, involving consideration, expenditure or liabilities in excess of £25,000 (exclusive of VAT), outside the ordinary course of business;

52

Schedule 5

CLOSING ARRANGEMENTS

Part A: Sellers' Obligations

- 1. At or before Closing, the Sellers shall deliver or ensure that there is delivered to the Purchaser (or made available to the Purchaser's satisfaction) in relation to their respective Set of Shares:
 - (a) all necessary documents, duly executed or endorsed where so required, to enable title in all the Shares to pass fully and effectively into the name of the Purchaser, including any documents, such as necessary waivers of pre-emption rights or other consents, as may be required to enable the Purchaser to be registered as the holder of the Shares;
 - (b) the share certificates (including in relation to any bearer shares) in respect of all of the Shares in respect of which certificates were issued or are required by law to be issued (or if such certificates have been lost or destroyed indemnities given by the relevant Seller in favour of the Purchaser);
 - (c) in respect of each Target Company, the certificate of incorporation, common seal (if it exists), share register or ledger and share certificate book (with any unissued share certificates) and all minute books and other statutory books (which shall be written up to but not including the Closing Date) as are kept by the relevant Target Company;
 - (d) letters of resignation in the Agreed Form duly executed by such directors as the Purchaser may notify to the Sellers or the Sellers' Representative prior to the Closing Date in respect of their directorships of the Target Companies;
 - (e) letters of resignation in the Agreed Form duly executed by such company secretaries as the Purchaser may notify to the Sellers or the Sellers' Representative prior to the Closing Date in respect of their position as company secretary of the Target Companies.
 - (f) letters of resignation in the Agreed Form duly executed by such auditors as the Purchaser may notify to the Sellers or the Sellers' Representative prior to Closing in respect of their position as auditors of the Target Companies, together with such further documents, if any, as are needed to effect a valid and legally binding resignation and satisfy any applicable legal on regulatory requirements in relation to that resignation;
 - (g) a certificate of each of the Sellers to the effect that each of the Warranties, insofar as they relate to such Seller's Set of Shares, was true and accurate in all material respects at the date of this Agreement; and
 - (h) a certificate of each of the Sellers to the effect that each of the Closing Warranties, insofar as they relate to such Seller's Set of Shares, is true and accurate in all material respects immediately before Closing.

53

- 2. The Sellers shall ensure that resolutions of the boards and/or supervisory board (as necessary to provide valid authorisation) of directors of each relevant Target Company (or if required by the law of its jurisdiction or its articles of association, by-laws or other constitutional documents, of its shareholders) are passed by which the following business is transacted:
 - (a) if appropriate, the registration is approved (subject to such legal or other requirements as are necessary for such registrations to be effected being satisfied) of the transfers in respect of the Shares are approved;
 - (b) if appropriate, the resignations referred to in each of paragraphs (d), (e) and (f) of this Part A above are accepted and such persons as the Purchaser may respectively notify to the Seller prior to Closing are appointed as directors, company secretaries and auditors of the Target Companies;
 - (c) the accounting reference date of each Target Company is changed to 30 September;
 - (d) the registered office of each Target Company is changed to such address as the Purchaser may respectively notify to the Seller prior to Closing; and
 - (e) each Target Company makes such amendments to its constitutional documents as the Purchaser may reasonably require and waives any rights of pre-emption in respect of the Shares set out in its respective constitutional documents or otherwise.

- 3. Choicebrand Supplies Limited shall have entered into an agreement with Geneva Industries Limited to acquire certain inventory for £40,000 plus VAT to be paid in equal monthly instalments of £5,000.
- 4. DMD shall procure, within 30 days of Closing (i) that the business of C.R. Brighton Limited shall be terminated (ii) a schedule of assets (to include a list of all customers and all registered web addresses and domain names of C.R. Brighton Limited) is prepared and delivered to the Purchaser and (iii) such assets shall promptly be transferred to any Target Company as specified by the Purchaser. For the avoidance of doubt, DMD shall not be liable for any inaccuracies in the information provided in pursuant to (ii).
- 5. The Sellers shall, or shall procure that, any and all Financial Debt will be repaid at Closing.
- 6. The Key Executives shall each (i) execute an objection to the transfer of their employment from the relevant Target Company to the Purchaser under Regulation 4(8) of the Regulations (in the Agreed Form) (or at the option of the Purchaser, a letter resigning from their employment with the relevant Target Company by mutual consent, and without liability (in the Agreed Form)) and (ii) execute a service agreement with the Purchaser (in the Agreed Form as set out in Exhibit 5).
- 7. DMJ and SJ shall deliver, or shall procure there is delivered, audited financial statements for the year ending 31 December 2007 in respect of Geneva Industries Limited, to the Purchaser at Closing.

54

8. DMJ and DLI shall deliver, or shall procure there is delivered, audited financial statements for the year ending 31 December 2007 in respect of Willen Trading Limited, to the Purchaser at Closing.

Part B: Purchaser Obligations

- At Closing, the Purchaser shall:
 - (a) deliver or ensure that there is delivered to the Sellers (or made available to the Seller's satisfaction) a copy of a resolution (certified by a duly appointed officer as true and correct) of the board and/or supervisory board (as necessary to provide valid authorisation) of directors of the Purchaser (or, if required by the law of its jurisdiction or its articles of association, by-laws or equivalent constitutional documents, of its shareholders) authorising the execution of and the performance by the relevant company of its obligations under this Agreement and each of the Transaction Documents to be executed by it; and
 - (b) instruct the Purchaser's Solicitors to instruct their bank to pay to the Sellers' Solicitors and the Escrow Agent the Initial Share Prices in accordance with clause 4.3.
 - (c) deliver a duly executed Secretary's Certificate of the Purchaser Parent certifying as to the adoption by the Purchaser Parent's board of resolutions authorizing the Purchaser Parent to enter into this Agreement.
 - (d) deliver the Loan Notes to each of DLI, DMJ and SJ pursuant to, and in the amounts stated in, Clauses 5, 6 and 7 of this Agreement.

Part C: General

- 1. The Sellers and the Purchaser shall negotiate in good faith with a view to agreeing before the Closing Date the final form of any Transaction Document which is not in Agreed Form at the date of this Agreement. If not so agreed by the Closing Date, the Transaction Document shall be in the form reasonably specified by the Purchaser provided it is consistent with the terms of this Agreement.
- 2. At or before Closing, the Seller and the Purchaser shall execute and deliver to each other (or if applicable, procure that their relevant Affiliates shall execute and deliver) the following other documents in the Agreed Form required by this Agreement to be executed on or before Closing, namely:
 - (a) the Lease with the schedule of condition provided by the Purchaser (the cost of which shall be borne equally by the Purchaser and DMJ) annexed thereto;
 - (b) a business services agreement between Geneva Industries Limited and CAT Europe Limited in the Agreed Form;
 - (c) all documents and items delivered at Closing shall be held by the recipient to the order of the person delivering the same until such time as Closing shall take place.

55

Schedule 6

SELLER WARRANTIES

Part A: General

- 1. The Seller and the Shares
- 1.1 Authorisations, valid obligations, filings and consents
 - (a) Each Target Company has obtained all corporate authorisations and all other governmental, statutory, regulatory or other consents, licences, authorisations, waivers or exemptions required to empower it to enter into and perform its obligations under this Agreement and any

Transaction Document to which it is a Party.

- (b) Neither entry into this Agreement nor implementation of the Proposed Transactions will (i) result in violation or breach of any laws or regulations in any relevant jurisdiction or (ii) amount to a violation or default with respect to any statute, regulation, order, decree or judgment of any court or any governmental or regulatory authority in any jurisdiction, by any Seller or any Target Company, where, in each case, the breach, conflict or violation would adversely affect to a material extent its ability to enter into or perform its obligations under this Agreement and/or any Transaction Document to which it is a Party.
- (c) Except as referred to in this Agreement, no Seller or Target Company (i) is required to make any announcement, consultation, notice, report or filing or (ii) requires any consent, approval, registration, authorisation or permit, in each case in connection with the execution and performance of this Agreement or any other Transaction Document.
- 1.2 The Sellers, the Shares and the Target Companies
 - (a) Each Target Company is validly incorporated, in existence and duly registered under the laws of its jurisdiction of incorporation and each Target Company has full power to conduct its business as conducted at the date of this Agreement.
 - (b) The Shares constitute the whole of the issued and allotted or, to the extent appropriate, registered, share capitals of the Target Companies listed in Column 2 of Schedule 1. All the Shares are fully paid or properly credited as fully paid and there is no liability to pay any additional contributions on the Shares.
 - (c) No person has the right (exercisable now or in the future and whether contingent or not) to call for the allotment or issue of any share or loan capital in any Target Company.

56

- (d) The information in respect of each of the Target Companies set out in Exhibit 1 is true, accurate and not misleading.
- 1.3 Other interests

No Target Company owns or has any interest of any nature in any shares, debentures or other securities issued by any undertaking.

1.4 CAT Europe Limited

Geneva Industries Limited has completed the divestment of all of its shares in CAT Europe Limited for £50,000.

1.5 Trade Restrictions

The Target Companies warrants that it has not and, between signing and Closing shall not:

- (a) engage in any business with or make any investments in, or make any payments to, (i) any government, entity or individual with which U.S. persons are prohibited from engaging in activities or doing business by U.S. laws or regulations, including the economic sanctions administered by the United States Office of Foreign Assets Control or (ii) any government, entity or individual targeted by any of the economic sanctions of the United Kingdom administered by the Bank of England; or
- (b) engage in or fund, directly or indirectly, any business activities with, or for the benefit of, a government, national, resident or legal entity of Cuba, Sudan, Iran, Myanmar, the Democratic Republic of Korea or Syria.

2. Financial Matters

2.1 The Accounts

The Last Accounts of each Target Company give a true and fair view of the state of affairs of that Target Company, and its assets and liabilities as at the Last Accounts Date and of the results thereof for the financial year ended on the Last Accounts Date and:

- (a) the Accounts for each of the last three (3) financial years ended on the Last Accounts Date were prepared in accordance with the requirements of all relevant laws and the relevant generally accepted accounting principles then in force;
- (b) the Accounts for each of the last three (3) financial years ended on the Last Accounts Date either make full provision for or disclose in accordance with the relevant generally accepted accounting principles all liabilities (whether actual, contingent or disputed and including finance lease commitments and pension liabilities), all outstanding capital commitments and all bad or doubtful debts of the Target Company or Target Companies to which they relate;

57

(c) the results shown by the Accounts for each of the last three (3) financial years ended on the Last Accounts Date were not affected by any extraordinary or exceptional item or by any other factor rendering such results for all or any of those periods unusually high or low.

2.2 Management Accounts

The Management Accounts of each Target Company for all periods ended after the Last Accounts Date have been properly prepared in all material respects. On the basis of the accounting bases, practices and policies used in their preparation and having regard to the purpose for which they were

prepared, the Management Accounts:

- (a) are not misleading in any material respect;
- (b) do not materially over-state the value of the assets nor materially under-state the liabilities of the relevant Target Company as at the dates to which they were drawn up; and
- (c) do not materially over-state the profits or materially under-state the losses of the relevant Target Company in respect of the periods to which they relate.

In the context of this paragraph 2.2, "material" and "materially" shall be deemed to refer to facts, matters, circumstances, issues or events which have or the absence of which would have an aggregate cost, benefit or value to the Target Companies of more than £50,000.

2.3 Position since Last Accounts Date

Since the Last Accounts Date:

- (a) there has been no Material Adverse Change;
- (b) each Target Company has carried on its Business in the ordinary and usual course, and no Target Company has made or agreed to make any payment other than routine payments in the ordinary and usual course of trading;
- (c) no Target Company has declared, authorised, paid or made, any dividend or other distribution (whether in Cash, stock or in kind) nor has it reduced paid-up share capital (except for any dividends provided for in the Accounts);
- (d) no Target Company has issued or agreed to issue any share or loan capital or other similar interest;
- (e) no Target Company has entered into any contract, liability or commitment (whether in respect of capital expenditure or otherwise) which:
 - (i) cannot be performed within its terms within six (6) months after the date on which it was entered into or cannot be terminated on less than 30 days' notice; or
 - (ii) involved or may involve expenditure of more than £50,000 or an obligation of a material nature or magnitude;

58

- (f) no Target Company has acquired or disposed of, or agreed to acquire or dispose of, any one or more assets (other than stock) relating to its Business in a single transaction or series of connected transactions, where the value of such assets exceeds £10,000;
- (g) no Target Company has made any changes in terms of employment, including pension fund commitments, which taken together could increase the total staff costs of its Business by more than £10,000 per annum or the remuneration of any one director or employee by more than £10,000 per annum;
- (h) no Target Company has repaid any borrowing or indebtedness in advance of its stated maturity;
- (i) there has been no material increase or decrease in the levels of debtors or creditors or in the average collection or payment periods for the debtors and creditors respectively of any Target Company;
- (j) no debtor of any Target Company has been released on terms that it pays less than the book value of its debt and no debt in excess of £10,000 owing to any Target Company has been deferred, subordinated or written off or has proved to any extent irrecoverable.

2.4 No undisclosed liabilities

There are no actual or contingent liabilities of any of the Target Companies except for (i) liabilities disclosed or provided for in the Last Accounts (ii) liabilities incurred in the ordinary and usual course of business since the Last Accounts Date, none of which results in a Material Adverse Change or (iii) liabilities disclosed elsewhere in the Agreement.

2.5 Past transactions in accordance with applicable laws

Each Target Company has carried out all transactions in material compliance with all applicable laws and regulations. No such transaction constituted a transfer at an undervalue or an unlawful distribution or unlawful financial assistance by or to any Target Company. At no time during the period have the net assets (being the aggregate value of all the assets less the aggregate value of all the liabilities of the relevant company at the relevant time) of any Target Company been less than the aggregate amount of its share capital and undistributable reserves.

2.6 Accounting and other records

The statutory books, books of account and other records of each Target Company required to be kept by applicable laws in any relevant jurisdiction are up-to-date and have been maintained in accordance with those laws and relevant generally accepted accounting practices on a proper and consistent basis and comprise in all material respects complete and accurate records of all information required to be recorded. All such statutory books, books of account and other records are in the possession or under the control of a Target Company together with all documents of title and executed copies of all existing agreements which are necessary for the proper conduct of the Business and to which the relevant Target Company is a Party.

3. Debt position

3.1 Debts owed to the Target Companies

No Target Company has lent any money which is due to be repaid and as at the date of this Agreement has not been repaid to it and no Target Company owns the benefit of any debt (whether trading or otherwise), in each case other than other trade debts incurred in the ordinary and usual course of business which do not exceed £500,000 in aggregate for all Target Companies (and none of which exceeds £75,000).

3.2 Book debts

So far as the Sellers are aware (without having made enquiry of any person) the list of trade debtors Disclosed in the Disclosure Letter and/or Disclosure Bundle is complete in all material respects.

3.3 Debts owed by the Target Companies

- (a) No Target Company owes any Financial Debt whether to any of the Sellers or any person connected to the Sellers (other than any intercompany debt owing between the Target Companies), except as Disclosed in the Disclosure Letter and / or Disclosure Bundle.
- (b) Trade debts incurred by each Target Company in the ordinary course of its Business since the Last Accounts Date and outstanding at the date of this Agreement do not exceed £850,000 in aggregate for all Target Companies (and none of them individually exceeds £75,000).

4. Regulatory matters

4.1 Licences

Each Target Company has obtained all licences, permissions, authorisations (public or private) and consents (together, "Approvals") required for carrying on its Business effectively in the places and in the manner in which it is carried on at the date of this Agreement and in accordance with all applicable laws and regulations in each case in all material respects. Copies of all Approvals are contained in the Disclosure Bundle in Exhibit 4. These Approvals are in full force and effect, are not limited in duration or subject to any materially unusual or onerous conditions, have been complied with in all material respects and are freely transferable to the Purchaser. So far as the Sellers are aware, there are no circumstances which indicate that any Approval will or is likely to be revoked or not renewed, in whole or in part, in the ordinary course of events (whether as a result of the Proposed Transactions or any of the Transaction Documents or otherwise).

4.2 Compliance with laws

Each Target Company has conducted its Business and corporate affairs in accordance with its memorandum and articles of association, by-laws or other equivalent constitutional documents and in all material respects in accordance with all applicable laws and regulations. No Target Company is in default of any statute, regulation,

60

order, decree or judgment of any court or any governmental or regulatory authority in any jurisdiction.

4.3 Competition and fair trading laws.

- (a) No Target Company is engaged in any agreement, arrangement, practices or conduct which would amount to an infringement of the prohibitions in the Competition Act 1998, under the Enterprise Act 2002, of Article 81 EC Treaty, Article 82 EC Treaty or of EC Regulation 4064/89 and no director of any Target Company is engaged in any activity which would be an offence under the Enterprise Act 2002.
- (b) No Target Company is to the Sellers' knowledge the subject of any investigation, inquiry or proceedings by the Office of Fair Trading, the Competition Commission, inquiry or proceedings, or the European Commission in connection with any actual or alleged infringement of the prohibitions of the Competition Act 1998, Article 81 EC Treaty, Article 82 EC Treaty, EC Regulation 4064/89 or under the provisions of the Fair Trading Act 1973 or the Enterprise Act 2002.
- (c) No such investigation, inquiry or proceedings as mentioned in paragraph (b) have been threatened or to the Sellers' knowledge are pending and the Sellers have no knowledge of any circumstances likely to give rise to any such investigation, inquiry or proceedings.
- (d) No Target Company is affected by any existing or pending decisions, judgements, orders or rulings of the Office of Fair Trading, the Competition Commission, or the European Commission and have not given any undertakings or commitments to such bodies which affect the conduct of the business of any Target Company.
- (e) No Target Company is in receipt of any payment, guarantee, financial assistance or other aid form the government or any state body which was not, but should have been, notified to the European Commission under Article 88 EC Treaty for decision declaring such aid to be compatible with the common market.

4.4 State aid

No Target Company has received, or is proposing to receive any aid (within the meaning of Articles 87 to 89 of the EC Treaty).

5. Contractual matters

5.1 Material contracts

No Target Company is a party to any agreement or arrangement:

(a) under which, by virtue of the Proposed Transactions, (i) any other party is likely to be relieved of any obligation or become entitled to exercise any right (including any termination right or any pre-emption right or other option) or (ii) any Target Company is likely to be in default or lose any benefit, right or

61

licence which it currently enjoys or (iii) a liability or obligation of a Target Company is likely to be created or increased;

- (b) which is not in the ordinary course of business or (except trading between Target Companies) not on arm's length terms;
- (c) which requires, or confers any right to require, the allotment or issue of any shares, debentures or other securities of any Target Company now or at any future time;
- (d) which establishes any joint venture, consortium, partnership or profit (or loss) sharing agreement or arrangement;
- (e) under which any Target Company has sold or disposed of any company or business where it remains subject to any liability (whether contingent or otherwise);
- (f) which involves or is likely to involve expenditure by any Target Company totalling in excess of £5,000 per annum or any obligation of a material nature or magnitude;
- (g) which cannot be performed within its terms within 12 months after the date on which it was entered into or cannot be terminated on less than 12 months' notice;
- (h) which establishes any agency, distributorship, marketing, purchasing, manufacturing or licensing agreement or arrangement;
- (i) which is a recognition, procedural or other agreement between any Target Company and any recognised independent trade union;
- (j) which is a bid, tender, proposal or offer which, if accepted, would result in any Target Company being committed to any agreement or arrangement of a kind described in paragraphs 5.1(a) to 5.1(i) above.

5.2 Defaults

No Target Company is in material default under any agreement or arrangement to which it is a party and, so far as the Seller is aware, there are no circumstances likely to give rise to such a default. So far as the Sellers are aware, no party with whom any Target Company has entered into any agreement or arrangement is in material default under it and there are no circumstances likely to give rise to such a default. For this purpose "material" refers to any default which would have a cost to any Target Company (including a loss of profit) of £10,000 or more.

5.3 Trading relationships

During the 12 months preceding the date of this Agreement none of the 10 largest Customers (by revenue) of any Target Company has ceased to deal with that Target Company or has indicated in writing an intention to do so, either in whole or in part. So far as the Sellers are aware, no such person is likely to cease to deal with any

62

Target Company or to deal with it on a smaller scale (whether as a result of the Proposed Transactions or otherwise).

5.4 Principal suppliers and Customers

No Customer (including any person connected in any way with any Customer) accounts either for more than 10 per cent. of the aggregate value of all purchases or for more than 10 per cent. of the aggregate value of all sales of any Target Company.

5.5 Grants

No Target Company has ever received, or is proposing to receive, any grant for use wholly or partly in its Business. No Target Company has done, agreed to do or omitted to do anything which could result in (i) any such grant paid to a Target Company being refunded in whole or in part or (ii) any such grant for which application has been made not being paid or being reduced (whether as a result of the Proposed Transactions or otherwise).

6. Litigation and investigations

6.1 Litigation

Except as claimant in the collection of debts arising in the ordinary course of business (none of which individually exceeds £5,000 and which do not exceed £50,000 in aggregate), no Target Company is a claimant or defendant in or otherwise a party to any litigation, arbitration or administrative proceedings (including any proceedings before any tribunal), which are in progress, threatened or, so far as the Sellers are aware, pending by or

against or concerning it or any of its assets. The Sellers, as appropriate, are not aware of any circumstances which are likely to give rise to any such proceeding.

6.2 Investigations

No governmental, administrative, regulatory or other official investigation or inquiry concerning any Target Company is in progress or, so far as the Sellers are aware, pending, and so far as the Sellers are aware, there are no circumstances likely to lead to any such investigation or inquiry.

7. Insolvency etc.

7.1 Winding up

No order has been made, petition presented or meeting convened for the winding up of any Target Company, or for the appointment of any provisional liquidator or in relation to any other process whereby the business is terminated and the assets of the company concerned are distributed amongst the creditors and/or shareholders or other contributors, and there are no cases or proceedings under any applicable insolvency, reorganisation or similar laws in any relevant jurisdiction, and so far as the Sellers are aware, no events have occurred which, under applicable laws, would be reasonably likely to justify any such cases or proceedings.

63

7.2 Administration and receivership

So far as the Sellers are aware, no person has taken any step, legal proceeding or other procedure with a view to the appointment of an administrator, whether out of court or otherwise, in relation to any Target Company, and no receiver (including any administrative receiver) has been appointed in respect of the whole or any part of any of the property, assets and/or undertaking of any Target Company nor has any such order been made (including, in any relevant jurisdiction, any other order by which, during the period it is in force, the affairs, business and assets of the company concerned are managed by a person appointed for the purpose by a court, governmental agency or similar body).

7.3 Voluntary arrangement etc.

No Target Company has taken any steps with a view to a suspension of payments or a moratorium of any indebtedness or has made any voluntary arrangement with any of their creditors or are insolvent or unable to pay their debts as they fall due.

7.4 Registration of charges.

All material charges in favour of each Target Company required to be registered have been so registered to comply with all necessary formalities as to registration or otherwise in any applicable jurisdiction.

Part B: IP/IT

1. Business IP

- 1.1 The information contained in Schedule 11 is complete and accurate. The owner or applicant for each Intellectual Property Right specified in Schedule 11 is correctly stated. The Owned IP is valid and subsisting.
- 1.2 Each Target Company owns all of the rights and interests in and has title to, or has validly licensed to it, all of the Business IP required to carry on its Business. The Business IP of each Target Company comprises all the Intellectual Property Rights required to carry on the Business as it is currently carried on.
- 1.3 The Owned IP is not subject to amendment, challenge, removal or surrender. There is nothing that might prevent applications from being granted. All fees relating to the Owned IP have been paid.
- 1.4 No compulsory licences, licences of right or anything similar have been, or are likely to be, granted for the Owned IP.

2. Licences

2.1 The licences of Intellectual Property Rights granted to, or by, any Target Company are binding and in force. None of the parties to them is in default and there are no grounds on which they might be terminated. No disputes have arisen or are foreseeable in connection with them.

64

3. No infringement by Target Companies

3.1 So far as the Sellers are aware, none of the operations of any Target Company infringes, is to the Sellers' knowledge likely to infringe or has in the last two (2) years infringed the Intellectual Property Rights of a third party. No third party has disputed the right of any Target Company to use the Business IP and there are no circumstances likely to give rise to a dispute.

4. No infringement by Third Parties and enforcement

4.1 So far as the Sellers are aware, no third party is infringing, or is likely to infringe, the Business IP. No Target Company has disputed the right of a third party to use the Intellectual Property Rights owned or used by the third party and each Seller is not aware of any circumstances likely to give rise to a dispute.

- 4.2 No Target Company has acquiesced in the unauthorised use by a third party of the Business IP.
- 4.3 There have been no acts or omissions that would prejudice the rights of a Target Company to enforce the Business IP. Transactions relating to the Business IP have been registered promptly, and within applicable time limits.

5. Confidential information

5.1 Confidential information of any Target Company or that has been used by any Target Company has been kept confidential and has not been disclosed by the Sellers or to the Sellers' knowledge by any other person to third parties except in the ordinary course of business and subject to written confidentiality obligations from the third party. So far as the Sellers are aware, these confidentiality obligations have not been breached.

6. Encumbrances

6.1 The Business IP is not subject to any security interest, option, mortgage, charge or lien.

7. Restrictions on use

7.1 There are no agreements or arrangements that restrict the disclosure, use or assignment by any Target Companies of its Business IP.

8. Loss of Intellectual Property Rights

8.1 No Business IP will be lost, or rendered liable to termination, by virtue of the acquisition of the Shares or the performance of this Agreement or any of the Transaction Documents.

9. Disclosure, assignment and inventions

9.1 Each employee and independent contractor of a Target Company who, either alone or with others, has created, developed or invented Intellectual Property Rights which the Target Companies uses or might use has entered into a written agreement with a

65

Target Company which obliges him to disclose and assign the Intellectual Property Rights to a Target Company. No claims have been made or threatened by employees or ex-employees of a Target Company in any jurisdiction for inventor compensation or anything similar.

10. Documentation

10.1 Complete and accurate records, files and documents have been maintained for all Business IP and the records, files and documents are in the possession or under the control of a Target Company.

11. Moral rights

11.1 No moral rights have been asserted or are likely to be asserted which would affect the use of the Business IP.

12. Information Technology

- 12.1 The IT Systems are owned by, or properly licensed, leased or supplied under third party contracts to, a Target Company ("**Third Party Contracts**"). The relevant Target Company is not in default under the Third Party Contracts. So far as the Sellers are aware, there are no grounds on which the Third Party Contracts might be terminated and there are no disputes or material service delivery issues existing or likely. There is no reason to believe that the Third Party Contracts will not be renewed when they expire on the same or substantially similar terms.
- 12.2 There are no circumstances in which the ownership, benefit, or right to use the IT Systems may be lost by virtue of the acquisition of the Shares or the performance of this Agreement or any of the Transaction Documents.
- 12.3 The IT Systems have not failed to any material extent in the past two (2) years and the data that they process has, so far as the Sellers are aware, not been corrupted. The IT Systems do not contain viruses, bugs or things that distort their proper functioning, permit unauthorised access or disable them without the consent of the user.
- 12.4 No Target Company nor any Seller has authorised a third party to modify, reverse engineer or create derivative works of software or systems included in the IT Systems.
- 12.5 No Target Company nor any Seller has disclosed the source code of software included in the IT Systems, or licensed by any Target Company to third parties, to any third party other than under a binding agreement with a reputable escrow agent on the agent's standard terms. Nothing has happened which might, and the acquisition of the Shares will not, trigger release of the source code to a third party under an escrow agreement.
- 12.6 The Target Companies and the Sellers have, in accordance with best industry practice, taken precautions to preserve the availability, security and integrity of the IT Systems and the data and information stored on the IT Systems.
- 12.7 The IT Systems are adequate for the needs of the Business as carried out at the date of this Agreement.

13. Data Protection

- 13.1 Each Target Company complies with all applicable data protection laws, guidelines and industry standards.
- 13.2 No notice or allegation has been received by any Target Company alleging that any Target Company has not complied with the things listed in this paragraph 13.
- Personal data relating to customers of the Business has been collected on terms that allow the data to be lawfully transferred to and processed by the Purchaser.

Part C: Real Estate

1. General

- 1.1 The Property comprises all the land and buildings owned, controlled, occupied or used by any Target Companies or in relation to which any Target Company has any right, interest or liability.
- 1.2 The information in respect of the Property set out in Schedule 10 is true, complete and accurate and not misleading in any respect.
- 1.3 The replies to enquiries provided in respect of the Properties are true, complete and accurate and not misleading in any respect.

2. Title

- 2.1 No person has or claims any security interest, lien, option, right of pre emption or other similar interest (including any arising by statute) in or over the Property or any relevant deeds or documents.
- 2.2 The Property is not affected by a subsisting contract for sale or other disposition of any interest in it.

3. Adverse Interests

3.1 So far as the Sellers are aware the Property or other land which benefits the Property is not subject to any order, resolution or proposal for compulsory acquisition or located in an area which is or is proposed to be subject to any statutory or other order.

4. Planning Matters

- 4.1 The Property and all uses of, and developments on, the Property comply with all Planning Legislation.
- 4.2 No Permission in respect of the Property is for a limited period or personal, and no Permission or Statutory Agreement affecting the Property contains (expressly or impliedly) any unusual or onerous conditions or obligations.

67

- 4.3 So far as the Sellers are aware all Planning Legislation and all Permissions or Statutory Agreements affecting the Property have been complied with to date and there is no reason why the same should not continue to be complied with and no Target Company is obliged to enter into any Statutory Agreement.
- 4.4 All Permissions relating to the Property not yet implemented in whole are capable of implementation by the relevant Target Company without the need for any further applications, registrations, notifications, payments or Permissions.
- 4.5 No Target Company is subject to any outstanding monetary claim or liability, actual or contingent, arising under Planning Legislation or any Permission or Statutory Agreement.
- 4.6 The Property has not been the subject of any environmental statements during the last six (6) years.
- 4.7 In this paragraph 4, the following words and expressions have the following meanings:
 - "Planning Legislation" means all legislation intended to control or regulate the construction, demolition, alteration or use of land or buildings or to preserve or protect the national heritage and any orders, bye laws or regulations made or granted under any of them;
 - "Permission" means any planning permission, approval or other equivalent consent, authorisation or licence given or deemed to be given pursuant to planning legislation and includes all conditions attached to it; and

"Statutory Agreement" means any agreement or planning obligation under Section 18 of the Public Health Act 1936, Section 52 of the Town and Country Planning Act 1971, Section 38 and Section 278 of the Highways Act 1980, Section 33 of the Local Government (Miscellaneous Provisions) Act 1982, Section 106 of the Town and Country Planning Act 1990, Section 104 of the Water Industry Act 1991 or any provision in legislation of a similar nature.

5. **Property Liabilities**

5.1 Each Target Company has no actual or contingent obligation or liabilities in relation to any freehold or leasehold property other than (where applicable) under its existing title to the Property.

6. State of Property

- 6.1 The buildings and other structures on, under or over the Property are in good and substantial repair and condition and, so far as the Sellers are aware, fit for the purposes for which they are presently used and there is no material defect (whether latent, inherent or otherwise) in the construction or condition of any of such buildings or other structures.
- 6.2 There are no development works, redevelopment works or fitting-out works outstanding in respect of the Property.

68

- 6.3 The Property has not suffered from any of the following:
 - (a) flooding (since the date on which DMJ acquired the Property, and otherwise so far as the Sellers are aware);
 - (b) subsidence (so far as the Sellers are aware);
 - (c) heave (so far as the Sellers are aware);
 - (d) landslip (so far as the Sellers are aware);
 - (e) mining activities (so far as the Sellers are aware);
 - (f) structural defects (so far as the Sellers are aware);
 - (g) defects in the drains and services from time to time serving the Property (so far as the Sellers are aware); or
 - (h) dry rot, wet rot, rising damp or any infestation (so far as the Sellers are aware).
- 6.4 No Target Company has received any adverse report from any engineer, surveyor or other professional relating to the Property and is not aware of any predecessor in title having done so.

7. Due Diligence and Information

7.1 Each Target Company has provided true and complete copies of all deeds, documents and other information in its possession, or of which it is aware, that is relevant to each Target Company's interest in or use of the Property or their value.

Part D: Employment

1. Employment Warranties

1.1 Remuneration and Benefits

The particulars of all Employees in the Disclosure Letter and Disclosure Bundle show all remuneration and other benefits:

- (a) actually provided; or
- (b) which the Target Companies are bound to provide (whether now or in the future),

to each Employee and are true and complete and include particulars of and details of participation in all profit sharing, incentive, bonus, commission, share option, medical, permanent health insurance, directors' and officers' insurance, travel, car, redundancy and other benefit schemes, arrangements and understandings (the "Employee Benefit Plans") operated for all or any Employees or their dependants and any payments in connection with the termination of any office or employment, in each case whether legally binding on the Target Companies or not.

69

1.2 Employee List

The particulars of all Employees in the Disclosure Letter and Disclosure Bundle show the employing entity, names, age, job title, date of birth, basic salary, pension contribution, date of commencement of employment and details of all other remuneration and benefits. Since the Accounts Date none of the Target Companies has made, announced or proposed any changes to the remuneration or other benefits of any Employees and none of the Target Companies are under an obligation to make any such changes with or without retrospective operation.

1.3 Terms and Conditions

The Disclosure Letter and Disclosure Bundle contain copies of all the:

- (a) standard terms and conditions, staff handbooks and policies which apply to Employees and identifies which terms and conditions apply to which employees;
- (b) terms of engagement or any consultants, agents or workers engaged by any of the Target Companies;
- (c) each of the Key Executive's service agreements and any other documentation which identities their terms and conditions of employment (including updates and any documentation relating to bonus and/or profit sharing); and

(d) terms of employment which apply to the Employees and which are variations from the standard terms and conditions.

All Employees have received a written statement of particulars of their employment as required by section 1 of the Employment Rights Act 1996. The terms of employment or engagement of all Employees, consultants and workers are such that their employment or engagement may be terminated by not more than twelve weeks' notice given at any time without liability for any payment including by way of compensation or damages (except for unfair dismissal or a statutory redundancy payment). None of the Target Companies have entered into any agreement or arrangement for the management or operation of its business or any part thereof other than with the Employees.

1.4 Operation of the Employee Benefit Plans

- (a) The Employee Benefit Plans have at all times been operated in accordance with their governing rules or terms and all applicable laws and all documents which are required to be filed with any regulatory authority have been so filed and all Tax clearances and approvals necessary to obtain favourable Tax treatment for the relevant Target Company and/or the participants in the Employee Benefit Plans have been obtained and not withdrawn and no act or omission has occurred which has or could prejudice any such Tax clearance and/or approval.
- (b) No Employee or any dependant has made any claim against any of the Target Companies in respect of any of the Employee Benefit Plans.

70

(c) Each of the Employee Benefit Plans are fully insured and the premiums are paid up to date. Full and up to date details of the terms and conditions applicable to the Employee Benefit Plans have been provided to the relevant Employees.

1.5 Claims by Employees

None of the current or former employees, consultants or workers of any of the Target Companies have now, or at any time in the last two (2) years, had any claim or right or action against any of the Target Companies (or any person indemnified by any of the Target Companies), including (but not limited to) any claim:

- (a) in respect of any accident or injury; or
- (b) for breach of contract of employment or services; or
- (c) for loss of office or arising out of or connected with the termination of his office, employment (including any redundancy payment) or engagement,

and there is no event which would or might give rise to any such claim. Any claim Disclosed in respect of paragraph 1.5(c) above is fully covered by insurance.

1.6 Industrial Relations and Training

- (a) The Disclosure Letter and Disclosure Bundle contain details (an on aggregated or departmental basis) of those Employees who are members of a trade union, staff association or any other body representing workers and of those Employees who are covered by collective bargaining arrangements.
- (b) The Disclosure Letter and Disclosure Bundle contain copies of and full details of all rights and liabilities relating to any Employees pursuant to any collective agreements (whether with a trade union, staff association or any other body representing workers and whether legally binding or not) concerning the Business.
- (c) The Disclosure Letter and Disclosure Bundle contain a summary of any negotiations (whether formal or informal) that are currently ongoing with any trade union.
- (d) Within the three (3) years preceding the date hereof none of the Target Companies have been engaged or involved in any trade dispute (as defined in section 218 of Trade Union Labour Relation (Consolidation) Act 1992) with any Employee, trade union, staff association or any other body representing any employees and no event has occurred which could or might give rise to any such dispute and no industrial action involving Employees, official or unofficial, is now occurring or threatened nor has any industrial relations or employment matter been referred either by the Target Companies or any Employee or by any trade union representing any of the Employees to the Advisory Conciliation and Arbitration Service (ACAS) for advice, conciliation or arbitration.

71

(e) In relation to the Employees, there are no training schemes, arrangements or proposals, whether past or present, in respect of which a levy may henceforth become payable by any of the Target Companies under the Industrial Training Act 1982 (as amended).

1.7 Compliance with Laws, Enquiries and Discrimination

- (a) Each of the Target Companies have complied in all material respects with all relevant provisions of the European Treaties, EC Directives, statutes, regulations, codes of conduct, collective agreements, terms and conditions of employment, orders, declarations and awards relevant to their current, former and prospective employees, consultants and workers or the relations between each of the Target Companies and any recognised trade union, staff association or any other body representing any of the Employees.
- (b) There are no enquiries or investigations existing, pending or threatened affecting any of the Target Companies in relation to any of any past or present employee, worker or consultant including by any health and safety enforcement authority, the Equal Opportunities Commission,

the Pensions Regulator, any predecessor body or any other governmental or quasi- governmental body.

(c) There are no terms or conditions under which any Employee or worker is employed, nor has anything occurred prior to Closing, that may give rise to any claim for sex, race, age, disability or other unlawful discrimination or equal pay either under English, United Kingdom or European Law whether by such Employees or otherwise.

1.8 Records

Each of the Target Companies has maintained adequate, suitable and up to date records regarding the service of each of the Employees and workers and, in particular, has maintained all records required under the Working Time Regulations 1998. All such records comply with the requirements of the Data Protection Act 1998.

1.9 Payment up to the date of this Agreement

- (a) The salaries, fees and wages and other benefits of all of the Employees, workers and consultants (including accruals of bonuses, commissions or other benefits), all related payments to third party benefit providers and all related payments to the relevant authorities have been paid or discharged in full in respect of the period up to the date of this Agreement. No Employee or worker has more than 25 days of accrued but untaken holiday.
- (b) There are no amounts owing or agreed to be loaned or advanced by any of the Target Companies to any Employee, worker or consultant (other than amounts representing remuneration accrued due for the current pay period, accrued holiday pay for the current holiday year or reimbursement of expenses).

72

1.10 Employees in the Business

- (a) Other than the Employees there are no other employees, workers or persons involved in the operation of the Business.
- (b) There is no person previously employed or engaged by any of the Target Companies who now has or may have statutory or contractual right to return to work or to be reinstated or to be reengaged.
- (c) No Employee, worker or consultant is on secondment, maternity leave or has been absent on grounds of disability, long term sickness or other leave of absence having lasted more than 14 days.
- (d) There are no outstanding offers of employment or engagement to work in the Business and no person has accepted such an offer but not yet taken up the position accepted.
- (e) Particulars of any employee who has been dismissed or subject to disciplinary action, any grievance raised by any Employee, any protected disclosure or subject access request made by any Employee or worker within the last three (3) years are contained in the Disclosure Bundle.
- (f) No person has ever become an employee of any of the Target Companies by virtue of the Regulations and none of the Target Companies are party to any agreements or arrangements on the termination of which the Regulations may apply.
- (g) No Employee is currently under notice of termination of employment.

1.11 Redundancy

Full and accurate details are Disclosure Letter and Disclosure Bundle of any redundancy payment (whether pursuant to a redundancy scheme or formula or policy or otherwise and whether contractual or discretionary) any of the Target Companies has made in excess of the statutory redundancy entitlement to any former employee in the last three (3) years.

1.12 Health and Safety

Full details of all health and safety committees, health and safety representatives, and any complaints, recommendations, investigations or claims relating to health and safety issues made or carried out in the last five (5) years and affecting any of the Target Companies and any of the Employees have been Disclosure Letter and Disclosure Bundle.

1.13 Immigration

Each of the Target Companies has satisfied the conditions referred to in section 8(1) of the Asylum and Immigration Act 1996 or the prescribed requirements of the Immigration, Asylum & Nationality Act in respect of each of their Employees. The Disclosure Letter and Disclosure Bundle contain a list of each of the Employees who has limited leave to remain in the United Kingdom, together the date upon which their

73

current period of leave expires, their immigration category and (where relevant) their work permit number.

1.14 Effect of this Agreement

(a) There are no agreements or arrangements (whether contractual, discretionary or customary) whereby any Employee, worker or consultant may become entitled to any payments, increased notice periods or other changes to their terms and conditions of employment by reason of

the transactions contemplated by this Agreement.

(b) As far as the Sellers are aware, no Employee intends to give notice to terminate his or her employment following the execution of this Agreement.

Part E: Pensions

1. Pensions

- 1.1 No Target Company has or may have any obligation (whether or not legally binding) to provide or contribute towards pension, lump sum, death or ill-health benefits (together, "**Retirement Benefits**") in respect of its past or present officers and employees or their dependants and no proposal or announcement has been made to any Employee or officer of any of the Target Companies about the introduction, continuance, increase or improvement of, or the payment of a contribution towards, any other pension, lump sum, death or ill health benefit. None of the Target Companies have had an obligation to provide such benefits under any other arrangement in the past.
- 1.2 Each of the Target Companies has facilitated access for its employees to a designated stakeholder scheme to the extent required by Section 3 of the Welfare Reform and Pensions Act 1999.
- 1.3 No claims, complaints or investigations (including, without limitation, contact with or correspondence from the Pensions Regulator, the Pensions Advisory Service, the Pensions Ombudsman or any employee, consultant or worker) have been made, received or undertaken in relation to the provision or lack of provision of any pension, lump sum, death or ill-health benefits by any of the Target Companies and nor are any such claims, complaints or investigations pending or threatened.

Part F: Insurance

1. Insurance

- 1.1 The Disclosure Letter and Disclosure Bundle contain complete and accurate details of all insurance policies maintained by the Target Companies
- 1.2 The Target Companies maintain, and have at all material times maintained, adequate insurance in respect of all assets of an insurable nature against fire, accident, theft and damage in amounts representing their full replacement or reinstatement values, against third party loss (including by way of employer's liability and public liability

74

- insurance), loss of profits and all other risks required by applicable law or regulation to be covered by insurance or normally insured against by companies carrying on the same business.
- 1.3 All premiums due on the subsisting insurance policies of the Target Companies have been duly paid, all other conditions of those policies have been performed and observed, and there are no circumstances which might result in an increase in premium or make any policy void or voidable.
- 1.4 No subsisting insurance policy of the Target Companies is subject to any special or unusual terms or restrictions, and the Target Companies have not been refused insurance during the period of three (3) years ending on the date of this Agreement.
- 1.5 The subsisting insurance policies of the Target Companies will continue in full force and effect notwithstanding the sale of the Shares to the Purchaser under this Agreement.
- 1.6 The Disclosure Letter and Disclosure Bundle contain complete and accurate details of all insurance claims made by the Target Companies during the period of two (2) years ending on the date of this Agreement, and there are no circumstances which would or might entitle the Target Companies to make such a claim or which would or might be required under any of the policies to be notified to the insurers.

Part G: Tax Warranties

1. General

- All notices, returns (including any land transaction returns), reports, accounts, computations, statements, assessments and registrations and any other necessary information submitted by any of the Target Companies to any Taxation Authority for the purposes of Taxation have been made on a proper basis, were submitted within applicable time limits, were accurate, true and complete when supplied and (so far as the Sellers are aware) remain accurate, true and complete in all material respects. None of the above is, or (so far as the Sellers are aware) is likely to be, the subject of any material dispute with any Taxation Authority.
- 1.2 All Taxation (whether of the UK or elsewhere), for which any of the Target Companies have been liable or are liable to account for, has been duly paid.
- 1.3 Each of the Target Companies has, within applicable time limits, maintained all records in relation to Taxation as are required to maintain.
- 1.4 Each of the Target Companies has complied within applicable time limits with all notices served on them and any other requirements lawfully made of them by any Taxation Authority.
- 1.5 No Target Company has made any payments representing instalments of corporation tax pursuant to the Corporation Tax (Instalment Payments) Regulations 1998 in respect of any current or preceding accounting periods and is not under any obligation to do so.

- 1.6 No Target Company has paid, within the past three (3) years ending on the date of this Agreement, nor is any Target Company under any existing liability to pay, any penalty, fine, surcharge or interest charged by virtue of the TMA 1970 or any other Taxation Statute nor are there any existing circumstances of default which could lead any Taxation Authority to impose any such penalty, fine, surcharge or interest on a Target Company.
- 1.7 All Taxation and national insurance deductible and payable under the PAYE system has, so far as is required to be deducted, been deducted from all payments made (or treated as made) by any of the Target Companies to (including without limitation all Tax chargeable on benefits provided for) the directors, employees or former employees of each Target Company or any persons required to be treated as such and to the extent that the due date for payment thereof fell before the date of this Agreement, all such amounts have been accounted for to the relevant Taxation Authority. All applicable regulations in respect of PAYE and national insurance have been complied with.
- 1.8 Each Target Company has deducted or withheld all Taxation which it has been obliged by law to deduct or withhold from payments made by it and has properly accounted to the relevant Taxation Authority for the Taxation so deducted.
- 1.9 No Target Company is involved in any dispute with any Taxation Authority nor has it, within the past 3 years, been subject to any visit, audit, investigation, discovery or access order by any Taxation Authority. So far as the Sellers are aware, there are no circumstances existing at the date of this Agreement which make it likely that a non-routine visit or audit or investigation, discovery or access order will be made in the next 12 months.
- 1.10 The Disclosure Letter contains details of any concession, agreements or other formal or informal arrangement (that is, an arrangement which is not based on a strict interpretation of all relevant Taxation legislation, published extra-statutory concessions and published statements of practice) made by a Target Company with any Taxation Authority.
- 1.11 The Disclosure Letter contains details of all transactions, schemes or arrangements in respect of which any Target Company has been a party or has otherwise been involved for which a statutory clearance application was made. The Disclosure Letter also contains copies of all relevant applications for clearances and copies of all clearances obtained in connection with such transactions, schemes or arrangements. All such clearances have been obtained on the basis of full and accurate disclosure of all material facts and considerations relating thereto. All such transactions, schemes or arrangements have been implemented strictly in accordance with the terms of such clearances.
- 1.12 The Sellers are not aware of any Event which has occurred as a result of which a Target Company is, or will become, liable to make to any person (including any Taxation Authority) any payment in respect of any liability to Taxation which is primarily or directly chargeable against, or attributable to, any other person (other than the relevant Target Company).

76

- 1.13 The relevant Accounts make full provision or reserve within generally accepted accounting principles for any period ended on or before the date to which they were drawn up for all Taxation assessed or liable to be assessed on the relevant Target Company, or for which the relevant Target Company is accountable at that date, whether or not the relevant Target Company has (or may have) any right of reimbursement against any other person. Proper provision has been made and shown in the relevant Accounts for deferred taxation in accordance with generally accepted accounting principles.
- 1.14 Each of the Target Companies has duly and properly submitted all claims, disclaimers and elections the making of which have been assumed for the purposes of the Accounts and in respect of which the time limit therefor has passed.
- 1.15 Each Target Company has sufficient records to determine the Tax consequence which would arise on the disposal or realisation of any asset owned at the Last Accounts Date or acquired since then.

2. Chargeable gains

2.1 The book value shown in, or adopted for the purposes of, the relevant Accounts as the value of each of the assets of the relevant Target Company, on the disposal of which a chargeable gain or allowable loss could arise, does not exceed the amount which on a disposal of such asset at the date of this Agreement would be deductible, in each case, disregarding any statutory right to claim any allowance or relief other than amounts deductible under section 38 of TCGA 1992.

3. Capital allowances

- 3.1 If any asset of any of the Target Companies was disposed of at Closing for its book value as shown in, or adopted for the purpose of, the relevant Accounts, or for the value of consideration actually given for it on its acquisition (if such asset were acquired since the Last Accounts Date), no balancing charge under CAA 2001 (or any other legislation relating to capital allowances) or similar clawback of relief in jurisdictions outside the UK would be made on the relevant Target Company.
- 3.2 No event has occurred since the Last Accounts Date whereby any balancing charge may fall to be made against, or any disposal value may fall to be brought into account by, that Target Company under CAA 2001 (or any other legislation relating to capital allowances) or similar legislation relating to relief for similar capital expenditure in jurisdictions outside the UK.

4. Distributions and other payments

- 4.1 No distribution or deemed distribution, within the meaning of sections 209, 210 or 211 of ICTA 1988, has been made (or will be deemed to have been made) by any Target Company since its incorporation, except dividends shown in their audited accounts, and no Target Company is bound to make any such distribution.
- 4.2 No sums of an income nature in excess of £2,000, paid or payable by any Target Company, or which any Target Company is under an existing obligation to pay in the

future, are or may be wholly or partially disallowable as deductions, management expenses or charges in computing taxable profits for Taxation purposes.

4.3 No Target Company has, within the period of seven (7) years preceding Closing, been engaged in, nor been a party to, any of the transactions set out in sections 213 to 218 (inclusive) of ICTA 1988, nor has it made or received a chargeable payment as defined in section 218(1) of ICTA 1988.

5. Loan relationships

All interests, discounts and premiums payable by any Target Company in respect of its loan relationships (within the meaning of section 81 of the Finance Act 1996) are eligible to be brought into account by the relevant Target Company (as the case may be) as a debit for the purposes of Chapter II of Part IV of the Finance Act 1996 at the time, and to the extent that such debits are recognised in the statutory accounts of the relevant Target Company.

6. Close companies

- 6.1 No Target Company is, or has ever been a close investment holding company within the meaning of section 13A of ICTA 1988.
- 6.2 No distribution within section 418 of ICTA 1988 has been made by any Target Company during the last six (6) years ending at the Accounts Date, nor have such distributions been made between the Accounts Date and Closing.
- 6.3 Any loans or advances made, or agreed to be made, by any Target Company within sections 419 and 420 or 422 of ICTA 1988 in the six (6) years ending on Closing have been disclosed in the Disclosure Letter. No Target Company has released or written off, or agreed to release or write off, the whole or any part of any such loans or advances.

7. Group relief

7.1 Except as provided in the relevant Accounts, no Target Company is, or will be, obliged to make or be entitled to receive any payment in pursuance of an agreement as respects amounts surrendered by way of group relief to or by any Target Company in respect of any period ending on or before Closing, or any payment for the surrender of the benefit of an amount of advance corporation tax or any repayment of such a payment.

8. Groups of companies

- 8.1 None of the Target Companies have in the six (6) years ending on the date of this Agreement been a member of a group for the purposes of Chapter IV of Part X of ICTA 1988.
- 8.2 No Target Company holds any asset which it acquired in the six (6) years ending on the date of this Agreement from another company which, at the time of such acquisition, was a member of the same group of companies as that Target Company.

78

8.3 No Target Company has ever been party to any arrangements pursuant to section 36 of the Finance Act 1998 (group payment arrangements).

9. Intangible assets

- 9.1 For the purposes of this paragraph 9, references to **intangible fixed assets** mean intangible fixed assets and goodwill within the meaning of Schedule 29 to the Finance Act 2002 to which that Schedule applies. References to an **intangible fixed asset** shall be construed accordingly.
- 9.2 No claims or elections have been made by any Target Company under Part 7 of, or paragraph 86 of Schedule 29 to, the Finance Act 2002 in respect of any intangible fixed asset of any Target Company.
- 9.3 Since the Last Accounts Date:
 - (a) no asset of a Target Company has ceased to be a chargeable intangible asset in the circumstances described in paragraph 108 of Schedule 29 to the Finance Act 2002;
 - (b) no Target Company has realised or acquired an intangible fixed asset for the purposes of Schedule 29 to the Finance Act 2002; and
 - (c) no circumstances have arisen which have required, or will require, a credit to be brought into account by any Target Company on a revaluation of an intangible fixed asset.

10. Company residence, Treasury consents and overseas interests

- 10.1 The Target Companies have always been resident in the UK for corporation tax purposes and have never been treated as resident in any other jurisdiction for the purposes of any double taxation arrangements having effect under section 249 of the Finance Act 1994, section 788 of ICTA 1988 or for any other tax purpose.
- 10.2 No Target Company has caused, permitted or entered into any of the transactions specified in section 765 of ICTA 1988 (migration of companies) without the prior written consent of HM Treasury, or without having duly provided the required information to HM Revenue & Customs (as appropriate).

- 10.3 No Target Company holds shares in a company which is not resident in the UK and which would be a close company if it were resident in the UK in circumstances such that a chargeable gain accruing to the company not resident in the UK could be apportioned to any Target Company pursuant to section 13 of TCGA 1992.
- 10.4 No Target Company holds any interest in a controlled foreign company within section 747 of ICTA 1988. No Target Company has any material interest in an offshore fund as defined in section 759 of ICTA 1988.
- 10.5 No Target Company has ever had a permanent establishment outside the UK.

79

10.6 No Target Company is an agent or permanent establishment of another company, person, business or enterprise for the purpose of assessing the company, person, business or enterprise to Taxation in the United Kingdom.

11. Anti-avoidance

- So far as the Sellers are aware, all transactions or arrangements made by any of the Target Companies have been made on fully arm's length terms. So far as the Sellers are aware, there are no circumstances in which section 770A of, or Schedule 28AA to, ICTA 1988 or any other rule or provision could apply causing any Taxation Authority to make an adjustment to the terms on which such transaction or arrangement is treated as being made for Taxation purposes, and no notice or enquiry has been made by any Taxation Authority in connection with any such transactions or arrangements.
- 11.2 None of the Target Companies have entered into any notifiable arrangements for the purposes of Part 7 of the Finance Act 2004 or any notifiable schemes for the purposes of Schedule 11A to the VATA 1994.
- 11.3 None of the Target Companies have at any time been a party to or otherwise involved in a transaction or series of transactions in relation to which advisers considered that there was a significant risk the relevant Target Company could be liable to Taxation as a result of the principles in W T Ramsey Limited v IRC (54 TC 101) or Furniss v Dawson (55 TC 324) as developed in subsequent cases.

12. Inheritance tax

- 12.1 No Target Company has:
 - (a) made any transfer of value within sections 94 and 202 of IHTA 1984; or
 - (b) received any value such that liability might arise under section 199 of IHTA 1984; or
 - (c) been a party to associated operations in relation to a transfer of value as defined by section 268 of IHTA 1984.
- There is no unsatisfied liability to inheritance tax attached to, or attributable to, the Shares or any asset of any Target Company. None of them are subject to any HM Revenue & Customs charge as mentioned in section 237 and 238 of IHTA 1984.
- 12.3 No asset owned by any Target Company, and none of the Shares, are liable to be subject to any sale, mortgage or charge by virtue of section 212(1) of IHTA 1984 as a result of an Event occurring before the date of this Agreement.

13. Value Added Tax

Each of the Target Companies is a taxable person and duly registered for the purposes of VAT. Such registration is not pursuant to paragraph 2 of Schedule 1 to the VATA 1994 or subject to any conditions imposed by or agreed with HM Revenue & Customs. No Target Company is (nor are there any existing circumstances by virtue of which it may become) under a duty to make monthly payments on account under the Value Added Tax (Payments on Account) Order 1993. Each Target Company has

80

- complied in all material respects with all statutory provisions, rules, regulations, orders and directions in respect of VAT.
- No Target Company is or has been in the period of six (6) years ending with the date of this Agreement, a member of a group of companies for VAT purposes except as disclosed in the Disclosure Letter.
- All supplies made by any Target Company are taxable supplies. No Target Company has in the past three (3) years been, nor has any Event occurred before Closing and in the prescribed accounting period current at Closing as a result of which any Target Company will be, denied full credit for all input tax paid or suffered by it before the Closing.
- No act or transaction has been effected in consequence of which any Target Company is, or so far as the Sellers are aware may be, held liable to account to the relevant Tax Authority for any VAT arising from supplies made by another company. No direction has been given nor so far as the Sellers are aware has any Event occurred as a result of which such a direction will be given by HM Revenue & Customs under Schedule 9A to the VATA 1994 as a result of which any Target Company would be treated for the purposes of VAT as a member of a group.
- 13.5 For the purposes of paragraph 3(7) of Schedule 10 to the VATA 1994 no Target Company nor any relevant associate of such Target Company (within the meaning of paragraph 3(7) of Schedule 10 to the VATA 1994) has exercised the election to waive exemption from VAT (pursuant to paragraph 2 of Schedule 10 to the VATA 1994) in respect of any property other than those Properties listed as having been the subject of such an election in the Disclosure Letter.

- No Target Company owns, or has at any time within the period of 10 years preceding the date of this Agreement owned, any assets which are capital items subject to the capital goods scheme under Part XV of the VAT Regulations 1995.
- 13.7 So far as the Sellers are aware, there are no existing circumstances by virtue of which any refund of VAT obtained or claimed under section 36 VATA 1994 may be required to be repaid.

14. Stamp duty, stamp duty land tax and stamp duty reserve tax

- Any document that may be necessary or desirable in proving the title of any Target Company to any asset which is owned by that Target Company at Closing, and each document which any Target Company may wish to enforce or produce in evidence, is duly stamped for stamp duty purposes. No such documents which are outside the UK would attract stamp duty if they were brought into the UK.
- 14.2 Neither entering into this Agreement nor Closing will result in the withdrawal of any stamp duty or stamp duty land tax relief granted on or before Closing which will affect any Target Company.
- 14.3 The Disclosure Letter sets out full and accurate details of any chargeable interest (as defined under section 48 of the Finance Act 2003) acquired or held by any Target Company before Closing in respect of which an additional land transaction return will

81

be required to be filed with a Taxation Authority and/or a payment of stamp duty land tax made on or after Closing.

The Target Companies have complied in all respects with the provisions of Part IV of Finance Act 1986 (Stamp Duty Reserve Tax) and any regulations made under such legislation.

15. Tax sharing

14.4

No Target Company is party to any Taxation indemnity, Taxation sharing or any Taxation allocation agreement in respect of which claims against any Target Company would not be time barred.

82

Schedule 7

TAXATION SCHEDULE

1. Interpretation

1.1 The definitions and rules of interpretation in this paragraph apply in this Tax Covenant.

"Event" means any transaction (including the execution and completion of all provisions of this Agreement), event, act or omission whatsoever, including (without limitation), the expiry of a period of time, any Target Company becoming or ceasing to be associated with any other person for any Tax purpose or ceasing to be or becoming resident in any country for any Tax purpose, the death or the winding up or dissolution of any person, and any reference to an Event occurring on or before a particular date shall include Events which, for Tax purposes, are deemed to have, or are treated or regarded as having, occurred on or before that date.

"Liability for Taxation" means any liability of any Target Company to make a payment of or in respect of Tax, whether or not the same is primarily payable by the relevant Target Company and whether or not the relevant Target Company has or may have any right of reimbursement against any other person or persons and shall also include:

- (a) the Loss of any Relief where such Relief has been taken into account in computing and so reducing or eliminating any provision for deferred Tax which appears in the Last Accounts (or which but for such Relief would have appeared in the Last Accounts) or where such Relief was treated as an asset of the relevant Target Company in the Last Accounts or was taken into account in computing any deferred Tax asset which appears in the Last Accounts ("Accounts Relief"), in which case the amount of the Liability for Taxation shall be the amount of Tax which would (on the basis of Tax rates as at the Closing Date) have been saved but for such Loss, assuming for this purpose that the relevant Target Company had sufficient profits or was otherwise in a position to use the Relief;
- (b) the Loss of any right to repayment of Tax which was treated as an asset in the Last Accounts ("**Repayment Relief**"), in which case the amount of the Liability for Taxation shall be the amount of the repayment subject of the Loss;
- save where such set off or use is assumed in computing the provision for Tax in the Last Accounts, the set off or use against income, profits or gains earned, accrued or received or against any Tax chargeable in respect of an Event occurring on or before Closing of any Accounts Relief or Repayment Relief or any Relief which is not available before Closing but arises as a result of an Event occurring after Closing ("Post Closing Relief") in circumstances where, but for such set off or use, the relevant Target Company would have had a liability to make a payment of or in respect of Tax for which the Purchaser would have been able to make a claim against the relevant Seller under this

Tax Covenant, in which case the amount of the Liability for Taxation shall be the amount of Tax saved by the relevant Target Company as a result of such set off or use;

(d) any liability of any Target Company to make a payment pursuant to an indemnity, guarantee or covenant entered into before Closing under which the relevant Target Company has agreed to meet or pay a sum equivalent to another person's Tax liability, in which case the Liability for Taxation shall be equal to the amount of the liability.

"Loss" means any reduction, modification, loss, counteraction, nullification, disallowance or clawback for whatever reason.

"Purchaser's Relief" means:

- (a) any Accounts Relief (as defined in paragraph (a) of the definition of Liability for Taxation) or Repayment Relief (as defined in paragraph (b) of the definition of Liability for Taxation);
- (b) any Post Closing Relief (as defined in paragraph (c) of the definition of Liability for Taxation); and
- (c) any Relief, whenever arising, of the Purchaser or any member of the Purchaser's Tax Group other than any Target Company.

"Purchaser's Tax Group" means the Purchaser and any other company or companies which either are or become after Closing, or have within the seven years ending at Closing, been treated as members of the same group as, or otherwise connected or associated in any way with, the Purchaser for any Tax purpose.

"Relief" means any loss, relief, allowance, credit, exemption or set off in respect of Tax or any deduction in computing income, profits or gains for the purposes of Tax and any right to a repayment of Tax.

"Saving" means the reduction or elimination of any liability of any Target Company to make an actual payment of corporation tax in respect of which the relevant Sellers would not have been liable under paragraph 2 or the obtaining of a repayment of Tax, by the use of any Relief to the extent arising as a result of a Liability for Taxation (or the matter or circumstances giving rise to such Liability for Taxation) in respect of which the relevant Sellers have made a payment under paragraph 2 of this Tax Covenant or the Tax Warranties.

"Tax" means all forms of taxation and statutory, governmental, state, federal, provincial, local, government or municipal charges, duties, imposts, contributions, levies, withholdings or liabilities wherever chargeable and whether of the UK or any other jurisdiction (but excluding business property rates, council taxes, water rates and all other similar or equivalent municipal rates, levies and charges), and any penalty, fine, surcharge, interest or Tax Authority imposed charges relating thereto (and penalties, fines, surcharges, interest or Tax Authority imposed charges attributable to the late filing of any notice, return or other document required for the purposes of any Tax), and "Taxation" shall have the same meaning.

84

"**Taxation Authority**" means HM Revenue and Customs, Department of Social Security and any other governmental or other authority whatsoever competent to impose any Tax whether in the United Kingdom or elsewhere.

"Tax Assessment" means, for the purposes of this Tax Covenant, any assessment (including self-assessment), notice, demand, letter or other document issued or action taken by or on behalf of any Taxation Authority or Target Company from which it is evident that any Target Company is or will be subject to a Liability for Taxation or other liability in respect of which any of the Sellers are or may be liable under this Tax Covenant or the Tax Warranties.

- 1.2 References to gross receipts, income, profits or gains earned, accrued or received shall include any gross receipts, income, profits or gains deemed pursuant to the relevant Taxation Statute to have been or treated or regarded as earned, accrued or received.
- 1.3 References to a repayment of Tax shall include any repayment supplement or interest in respect of it.
- 1.4 A reference to an Event occurring on or before Closing includes a series or combination of Events, all of which occurred on or before Closing.
- 1.5 Unless the contrary intention appears, words and expressions defined in this Agreement have the same meaning in this Tax Covenant and any provisions in this Agreement concerning matters of construction or interpretation also apply in this Tax Covenant.

2. Covenant

- 2.1 The Sellers covenant with the Purchaser that, subject to the provisions of this Tax Covenant, the Sellers shall be severally liable, *pro rata* to their respective Set of Shares, and in respect only of that Target Company in which the relevant Set of Shares is held and subject further to the provisions of clause 11.2 of this Agreement, which shall apply *mutatis mutandis* to determine the liability of any particular Seller under this Schedule, to pay to the Purchaser by way of repayment of the Final Share Price for their respective Set of Shares, to the extent possible, but not so as to limit the amount payable where not wholly possible, an amount equal to any:
 - (a) Liability for Taxation arising as a result of any Event occurring on or before Closing or in respect of any gross receipts, income, profits or gains earned, accrued or received by the relevant Target Company on or before Closing;
 - (b) Liability for Taxation which arises in consequence of an Event occurring at any time for which the relevant Target Company is liable as a result of the relevant Target Company having at any time before Closing been a member of a group for Tax purposes or controlled by any person;
 - (c) Liability for Taxation falling within paragraph (a) to paragraph (c) of the definition of Liability for Taxation;
 - (d) any Liability for Taxation which is a liability for inheritance tax which:

- (i) arises as a result of a transfer of value occurring or being deemed to occur on or before Closing (whether or not in conjunction with the death of any person whensoever occurring); or
- (ii) has given rise at Closing to a charge on any of the Shares or assets of the relevant Target Company; or
- (iii) gives rise after Closing to a charge on any of the Shares in or assets of the relevant Target Company as a result of the death of any person within seven (7) years of a transfer of value which occurred before Closing;

and for these purposes the provisions of section 213 of IHTA 1984 (refund by instalments) shall be deemed not to apply to any liability for inheritance tax within this paragraph 2; and

- (e) reasonable costs and expenses properly incurred by the Purchaser and/or the relevant Target Company in connection with any Liability for Taxation within this paragraph 2.1 for which the Sellers are liable to make a payment under this Schedule and the successful enforcement of any rights under this Tax Covenant.
- 2.2 For the purposes of this Tax Covenant, in determining whether a charge on the Shares in or assets of the relevant Target Company arises at any time or whether there is a liability for inheritance tax, the fact that any Tax may be paid in instalments shall be disregarded and such Tax shall be treated for the purposes of this Tax Covenant as becoming due or to have become due and a charge as arising or having arisen on the date of the transfer of value or other date or Event on or in respect of which it becomes payable or arises.
- 2.3 The aggregate liability of the Sellers in respect of claims under paragraph 2.1(a), (b) and (c) and any related claim under 2.1(e) shall be limited to £1,000,000 and the aggregate liability of the Sellers in respect of all such claims together with all other Claims (including claims under 2.1(d) (inheritance tax) and any related claim under 2.1(e)) shall be limited to the Final Share Price.

3. Payment date and interest

- 3.1 Where any of the Sellers are liable to make any payment under paragraph 2 (including any payment pursuant to paragraph 2.1(e)), the payment shall be made in cleared funds and the due date for the making of that payment ("**Due Date**") shall be the later of the date falling five (5) Business Days after the Purchaser has served a notice on the relevant Seller demanding that payment and in a case:
 - (a) that involves an actual payment of Tax three (3) Business Days before the date on which the Tax in question would have had to have been paid to the relevant Taxation Authority in order to prevent a liability to interest or a fine, surcharge or penalty from arising in respect of the Liability for Taxation in question; or
 - (b) that falls within paragraph (a) of the definition of Liability for Taxation, the last date upon which the Tax is or would have been required to be paid to the

86

- relevant Taxation Authority in respect of the period in which the Loss of the Relief occurs (assuming for this purpose that the relevant Target Company had sufficient profits or was otherwise in a position to use the Relief); or
- (c) that falls within paragraph (b) of the definition of Liability for Taxation, the date upon which the repayment was due from the relevant Taxation Authority; or
- (d) that falls within paragraph (c) of the definition of Liability for Taxation, the date upon which the Tax saved by the relevant Target Company would have been required to be paid to the relevant Taxation Authority; or
- (e) that falls within paragraph (d) of the definition of Liability for Taxation not later than the third Business Day before the day on which the relevant Target Company is due to make the payment; or
- (f) that falls within paragraph 2.1(e), three (3) Business Days before the costs and expenses are due to be paid.
- 3.2 Any dispute as to the amount specified in any notice served on the relevant Seller under paragraph 3.1(b) to paragraph 3.1(d) shall be determined by the auditors of the relevant Target Company for the time being, acting as experts and not as arbitrators (the costs of that determination being shared equally by the relevant Seller and the Purchaser).
- 3.3 If any sums required to be paid by any of the Sellers under this Tax Covenant are not paid on the Due Date, then, except to the extent that the relevant Seller's liability under paragraph 2 compensates the Purchaser for the late payment by virtue of it extending to interest and penalties, such sums shall bear interest (which shall accrue from day to day after as well as before any judgment for the same) at the rate of Default Interest from the day following the Due Date up to and including the day of actual payment of such sums, such interest to be compounded quarterly.

4. Exclusions

- In addition to the exclusions set out in paragraph 4.2 below, the liability of the Sellers under the covenant contained in paragraph 2 shall be subject to and limited by the provisions of paragraphs 1 (time limits), 3 (maximum limits), and 8 (no double recovery) of Schedule 9 to this Agreement.
- 4.2 Without limitation to paragraph 4.1 (and the provisions of the said Schedule 9) the covenant contained in paragraph 2 shall not cover any Liability for Taxation (and the Sellers shall not be liable under the Tax Warranties in respect of any liability) to the extent that:

Closing, or the Sellers' failure to comply or their delay in complying with any of their obligations in this Tax Covenant; and/or

- (b) it would not have arisen but for the failure by the Purchaser or a Target Company to make a payment to the relevant Tax Authority by the due date for payment of an amount of Tax equal to the payment made by the Sellers in respect of such Tax under this Tax Covenant or the Tax Warranties except to the extent that such delay is a consequence of the failure of the Sellers to make the payment on the due date according to the provisions of this Schedule or the Agreement (as the case may be);
- (c) the Purchaser or the relevant Target Company has already recovered any amount in respect of the liability in question under any other provision of this Agreement or under this Tax Covenant or the liability has been made good or otherwise compensated for by any other person (other than the Purchaser or a member of the Purchaser Group) net of the costs of such recovery (including additional Taxation in respect of any sum recovered) suffered by the Purchaser or any member of the Purchaser Group.
- (d) the Liability for Taxation arises or is increased by virtue of the relevant Target Company ceasing to be entitled to the smaller companies or marginal rate of corporation tax (or ceasing to be entitled to any Relief) as a result of the acquisition of the relevant Set of Shares by the Purchaser.
- (e) provision (including, without limitation, deferred tax provision), or reserve was made in the Last Accounts or (as the case may be a provision or accrual was made in the Closing Statement) for, or in respect of, the liability or other matter giving rise to the Liability for Taxation.
- (f) the Liability for Taxation arises or occurs or is increased (or any provision or reserve made in the Last Accounts (or, as the case may be, a provision or accrual made in the Closing Statement) in respect of the liability is insufficient) as a result of any retrospective change in legislation after the Closing Date or the introduction of any retrospective legislation not in force at the Closing Date or as a result of the withdrawal after the Closing Date of any published concession or published practice previously made by HM Revenue & Customs or other Taxation Authority.
- (g) the Liability for Taxation is increased (or any provision or reserve made in the Last Accounts (or, as the case may be a provision or accrual made in the Closing Statement) in respect of the liability is insufficient) as a result of any increase (with retrospective effect) in the rate or rates of Taxation in force at the Closing Date.
- (h) the Liability for Taxation arises as a result of the Purchaser or a Target Company disclaiming after Closing any part of the benefit of capital or other allowances against Taxation properly and validly claimed before Closing or assumed in computing the provision for Tax in the Last Accounts (or, as the case may be, in the Closing Statement) of that Target Company, unless such disclaimer is made at the written request of any Sellers or his duly appointed agents.

88

- the Liability for Taxation would not have arisen or would have been reduced or eliminated but for the failure or omission on the part of the Purchaser or the relevant Target Company after the date of Closing to make any proper and valid claim election surrender or disclaimer or give notice or consent or do any other thing under the provisions of any enactment or regulation relating to Taxation after the date of Closing the making giving or doing of which was taken into account in computing the provision for Taxation in the Last Accounts or (as the case may be) in the Closing Statement of the relevant Target Company and in the case of the Last Accounts, sufficient details of which were provided in writing to the Purchaser no less than 20 Business Days before the last date on which such claim, election surrender or disclaimer could be made, unless such failure or omission is at the request of the Sellers' Representative or his duly appointed agents.
- (j) the Liability for Taxation would not have arisen but for a change after Closing in the accounting policies or practice adopted by any Target Company or in the accounting reference date of any Target Company unless such change is required to secure compliance with UK GAAP or the requirements of any Taxation Authority or any other legal requirements in force at Closing in circumstances where the relevant Target Company was not compliant therewith before Closing.
- (k) the Liability for Taxation arises in connection with any matter or thing done or omitted to be done prior to Closing by the Sellers or a Target Company at the written request of or with the specific written approval of the Purchaser, its representatives or its professional advisors or any of them.
- (l) The Liability for Taxation arises in respect of actual as opposed to deemed income, profits or gains which are actually received by a Target Company but which were not reflected in the Closing Statement;

and/or

- (m) such Liability for Taxation would not have arisen but for a voluntary act or transaction carried out by the Purchaser or the relevant Target Company after Closing otherwise than:
 - (i) (in the case of a Target Company) in the ordinary course of business of the Target Company as such business is conducted at Closing; or

- (ii) pursuant to any legally binding obligation incurred by a Target Company before Closing; or
- (iii) pursuant to the requirements of any applicable law or regulation in force at Closing; or
- (iv) a disclosure to or the submission of any return or other document which in each case is lawfully required by the Taxation Authority.

89

Savings

- If (at the relevant Seller's request and expense) the auditors for the time being of the Target Companies determine that the relevant Target Company has obtained and utilised a Saving, the Purchaser shall repay to the relevant Seller the lesser of:
 - (a) the amount of the Saving (as determined by the auditors) less any costs incurred by the Purchaser or the relevant Target Company which are directly referable to obtaining the same; and
 - (b) the amount paid by the relevant Seller under paragraph 2 or the Tax Warranties in respect of the Liability for Taxation which gave rise to the Saving less any part of that amount previously repaid to the relevant Seller in respect thereof under any provision of this Tax Covenant or otherwise.

6. Recovery

- If the Purchaser, a Target Company or any member of the Purchaser Group is or becomes entitled to recover from any other person (not being a member of the Purchaser Group but including a Tax Authority) any sum in respect of any Liability for Taxation in relation to which a payment has been made by the Sellers under this Schedule or the Tax Warranties, the Purchaser shall notify the Sellers accordingly, and provided that the Sellers shall first indemnify the Purchaser or the relevant Target Company, as appropriate, to the Purchaser's reasonable satisfaction against all reasonable costs which it/they may incur in connection therewith, the Purchaser shall, and shall procure that the relevant Target Company shall, take such action as the Sellers' Representative may reasonably request in writing to enforce the recovery against the other person. Provided that the Purchaser or the Target Company shall not be obliged to take any action which in its discretion (acting in good faith) it determines likely to be prejudicial to its relationship with any person.
- In the event that any such sum as is referred to in paragraph 6.1 is recovered by a Target Company or any other member of the Purchaser Group (whether or not as a result of taking any such action as is referred to in paragraph 6.1), the Purchaser shall account to the Sellers within 10 Business Days of the recovery of such sum for whichever is the lesser of:
 - (a) any sum so recovered less any Tax chargeable in respect thereof, and after deduction of all reasonable costs and expenses properly incurred in enforcing such recovery not already recovered pursuant to the indemnity referred to in paragraph 6.1; and
 - (b) the aggregate of the amount paid by the Sellers pursuant to this Schedule or the Tax Warranties in relation to the Liability for Taxation to which paragraph 6.1 refers,

together with any interest and/or repayment supplement (less Tax thereon) received in respect of the sum recovered to the extent that such interest and/or repayment supplement relates to the Liability for Taxation to which paragraph 6.1 refers and relates to the period after the Sellers made payment under this Schedule or the Tax Warranties.

90

7. Overprovisions

- 7.1 If any provision for Tax (other than deferred tax) made in the Last Accounts or the Closing Statement proves to be an overprovision (or if any right to repayment proves to be understated therein) ("**Overprovision**") then such Overprovision shall be dealt with as follows:
 - (a) first it shall be set-off against any amount for which the Sellers are liable under this Schedule or the Tax Warranties;
 - (b) to the extent of any excess it shall be used to repay to the Sellers any amount already paid by the Sellers in respect of a Liability for Taxation (under this Schedule or the Tax Warranties); and
 - (c) any balance shall be carried forward and offset against any future liability of the Sellers to make any payment or payments under this Schedule or the Tax Warranties.
- 7.2 The Purchaser shall notify the Sellers as soon as reasonably practicable of any Overprovision and the Sellers' Representative shall be entitled (at the Sellers' expense) to require that the auditors for the time being of the relevant Target Company certify the existence and amount of any Overprovision.

8. Corporation tax returns

8.1 The Purchaser shall or shall procure that the relevant Target Company shall provide to the Sellers' Representative for his comment (in the case of returns relating to accounting periods ending on or after the Last Accounts Date no later than 15 Business Days prior to the due date) for submission drafts of the corporation tax returns of each Target Company for all accounting periods ended on or before Closing and (to the extent relating to matters for which the Sellers could be liable under this Tax Covenant) the accounting period current at Closing and shall procure that all such reasonable and timely comments of the Sellers (and/or their professional advisers) in respect of such tax returns are taken into account therein.

- 8.2 The Sellers shall provide all such information and assistance as the Purchaser may reasonably request in relation to the preparation of the tax returns, computations and other such documents relating to any period prior to Closing.
- 8.3 For the avoidance of doubt:
 - (a) where any matter relating to Tax gives rise to a Tax Claim, the provisions of paragraph 9 shall take precedence over the provisions of this paragraph 8; and
 - (b) the provisions of this paragraph 8 shall not prejudice the rights of the Purchaser to make a Tax Claim under this Tax Covenant or the Tax Warranties in respect of any Liability for Taxation.

9. Conduct of Tax Claims

9.1 If the Purchaser or any Target Company becomes aware of a Tax Assessment, the Purchaser shall give or procure that notice in writing is given to the relevant Seller

91

within a reasonable period (and in any event where there is a time limit for the making of an appeal or the taking of any other relevant action, no later than the later of 13 Business Days thereafter and 10 Business Days prior to the expiry of such time limit for the making of an appeal or the taking of any other relevant action), provided that if the relevant Seller receives any Tax Assessment for whatever reason, he shall notify the Purchaser in writing as soon as is reasonably practicable and the Purchaser shall be deemed on receipt of such notification to have given the relevant Seller notice of such Tax Assessment in accordance with the provisions of this paragraph 9 provided always that the giving of such notice shall not be a condition precedent to the relevant Seller's liability under this Tax Covenant.

- 9.2 Provided the relevant Seller indemnifies the Purchaser and the relevant Target Company (on an after-tax basis) to the Purchaser's reasonable satisfaction against all liabilities, costs, damages or expenses which may be incurred thereby including any additional Liability for Taxation, the Purchaser shall take and shall procure that the relevant Target Company shall take such action as the Sellers' Representative may reasonably request by notice in writing given to the Purchaser or the relevant Target Company to avoid, dispute, defend, resist, appeal or compromise any Tax Assessment (such a Tax Assessment where action is so requested being hereinafter referred to as a "Dispute").
- 9.3 If:
 - (a) the Sellers' Representative does not request the Purchaser or the relevant Target Company to take any action under paragraph 9.2 or the Sellers fail to indemnify the Purchaser or the relevant Target Company to the Purchaser's reasonable satisfaction within a period of time (commencing with the date of the notice given to the relevant Seller) that is reasonable, having regard to the nature of the Tax Assessment and the existence of any time limit in relation to avoiding, disputing, defending, resisting, appealing or compromising such Tax Assessment, and which period shall not in any event exceed a period of 15 Business Days; or
 - (b) it is alleged by a Tax Authority in writing, and such allegation is not subsequently withdrawn, the relevant Seller (or the relevant Target Company before Closing) has been involved in a case involving fraudulent conduct or negligent conduct (as that term is interpreted for the purposes of Schedule 18 Finance Act 1998) in respect of the Liability for Taxation which is the subject matter of the Dispute or the relevant Seller takes action or other steps are taken or legal proceedings are started for his bankruptcy; or
 - (c) the Dispute involves an appeal against a determination by the General or Special Commissioners or the VAT and Duties Tribunal, unless the relevant Seller has obtained the opinion of Tax counsel of at least six (6) years' standing that on a balance of probabilities the appeal will succeed,

the Purchaser and the Target Company shall have the conduct of the Dispute absolutely (without prejudice to its rights under this Tax Covenant) and shall be free to pay or settle the Tax Claim on such terms as the Purchaser or the Target Company may in its absolute discretion (acting in good faith) consider fit.

92

- 9.4 Subject to paragraph 9.3, by agreement in writing between the Purchaser and the Sellers' Representative, the conduct of a Dispute may be delegated to the Sellers' Representative upon such terms as may be agreed from time to time between the Purchaser and the Sellers' Representative provided that, unless the Purchaser and the Sellers' Representative specifically agree otherwise in writing, the following terms shall be deemed to be incorporated into any such agreement:
 - (a) the Purchaser or the relevant Target Company shall promptly be kept fully informed of all matters pertaining to a Dispute and shall be entitled to see and keep copies of all correspondence and notes or other written records of telephone conversations or meetings and, in the event that there is no written record, shall be given an immediate report of material all telephone conversations with any Taxation Authority to the extent that it relates to a Dispute;
 - (b) the appointment of solicitors or other professional advisers shall be subject to the written approval of the Purchaser, such approval not to be unreasonably withheld or delayed;
 - (c) all material written communications pertaining to the Dispute which are to be transmitted to the relevant Taxation Authority shall first be submitted to the Purchaser or the relevant Target Company for approval and shall only be finally transmitted if such approval is given, such approval not to be unreasonably withheld or delayed; and

- (d) the Sellers' Representative shall make no settlement or compromise of the Dispute or agree any matter in the conduct of the Dispute which is likely to affect the amount thereof or the future liability to Tax of the relevant Target Company without the prior written approval of the Purchaser or the relevant Target Company, such approval not to be unreasonably withheld or delayed.
- 9.5 The Purchaser shall provide and shall procure that the relevant Target Company provides to the Sellers' Representative and the Sellers' Representative's professional advisors reasonable access to premises and personnel and to any relevant assets, documents and records within their power, possession or control for the purpose of investigating the matter and enabling the Sellers' Representative to take such action as is referred to in this paragraph 9.
- 9.6 Neither the Purchaser nor any Target Company shall be subject to any claim by or liability to any of the Sellers for non-compliance with any of the foregoing provisions of this paragraph 9 if the Purchaser and any Target Company have bona fide acted in accordance with the instructions of the Sellers' Representative.

10. Grossing up

All sums payable by any of the Sellers to the Purchaser under this Tax Covenant shall be paid free and clear of all deductions or withholdings whatsoever unless the deduction or withholding is required by law. If any deductions or withholdings are required by law to be made from any of the sums payable under this Tax Covenant, the relevant Seller shall pay to the Purchaser such sum as will, after the deduction or withholding has been made (and after taking into account any credit in respect of Tax

93

- to which the Purchaser is entitled in respect of any such deduction or withholding), leave the Purchaser with the same amount as it would have been entitled to receive in the absence of any such requirement to make a deduction or withholding.
- If the Purchaser incurs a Taxation liability which results from, or is calculated by reference to, any sum paid under this Tax Covenant, the amount so payable shall be increased by such amount as will ensure that, after payment of the Taxation liability, the Purchaser is left with a net sum equal to the sum it would have received had no such Taxation liability arisen.
- 10.3 If the Purchaser would, but for the availability of a Purchaser's Relief, incur a Taxation liability falling within paragraph 10.2, it shall be deemed for the purposes of that paragraph to have incurred and paid that liability.
- The Sellers shall not be liable to make any payment or payments under either paragraph 10.1 or paragraph 10.2 to the extent such liability would not have arisen but for any assignment of or other succession to the benefit of this Schedule or any other provision of this Agreement.

11. Purchaser's Undertaking

- 11.1 The Purchaser shall pay to the Sellers within five (5) Business Days of written demand (accompanied by reasonable evidence that an amount is due under this paragraph) an amount equal to any Tax assessed on the Sellers (or any of them or any person connected with any of them) at any time (together with reasonable costs and expenses properly incurred by the Sellers or such other person in relation to such Tax or assessment) by reason of a Target Company failing after Closing to pay Tax for which it is primarily liable, unless such failure arises out of the Sellers' failure or delay in making a payment to the Purchaser under this Schedule or the Tax Warranties.
- 11.2 The provisions of paragraphs 3.3 and 10 of this Schedule shall apply *mutatis mutandis* to any payment made or required to be made by the Purchaser under this paragraph 11.

94

Schedule 8

PURCHASER WARRANTIES

- 1. The Purchaser is validly incorporated, in existence and duly registered under the laws of its jurisdiction and has full power to conduct its business as conducted at the date of this Agreement.
- 2. The Purchaser has obtained all corporate authorisations and all other governmental, statutory, regulatory or other consents, licences, authorisations, waivers or exemptions required to empower it to enter into and perform its obligations under this Agreement where failure to obtain them would adversely affect to a material extent its ability to enter into and perform its obligations under this Agreement.
- 3. This Agreement and the Transaction Documents which are to be entered into by the Purchaser will, when executed, constitute valid and binding obligations of the Purchaser.
- 4. Entry into and performance by the Purchaser of this Agreement and/or any Transaction Document to which it is a Party will not (i) breach any provision of its memorandum and articles of association, by-laws or equivalent constitutional documents or (ii) result in a breach of any laws or regulations in its jurisdiction of incorporation or of any order, decree of judgment of any court or any governmental or regulatory authority, where any such breach would affect to a material extent its ability to enter into or perform its obligations under this Agreement and/or any Transaction Document to which it is a Party.

Schedule 9

LIMITATIONS ON LIABILITY

1. Time Limits

No Seller shall be liable for any Claim unless that Seller receives from the Purchaser written notice containing such details as are then available of the matter giving rise to the Claim ("Notice of Claim"):

- (a) prior to the second anniversary of Closing in the case of a Non-Tax Claim; or
- (b) prior to the seventh anniversary of Closing in the case of a Tax Claim.

2. Thresholds for Claims

No Seller shall be liable for a Warranty Claim:

- (a) unless the amount of the liability pursuant to that Claim (and, for these purposes, a number of Claims arising out of the same or similar subject matter, facts, events or circumstances may be aggregated and form a single Claim) exceeds £5,000 (and in each case the Purchaser shall be able to claim the whole amount of any Claim reaching the relevant threshold and not merely the excess);
- (b) unless the aggregate amount of the liability of a Seller for all Warranty Claims not prohibited by paragraph (a) above in relation to the assets or affairs of Geneva Industries Limited exceeds £50,000 (in which case the Purchaser shall be entitled to claim the whole amount of such Claims and not merely the excess); and
- (c) unless the aggregate amount of the liability of the Sellers for all Claims in relation to the assets or affairs of Willen Trading Limited not prohibited by paragraph (a) above excess £50,000 (in which case the Purchaser shall be entitled to claim the whole amount of such Claims and not merely the excess); and
- (d) unless the aggregate amount of the liability of the Sellers for all Warranty Claims not prohibited by paragraph (a) above in relation to the assets or affairs of Geneva Auctions Limited exceeds £25,000 (in which case the Purchaser shall be entitled to claim the whole amount of such Claims and not merely the excess).

For the avoidance of doubt, the Purchaser may give notice of any single Claim in accordance with and for the purpose of paragraph 1 above, irrespective of whether, at the time the notice is given, the amount set out in paragraph 2(b) has been exceeded.

3. **Maximum limit for all Claims**

3.1 The aggregate amount of the liability of each Seller in respect of a breach of:

96

- (a) the Warranties in clause 11.1 of the Agreement;
- (b) paragraphs 1.1 and 1.2 of Schedule 6;
- (c) Part B of Schedule 6; and
- (d) paragraph 12 of Part G of Schedule 6,

shall not exceed the Final Share Price.

- 3.2 The aggregate amount of the liability of each Seller in respect of any Claim other than a Claim arising out of:
 - (a) paragraph 3.1 above; or
 - (b) breach of the Tax Covenant,

shall not exceed his Escrow Amount (together with any interest accrued thereon).

4. Claim to be withdrawn unless litigation commenced

Any Claim (other than Tax Claims) shall (if it has not been previously satisfied, settled or withdrawn) be deemed to have been withdrawn unless legal proceedings in respect of it have commenced by both being issued and served within nine (9) months of notification to the relevant Seller pursuant to paragraph 1 above except:

- (a) where the Claim relates to a contingent liability, in which case it shall be deemed to have been withdrawn unless legal proceedings in respect of it have been commenced by being both issued and served within nine (9) months of it having become an actual liability; or
- (b) where the Claim is a Claim for breach of Warranty of which notice is given for the purpose of paragraph 1 above at a time when the amount set out in paragraph 2(b) has not been exceeded, in which case it shall be deemed to have been withdrawn unless legal proceedings in respect of it have been commenced by being both issued and served within nine (9) months of the date of any subsequent notification to the relevant Seller pursuant to paragraph 1 above of one or more Claims for breach of Warranty which result(s) in the total amount claimed in

all Claims for breach of Warranty notified to the relevant Seller pursuant to paragraph 1 exceeding the amount set out in paragraph 2(b) for the first time.

5. Limitations not applicable if fraud etc.

None of the limitations contained in paragraphs 1, 2, 3 and 4 shall apply to any Claim to the extent that the Claim (or the delay in discovery of it) is the consequence of, or is increased as a consequence of, dishonest or deliberate mis-statement or concealment or other fraud by the relevant Seller.

97

6. Matters Disclosed or taken into account

A Seller shall not be liable for any Claim for breach of Warranty if and to the extent that the fact, matter, event or circumstance giving rise to such claim was Disclosure Letter and / or the Disclosure Bundle.

7. **Contingent liabilities**

If any Claim for breach of Warranty is based upon a liability which is contingent only, a Seller shall not be liable to make payment unless and until the contingent liability gives rise to an obligation to make a payment. This is without prejudice to the right of the Purchaser to give notice of the Claim in accordance with paragraph 1 and to issue and serve proceedings in respect of it before such time. For the avoidance of doubt, the fact that the liability may not have become an actual liability by the relevant date provided in paragraph 1 shall not exonerate a Seller in respect of any Claim properly notified before that date.

8. No double recovery

For the avoidance of doubt the Purchaser shall not be entitled to recover damages in respect of any Claim and/or to recover pursuant to the Tax Covenant to the extent recovery would involve recovery more than once in respect of the same loss, liability or damage.

9. Seller to have opportunity to remedy breaches

A breach of any Warranty which is capable of remedy shall not entitle the Purchaser to compensation unless the relevant Seller is given written notice of the breach by the Purchaser and the breach is not remedied within 30 days after the date on which notice is served on the relevant Seller. If the breach has not been remedied within that 30 day period, then the date on which notice of a Claim in respect of that breach shall be deemed to have been given to the relevant Seller for the purpose of paragraph 1 above shall be the date on which notice was given under this paragraph 9, provided that the notice satisfied the other requirements of paragraph 1 above when so given.

10. Further Limitations

- 10.1 The Sellers shall not be liable for (or such liability shall be reduced in respect of) a Claim (other than a Tax Claim):
 - (a) to the extent that a specific and identified allowance, provision (including, without limitation, deferred tax provision), or reserve was made in the Last Accounts or in the Closing Statement, for or in respect of the liability or other matter giving rise to the Warranty Claim;
 - (b) to the extent that such Claim arises or occurs or is increased (or any provision, reserve or allowance made in the Last Accounts in respect of the liability or other matter giving rise to such Claim is insufficient) as a result of any retrospective change in legislation after the Closing Date or the introduction of any retrospective legislation not in force at the Closing Date or as a result of the withdrawal after the Closing Date of any published concession or

98

published practice previously made by HM Revenue & Customs or other Taxation Authority;

- (c) to the extent that the Claim is increased (or any specific and identified provision, reserve or allowance was made in the Last Accounts in respect of the liability giving rise to such Claim is insufficient) as a result of any increase (with retrospective effect) in the rate or rates of Taxation in force at the Closing Date;
- (d) to the extent that the Claim arises as a result of the Purchaser or a Target Company disclaiming after Closing any part of the benefit of capital or other allowances against Taxation properly and validly claimed or proposed to be claimed before Closing and assumed in computing the provision for Tax in the Last Accounts of that Target Company and which the Purchaser was aware or should reasonably have been aware would give rise to a Claim, unless such disclaimer is made at the written request of any of the Sellers' Representative or his duly appointed agents;
- (e) to the extent that the Claim would not have arisen or would have been reduced or eliminated but for the failure or omission on the part of the Purchaser or the relevant Target Company after the date of Closing to make any proper and valid claim election surrender or disclaimer or give notice or consent or do any other thing under the provisions of any enactment or regulation relating to Taxation after the date of Closing the making giving or doing of which was taken into account in computing the provision for Taxation in the Last Accounts of the relevant Target Company and was set out in the Disclosure Letter and Disclosure Bundle, unless such failure or omission is at the request of the Sellers' Representative or his duly appointed agents;
- (f) to the extent that the Claim would not have arisen but for a change after Closing in the accounting policies or practice adopted by any Target Company or in the accounting reference date of any Target Company unless such change is required to secure compliance with UK

GAAP or the requirements of any Taxation Authority or any other legal requirements in force at Closing in circumstances where the relevant Target Company was not compliant therewith before Closing;

- (g) to the extent that the Purchaser or the relevant Target Company has already recovered any amount in respect of the liability in question under any other provision of this Agreement or under the Tax Covenant or the Claim has been made good or otherwise compensated for by any other person (other than the Purchaser or a member of the Purchaser Group) at no cost to the Purchaser or any member of the Purchaser Group;
- (h) to the extent the liability arises in connection with any matter or thing done or omitted to be done prior to Closing by the Sellers or a Target Company at the written request of or with the specific written approval of the Purchaser, its representatives or its professional advisors or any of them;

99

- (i) to the extent that such Claim or liability would not have arisen but for a voluntary act or transaction carried out by the Purchaser or the relevant Target Company after Closing otherwise than;
 - (i) (in the case of a Target Company) in the ordinary course of business of that Company as such business is conducted at Closing; or
 - (ii) pursuant to any legally binding obligation incurred by a Target Company before Closing; or
 - (iii) pursuant to the requirements of any applicable law or regulation in force at Closing; or
 - (iv) a disclosure to or the submission of any return or other document which in each case is lawfully required by the Taxation Authority.
- If any of the Sellers make any payment to or for the benefit of the Purchaser in respect of a Warranty Claim, other than a Claim under the Tax Warranties, and the Purchaser or a Target Company subsequently recovers (whether by payment, discount, credit or otherwise) from a third party (including any Taxation Authority) any amount which is referable to any part of the loss for which payment was made by or on behalf of the Sellers the Purchaser shall as soon as reasonably practicable repay or procure the repayment to the Sellers of so much of the amount paid by the Sellers as does not exceed the amount recovered (less any reasonable costs and expenses incurred reasonably and properly in effecting any such recovery and any Taxation which may be payable by the Purchaser or the relevant Target Company as a result of its receipt).
- 10.3 If, in respect of any matter which would give rise to a Claim, the Purchaser or the relevant Target Company is entitled to make a claim under any policy of insurance, then no such matter shall be the subject of a Claim and no Claim shall lie unless and until the Purchaser or the Target Company (as the case may be) has made a claim against its insurers. Liability in respect of any such Claim shall then be reduced by the amount recovered under such policy of insurance less all reasonable costs, charges and expenses (and Taxation which may be payable by the Purchaser or the relevant Target Company) incurred by the Purchaser in recovering that sum from its insurers, or extinguished if the amount recovered exceeds the amount of the Claim (less reasonable costs, charges and expenses and Taxation), provided that the time elapsed to process the claim under the policy of insurance shall not be taken into account for the purposes of the time limit to submit such Claim to the Sellers under paragraph 1 of this Schedule 9.
- Where the Purchaser or a Target Company is or may be entitled to recover from some other person any sum in respect of any matter giving rise to a Non-Tax Claim then subject to the Sellers having made payment to the Purchaser in respect of the relevant Non-Tax Claim and providing appropriate indemnities and security for any costs and expenses incurred, the Purchaser shall procure that all appropriate steps are taken to enforce such recovery. If any sum is so recovered then the amount payable by the Sellers in respect of that Non-Tax Claim shall be reduced by an amount equal to the sum so recovered (less the reasonable costs and expenses properly incurred in effecting such recovery and any Taxation which may be payable by the Purchaser or the relevant Target Company as a result of its receipt).

100

- The Purchaser shall not make a claim or take any proceedings against the Sellers at any time in respect of any matter, thing or omission occurring prior to Closing, save under the Warranties or otherwise under this Agreement.
- 10.6 Nothing in this Agreement shall operate to reduce the Purchaser's common law duty to mitigate any loss giving rise to a Claim (other than the provisions of the Tax Covenant).
- After Closing the sole remedy of the Purchaser against the Sellers under or in respect of the Warranties shall be in damages for breach of contract and the Purchaser shall have no right to rescind this Agreement after the date hereof for breach of any of the Warranties or under the provisions of the Misrepresentation Act 1967 or for any reason whatsoever.
- 10.8 The Purchaser shall not make a claim or take any proceedings against the Sellers at any time in respect of any loss suffered by it, or by a Target Company, arising from the failure by Alba plc or Alba Radio Limited to deliver stock (free of charge) to any Target Company in an amount of £230.000.
- 10.9 The Sellers are not liable for any Warranty Claim to the extent that the Warranty Claim relates to matters Disclosed in the Disclosure Letter and / or the Disclosure Bundle.

11. Warranties in respect of Geneva Industries Limited

The Purchaser agrees, in favour of SJ, that to the extent it receives payment from DMJ, on behalf of and by way of discharge of SJ's liability for

Schedule 10

PROPERTY

1. The freehold property known as Land at Northfield Drive, Northfield registered at the Land Registry with title number BM278640, in relation to which the Lease has been granted to the Purchaser

102

Schedule 11

OWNED IP

1.	Domain Names
	www.willentrading.co.uk
	www.genevaindustries.co.uk
	www.genevaauctions.co.uk
	www.stockauctions.co.uk
	www.genevastockclearance.co.uk
	www.stockauctions.com
	www.stockauctionsadmin.co.uk
	www.stockrunner.co.uk
	www.stockhunter.co.uk

103

IN RELATION TO THE SALE OF SHARES IN GENEVA INDUSTRIES LIMITED and WILLEN TRADING LIMITED, THIS AGREEMENT has been entered into by the Parties on 5 April 2008.

SIGNED by DARREN LEE INNOCENT as attorney for DAVID MARK JACOBS under a power of attorney dated 20 March 2008))) /s/ Darren Lee Innocent	
SIGNED by DARREN LEE INNOCENT as attorney for SIMON JACOBS under a power of attorney dated 20 March 2008))) /s/ Darren Lee Innocent	
SIGNED by DARREN LEE INNOCENT)) /s/ Darren Lee Innocent	
SIGNED by for and on behalf of LIQUIDITY SERVICES LIMITED)))) /s/ William P. Angrick II	I
SIGNED by LIQUIDITY SERVICES, INC. acting by:))) /s/ William P. Angrick II	ī

IN RELATION TO THE SALE OF SHARES IN GENEVA April 2008.	AUCTIONS LIMITED, THIS AGREEMENT has been entered into by the Parties on 6
SIGNED by DARREN LEE INNOCENT as attorney for DAVID MARK JACOBS under a power of attorney dated 20 March 2008))) /s/ Darren Lee Innocent
SIGNED by DARREN MALCOLM DORRINGTON)) /s/ Darren Malcolm Dorrington
SIGNED by for and on behalf of LIQUIDITY SERVICES LIMITED))) /s/ William P. Angrick III
SIGNED by LIQUIDITY SERVICES, INC. acting by:)) /s/ William P. Angrick III
	Duly authorised officer

SALE OF GOVERNMENT PROPERTY							
AMENDMENT OF INVITATION FOR BIDS/MODIFICATION OF CONTRACT							
1. AMENDMENT TO INVITATION FOR BIDS NO.:	2. EFFECTIVE DATE	PAGE 1 O	F 3 PAGES				
SUPPLEMENTAL AGREEMENT NO.:14		05/13/08					
3. ISSUED BY	NAME AND ADDRESS WHERE BIDS ARE RECEIVED						
DEFENSE REUTILIZATION AND MARKETING SE INTERNATIONAL SALES OFFICE, ATTN:DRMS-J 74 N WASHINGTON STREET BATTLE CREEK MI 49017-3092	DEFENSE REUTILIZATION AND MARKETING SERVICE INTERNATIONAL SALES OFFICE, ATTN:DRMS-J362 74 N WASHINGTON STREET BATTLE CREEK MI 49017-3092						
o AMENDMENT OF INVITATION FOR BIDS NO. (See Item 6)	DATED	x MODIFICATION OF CONTRACT NO 8) 99-0001-0002). (See Item	DATE 06/13/01			
6 THIS DLOCK ADDITIS ONLY TO AMENDMENTS INVITATIONS FOR DIDS							

THIS BLOCK APPLIES ONLY TO AMENDMENTS INVITATIONS FOR BIDS

The above numbered invitation for bids is amended as set forth in Item 9. Bidders must acknowledge receipt of this amendment unless indicated otherwise in Item 11 prior to the hour and date specified in the invitation for bids, or as amended, by one of the following methods:

- By signing and returning copies of this amendment;
- By acknowledging receipt of this amendment on each copy of the bid submitted; or
- (c) By separate letter or telegram which includes a reference to the invitation for bids and amendment number.

FAILURE OF YOUR ACKNOWLEDGMENT TO BE RECEIVED AT THE ISSUING OFFICE PRIOR TO THE HOUR AND DATE SPECIFIED MAY RESULT IN REJECTION OF YOUR BID. If by virtue of this amendment you desire to change a bid already submitted, such change may be made by telegram or letter, provided such telegram or letter makes reference to the invitation for bids and this amendment, and is received prior to the opening hour and date specified.

7. ACCOUNTING AND APPROPRIATION DATA (If required)

THIS APPLIES ONLY TO MODIFICATION OF CONTRACTS

This Supplemental Agreement is entered into pursuant to authority of Mutual Agreement

DESCRIPTION OF AMENDMENT/MODIFICATION (Except as provided below all terms and conditions of the document referenced in Item 5 remain in full force and effect)

Whereas Contract 99-0001-0002 was entered into on June 13, 2001 by and between the United States of America, hereinafter referred to as the Government, and SURPLUS ACQUISTION VENTURE (SAV), LLC, hereinafter referred to as the Contractor, and GOVERNMENT LIQUIDATION (GL), LLC, formed by the contractor to serve as the entity that processes DRMS assets, hereinafter referred to as the Purchaser, and whereas the contract involved the following in Invitation For Bid 99-0001:

0001: All Federal Stock Classes (FSCs) listed in Table IV-1 of solicitation on the DRMS accountable record that are demilitarization code A, B, or Q, located at various U.S. (to include Alaska and Hawaii), Puerto Rico and Guam military installations.

THE HOUR AND DATE FOR RECEIPT OF BIDS o IS NOT EXTENDED, o IS EXTENDED UNTIL O'CLOCK (LOCAL TIME) DATE IO. BIDDER/PURCHASER NAME AND ADDRESS (Include ZIP Code) 11. o BIDDER IS NOT REQUIRED TO SIGN THIS DOCUMENT x PURCHASE IS REQUIRED TO SIGN THIS DOCUMENT AND SURPLUS ACQUISITION VENTURE, LLC RETURN ORIGINAL AND 0 COPIES TO THE ISSUING 2131 K Street NW. 4th Floor OFFICE Washington, DC 20037 12. SIGNATURE FOR BIDDER/PURCHASER 15. UNITED STATES OF AMERICA /s/ William Angrick /s/ Neil A. Watters (SIGNATURE OF PERSON AUTHORIZED TO SIGN) (SIGNATURE OF CONTRACTING OFFICER) 13. NAME & TITLE OF SIGNER (Type or print) 17. DATE SIGNED 14. DATE SIGNED 16. NAME OF CONTRACTING OFFICER. (Type or print) WILLIAM ANGRICK 05/13/08 05/13/08 CEO NEIL A. WATTERS

GPO: 1970 OF-390-461 (40-X)

STANDARD FORM 114D JAN. 1970 EDITION GENERAL SERVICES ADMINISTRATION FPMR (41 CFR) 101-45.3 114-501

CONTRACT NUMBER 99-0001-0002 Supplemental Agreement 14

- a. To extend the performance period of sales contract number 99-0001-0002 by 180 days;
- b. To modify the product pool available to be delivered under this contract during the contract extension period;
- c. To provide for a termination clause whereby either party, with 60 days written notice to the other party, may terminate the contract without cost to the Government; and
- d. The Contractor's revenue share will be increased to 39.5 percent during the contract extension period for property issued to the Contractor commencing with Delivery Order 358.

NOW, therefore, it is mutually agreed between the parties hereto:

Part II, GENERAL STATEMENTS OF CONTRACT TERMS, is modified as follows:

Paragraph B, Product Pool and Property Flow, the following language added therein:

B. PRODUCT POOL AND PROPERTY FLOW

During the 180-day extension, hereinafter referred as "contract extension", Contractor will receive property at CONUS locations to include Alaska, Hawaii, and Guam. Contractor is only entitled to property that appears on a Delivery Order during the contract extension. The product pool during the contract extension will primarily consist of demilitarization code A items. The Government will utilize this contract as the primary channel for the disposition of usable surplus property that survives Reutilization, Transfer, and Donation (RTD) screening and is determined by DRMS to be "safe to sell". However, the Government, in its sole discretion, during the contract extension, may include certain demilitarization code B and Q items on a Delivery Order, or exclude certain demilitarization code A items from sale, based on national security concerns. Contractor will be guaranteed property with a minimum monthly acquisition value of \$100,000,000, averaged over the term of the contract extension. Upon mutual decision between DRMS and the Contractor, the product pool may include non-DRMS assets that are not on the DRMS accountable record. Title to these assets will remain with the Government until resale to the Contractor's purchaser(s) and removal of the items from Government premises. The Contractor will be required to account for, identify, and pay DRMS separately for non-DRMS assets. A 12% sales fee applies to the resale of these assets instead of the typical revenue share identified elsewhere in this contract. DRMS has the option to phase-out the property stream at any location once DRMS awards Invitation For Bid (IFB) 08-0001.

Paragraph D, Phase-In and Performance Period, the following language added therein:

D. PHASE-IN PERIOD AND PERFORMANCE PERIOD

Performance period of this contract is extended 180 days. The contract may be terminated by either party without cost to the Government upon a 60-day written notice to the other party. Notice of termination may be issued anytime after the issuance of Delivery Order 358.

Part VI, ADDITIONAL TERMS AND CONDITIONS OF SALE, is modified as follows:

2

Article Two, Section 1, Performance Period, the following language is added therein:

Performance period of this contract extension is 180 days until December 19, 2008. The contract may be terminated by either party without cost to the Government upon a 60-day written notice to the other party. Notice of termination may be issued anytime after the issuance of Delivery Order 358.

Article Three, Section 1(A), R/T/D Review is deleted in its entirety during the contract extension period.

Contractor is only entitled to property that appears on Delivery Order 358 and thereafter.

Article Twenty-One, Section 2, Cessation of Property Referrals, is deleted in its entirety and the following language included therein:

There shall be no further referrals of property by DRMS for sale to Contractor from the Wind-Up commencement date forward. Submission of monthly reports shall continue until Wind-Up is completed.

Article Twenty-One, Section 3, Conduct of Wind-Up is deleted in its entirety and the following language included therein:

From the Wind-Up commencement date forward, Purchaser and Contractor shall perform their remaining obligations hereunder in accordance with the provisions of this contract. If the Contractor is unable or unwilling to do so, or if DRMS reasonably concludes that the value of its remaining interest in Contractor's net proceeds is at risk by reason of actual or threatened failure to comply with provisions of the contract, DRMS may seek administrative or judicial relief to preserve the value of such remaining interest in Contractor's net proceeds. Contractor must vacate all property from Government provided warehouse locations within 90 days of the last Delivery Order. Property issued under this contract must be removed from Government premises within 90 days of the last Delivery Order.

Article Sixteen, Section 4(D)(5), Distributions, the following language is included therein:

During the contract extension period, the Contractor's revenue share is 39.5% for all property issued from Delivery Order 358 and thereafter that is resold under the terms and conditions of this contract.

The terms and conditions not specifically addressed in this contract modification shall remain in full force and effect during the contract extension period. Parties hereto acknowledge and agree that any terms and conditions of sale contract number 99-0001-0002 that deviate from the explicit language of the terms and conditions of this contract modification shall be considered non-binding and of no force and effect.

CERTIFICATION PURSUANT TO RULE 13a-14(a) OF THE SECURITIES EXCHANGE ACT OF 1934

I, William P. Angrick, III, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of Liquidity Services, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal controls over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about
 the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such
 evaluation; and
 - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 8, 2008

/s/ William P. Angrick, III

By: William P. Angrick, III

Title: Chairman of the Board of Directors and

Chief Executive Officer

CERTIFICATION PURSUANT TO RULE 13a-14(a) OF THE SECURITIES EXCHANGE ACT OF 1934

I, James M. Rallo, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of Liquidity Services, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal controls over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about
 the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such
 evaluation; and
 - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 8, 2008

/s/ James M. Rallo

By: James M. Rallo

Title: Chief Financial Officer and Treasurer

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the quarterly report of Liquidity Services, Inc. (the "Company") on Form 10-Q for the period ended June 30, 2008 as filed with the Securities and Exchange Commission (the "Report"), I, William P. Angrick, III, Chief Executive Officer of the Company, certify, to the best of my knowledge, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that:

- (1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

August 8, 2008

/s/ William P. Angrick, III
William P. Angrick, III
Chairman of the Board of Directors and Chief
Executive Officer

THE FOREGOING CERTIFICATION IS BEING FURNISHED SOLELY PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002 AND IS NOT BEING FILED AS PART OF THE FORM 10-Q OR AS A SEPARATE DISCLOSURE DOCUMENT.

A SIGNED ORIGINAL OF THIS WRITTEN STATEMENT REQUIRED BY SECTION 906, OR OTHER DOCUMENT AUTHENTICATING, ACKNOWLEDGING, OR OTHERWISE ADOPTING THE SIGNATURE THAT APPEARS IN TYPED FORM WITHIN THE ELECTRONIC VERSION OF THIS WRITTEN STATEMENT REQUIRED BY SECTION 906, HAS BEEN PROVIDED TO LIQUIDITY SERVICES, INC. AND WILL BE RETAINED BY LIQUIDITY SERVICES, INC. AND FURNISHED TO THE SECURITIES AND EXCHANGE COMMISSION OR ITS STAFF UPON REQUEST.

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the quarterly report of Liquidity Services, Inc. (the "Company") on Form 10-Q for the period ended June 30, 2008 as filed with the Securities and Exchange Commission(the "Report"), I, James M. Rallo, Chief Financial Officer of the Company, certify, to the best of my knowledge, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that:

- (1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

August 8, 2008

/s/ James M. Rallo James M. Rallo

Chief Financial Officer and Treasurer

THE FOREGOING CERTIFICATION IS BEING FURNISHED SOLELY PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002 AND IS NOT BEING FILED AS PART OF THE FORM 10-Q OR AS A SEPARATE DISCLOSURE DOCUMENT.

A SIGNED ORIGINAL OF THIS WRITTEN STATEMENT REQUIRED BY SECTION 906, OR OTHER DOCUMENT AUTHENTICATING, ACKNOWLEDGING, OR OTHERWISE ADOPTING THE SIGNATURE THAT APPEARS IN TYPED FORM WITHIN THE ELECTRONIC VERSION OF THIS WRITTEN STATEMENT REQUIRED BY SECTION 906, HAS BEEN PROVIDED TO LIQUIDITY SERVICES, INC. AND WILL BE RETAINED BY LIQUIDITY SERVICES, INC. AND FURNISHED TO THE SECURITIES AND EXCHANGE COMMISSION OR ITS STAFF UPON REQUEST.